Access Services Tariff Section 9 1st Revised Sheet 1 Replacing Original Sheet 1

#### ACCESS SERVICES

### 9. DIRECTORY ASSISTANCE SERVICE

The Telephone Company will provide Directory Assistance (DA) Service to an IC from Directory Assistance Service locations (DA location).

9.1 General Description

DA Service provides Directory Access Service to DA locations, use of DA access equipment and use of DA operators.

- 9.2 Undertaking of the Telephone Company
  - A. A Telephone Company DA operator, when furnished a name and locality, will provide or attempt to provide the telephone number listed in the Telephone Company DA records associated with the name given at the rates and charges as set forth in Paragraph 9.6, following. The Telephone Company's contact with the IC's End User shall be limited to that effort necessary to process an IC's End User's request for a telephone number; and the Telephone Company will not transfer, forward or redial an IC's end user call to any other location for any purpose other than provision of DA Service. Telephone Company DA Service is not available for 800-555-traffic.
  - B. A maximum of two (2) requests for telephone numbers will be accepted per call to the DA operator.
  - C. A telephone number which is not listed in DA records will not be available to the IC's End User.
  - D. The Telephone Company will specify the DA location which provides the DA Service for each Numbering Plan Area Code (NPA).

When it becomes necessary, as determined by the Telephone Company, to change a DA location, the Telephone Company will notify the IC's involved six months prior to the change. For such changes, the regulations as set forth in Section 2, Paragraph 2.1.4, preceding, apply.

E. When DA Service is ordered, Directory Access Service will be provided between the IC terminal location and the DA location by the Telephone Company at rates and charges as set forth in Paragraph 9.6, following.

Issued:

October 14, 1987

Effective:

October 16, 1987



#### Access Services Tariff Section 9 Original Sheet 1

DEC 20 0

#### ACCESS SERVICES

9. DIRECTORY ASSISTANCE SERVICE

The Telephone Company will provide Directory Assistance (DA) Service to an IC from Directory Assistance Service locations (DA location).

9.1 General Description

DA Service provides Directory Access Service to DA locations, use of DA access equipment and use of DA operators.

9.2 Undertaking of the Telephone Company

A. A Telephone Company DA operator, when furnished a name and locality, will provide or attempt to provide the telephone number listed in the Telephone Company DA records associated with the name given at the rates and charges as set forth in Paragraph 9.6, following. The Telephone Company's contact with the IC's End User shall be limited to that effort necessary to process an IC's End User's request for a telephone number; and the Telephone Company will not transfer, forward or redial an IC's end user call to any other location for any purpose other than provision of DA Service.

- B. A maximum of two (2) requests for telephone numbers will be accepted Fibr call to the DA operator.
- C. A telephone number which is not listed in DA records will not the favailable to the IC's End User.
- D. The Telephone Company will specify the DA location which provides the DA Service for each Numbering Plan Area Code (NPA).

When it becomes necessary, as determined by the Telephone Company, to change a DA location, the Telephone Company will notify the IC's involved six months prior to the change. For such changes, the regulations as set forth in Section 2, Paragraph 2.1.4, preceding, apply.

E. When DA Service is ordered, Directory Access Service will be provided between the IC terminal location and the DA location by the Telephone Company at rates and charges as set forth in Paragraph 9.6, following.

Issued: DEC 2 9 1983 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 9 2nd Revised Sheet 2 Replacing 1st Revised Sheet 2

### ACCESS SERVICES

### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.2 Undertaking of the Telephone Company-(Continued)

- E. (Continued)
  - 1. General

Each Directory Access Service will consist of the following:

- An Interface Group equipped with an available Premises Interface Code at the customer's premises.
- Directory Transport between the premises of the ordering customer and the DA location.

When required by the Telephone Company, a separate Directory Access Service trunk group will be provided for DA Service for each NPA. Separate trunk groups will be required when the Telephone Company notifies the customer that the mechanized search of its data base and its mechanized operator practices require a mechanized identification of the NPA code for which the customer's End User desires DA information.

Further, when an access tandem is available and is provided, the Directory Access Service will be provided at Telephone Company's choice either as a separate Directory Access Service trunk group or in combination with FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service.

2. Interface Group and Premise Interface Code

Interface Groups 2 through 10 as set forth in Paragraph 6.1.2, A., preceding, are available for Directory Access Service. When only Directory Access Service is provided, only the following Premise Interface Codes are available:

4DS9-15	6EA2-E	4RV2-0
4DS9-31	6EA2-M	4AH5-B
4DS0-63	4SF3	4AH6-C
4DS6-44		4AH6-D
4DS6-27		

Issued:

March 26, 1993

Effective:

April 11, 1993





No supplement to this Access Services Tariff tariff will be issued Section 9 except for the purpose 1st Revised Sheet 2 of canceling this tariff. Replacing Original Sheet 2 ACCESS SERVICES REGEIVED 9. DIRECTORY ASSISTANCE SERVICE-(Continued) 9.2 Undertaking of the Telephone Company-(Continued) E. (Continued) 1. General Each Directory Access Service ution of Service Company Sign JUN 27 1986 MISSUURI Public Service Commission (CT) Each Directory Access Service with on the following: - An Interface Group pricipled crath an available Premises Interface Code at the customer's premises. - Directory Transport between the premises of the ordering customer and the DA location. When required by the Telephone Company, a separate Directory Access Service trunk group will be provided for DA Service for each NPA. Separate trunk groups will be required when the Telephone Company notifies the customer that the mechanized search of its data base and its mechanized operator practices require a mechanized identification of the NPA code for which the customer's End User desires DA information. Further, when an access tandem is available and is provided, the Directory Access Service will be provided at Telephone Company's choice either as a separate Directory Access Service trunk group or in combination with Feature Group B, C or D Switched Access Service. (CT) 2. Interface Group and Premise Interface Code Interface Groups 2 through 10 as set forth in Paragraph 6.1.2, A., preceding, are available for Directory Access Service. When only Directory Access Service is provided, only the following Premise Interface Codes are available: 4DS9-15 6EA2-E 4RV2-0 4DS9-31 6EA2-M 4AH5-B FNLED 4SF3 4AH6-Č 4DS0-63 4DS6-44 4AH6-Ď 4DS6-27 HH 1 1986 86-84 Public Service Commission Effective: JUL 1 1986 Issued: JUN 27 1986

#### ACCESS SERVICES

9. DIRECTORY ASSISTANCE SERVICE-(Continued)

DEC 21

Original Sheet 2

Section 9

Access Services Tariff

9.2 Undertaking of the Telephone Company-(Continued)

E. (Continued)

1. General

Each Directory Access Service will consist of the following:

- A Switched Access Service Access Connection equipped with one of the following Switched Access Service Access Connection IC Interface codes:

4DS9-15	6EA2-E	4RV2-0
4DS9-31	6EA2-M	4AH5-B
4DS0-63	6EX2-B	4AH6-C
4DS6-44	4SF3	4AH6-D
4DS6-27	2RV3-0	

- Directory Transport between the IC terminal location serving wire center and the DA location.

When required by the Telephone Company, a separate trunk group will be provided for DA Service for each NPA.

2. Access Connection and IC Interface Code

The Switched Access Service Access Connection and IC Interface Codes are provided as set forth in Section 6, preceding. Further, when an access tandem is provided, the Access Connection for Directory Access Service will be provided, at Telephone Company choice, either as a separate trunk group or in association with Switched Access Service. Except as set forth in Paragraph 9.4, A., following the Access Connection and IC Interface Codes provided under a Special order for Directory Access Service are subject to the order conditions as set forth in Section 5, preceding. For purposes of applying the order regulations, an DA location is considered to be an End User serving wire center.

BY CR S. H Z-PUBLIC SERVICE COMMISSION

Effective: JAN 0 1 1984 83 - 253

3. Directory Transport

Directory Transport provides the transmission facilities and transport termination between the IC terminal location and the DA location. For purposes of determining Directory Transport mileage, distance will be measured from the wire center that normally serves the IC terminal location to the DA location(s).

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri <u>Public Series</u> (a. Southwestern Bell Telephone Company St. Louis, <u>Mis</u>souri

#### ACCESS SERVICES

#### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.2 Undertaking of the Telephone Company-(Continued)

- E. (Continued)
  - 2. (Continued)

Such Premise Interface Codes are described in Section 6, Paragraph 6.1.2, A.11., preceding. When Directory Access Service is combined with FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service, the Premise Interface Code for the combination will be the available Premise Interface Code provided for the FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service ordered by the customer. Except as set forth in Paragraph 9.4, A., following, the Interface Groups and Premises Interface Codes provided under a Special Order for Directory Access Service are subject to the order conditions as set forth in Section 5., preceding. For purposes of applying the order regulations, a DA location is considered to be a customer End User serving wire center.

3. Directory Transport

Directory Transport provides the transmission facilities and transport termination between the IC terminal location and the DA location. For purposes of determining Directory Transport mileage, distance will be measured from the wire center that normally serves the IC terminal location to the DA location(s).

Issued: Mar

March 26, 1993

Effective:

April 11, 1993



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 9 Original Sheet 2.01

RECEIVED

JUN 2 7 1986

MISSURF

Public Service Commission

ACCESS SERVICES

- 9. DIRECTORY ASSISTANCE SERVICE-(Continued)
- 9.2 Undertaking of the Telephone Company-(Continued)
  - E. (Continued)

Issued:

JUN 27 1986

2. (Continued)

Such Premise Interface Codes are described in Section 6, Paragraph 6.1.2, A.11., preceding. When Directory Access Service is combined with Feature Group B, C or D Switched Access Service, the Premise Interface Code for the combination will be the available Premise Interface Code provided for the Feature Group B, C or D Switched Access Service ordered by the customer. Except as set forth in Paragraph 9.4, A., following, the Interface Groups and Premises Interface Codes provided under a Special Order for Directory Access Service are subject to the order conditions as set forth in Section 5., preceding. For purposes of applying the order regulations, a DA location is considered to be a customer End User serving wire center.

3. Directory Transport

Directory Transport provides the transmission facilities and transport termination between the IC terminal location and the DA location. For purposes of determining Directory Transport mileage, distance will be measured from the wire center that normally serves the IC terminal location to the DA location(s).

CANCELLED Public Service Commission MISSOURI FILED 1986 blic Service Commission 1 1986 JUL Effective:

### Access Services Tariff Section 9 2nd Revised Sheet 3 Replacing 1st Revised Sheet 3

# ACCESS SERVICES

# 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.2 Undertaking of the Telephone Company-(Continued)

- E. (Continued)
  - 3. Directory Transport-(Continued)

Directory Transport is a two-way voice frequency transmission path composed of facilities determined by the Telephone Company. The two-way voice frequency path transports calls in the terminating direction (from the IC terminal location to the DA location). The voice frequency transmission path may be comprised of any form or configuration of plant capable of and typically used in the telecommunications industry for the transmission of voice and associated telephone signals within the frequency band width of approximately 300 to 3000 Hz.

The Telephone Company will determine whether the Directory Access Service is to be routed directly to a DA location or through an access tandem switch appropriately equipped for DA measurement and served by DA trunks to the DA location when such an access tandem switch is available. The combination of FGB, FGC, FGD, BSA-B, BSA-C or BSA-D Switched Access Service with DA Service will only be provided at such available and appropriately equipped access tandem switches. If the IC desires the traffic routing to be other than that selected by the Telephone Company, it may request a cooperative effort to determine if customer-specified traffic routing can be used in lieu of the Telephone Company-selected traffic routing.

When Directory Transport is provided using a direct route to the DA location, no address signaling is provided. When Directory Transport is provided with the use of an access tandem switch, wink start-start pulsing signaling is provided at the access tandem switch. The IC will be notified by the Telephone Company when access tandem routing is provided, and the IC shall address each call to the DA location using NPA + 555-1212 or when required by the Telephone Company, 555-1212. Only NPA codes handled by the DA Service access location served by the access tandem switch will be processed.

The number of Directory Transport transmission paths provided is based on the IC's order and is determined by the Telephone Company in a manner similar to Switched Access Service transmission paths as set forth in Section 6, Paragraph 6.8.5, preceding.

Directory Transport may, at the option of the IC, be provided for both interstate and intrastate communications. When the IC requests such mixed access, the intrastate Directory Transport charges will be determined by the Telephone Company using the reports furnished by the IC as set forth in Section 2, Paragraph 2.3.12, preceding.



(FC)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

- Access Services Tariff Section 9
  - lst Revised Sheet 3

REGEIVED

JUN 2 7 1986

Replacing Original Sheet 3

#### ACCESS SERVICES

- 9. DIRECTORY ASSISTANCE SERVICE-(Continued)
  - 9.2 Undertaking of the Telephone Company-(Continued)
  - E. (Continued)
    - 3. Directory Transport-(Continued)

Directory Transport is a two-way voice frequency transmission bach composed of facilities determined by the Telephone Company Sign two-way voice frequency path transports calls in the terminating direction (from the IC terminal location to the DA location). The voice frequency transmission path may be comprised of any form or configuration of plant capable of and typically used in the telecommunications industry for the transmission of voice and associated telephone signals within the frequency band width of approximatel CELLED 300 to 3000 Hz.

(AT) The Telephone Company will determine whether the Directory AcAPS 11 1993 Service is to be routed directly to a DA location or through an access fraction of the switch appropriately equipped for DA measurement and Barsed by DA trunks to the DA location when such an access tandan to Service Commission available. The combination of Feature Group B, C or D Switched Access [R] Service with DA Service will only be provided at such available and appropriately equipped access tandem switches. If the IC desires the traffic routing to be other than that selected by the Telephone Company, it may request a cooperative effort to determine if customer-specified traffic routing.

> When Directory Transport is provided using a direct route to the DA location, no address signaling is provided. When Directory Transport is provided with the use of an access tandem switch, wink start-start pulsing signaling is provided at the access tandem switch. The IC will be notified by the Telephone Company when access tandem routing is provided, and the IC shall address each call to the DA location using NPA + 555-1212 or when required by the Telephone Company, 555-1212. Only NPA codes handled by the DA Service access location served by the access tandem switch will be processed.

The number of Directory Transport transmission paths provided is based on the IC's order and is determined by the Telephone Company in a manner similar to Switched Access Service transmission paths as set forth in Section 6, Paragraph 6.5.5, preceding.

Directory Transport may, at the option of the IC, be provided for obth interstate and intrastate communications. When the IG, requests, such mixed access, the intrastate Directory Transport charges will be delimited termined by the Telephone Company using the reports furnished by the IC as set forth in Section 2, Paragraph 2.3.12, preceding.

Issued: JUN 27 1986

(AT)

(CT)

Effective: JUL 1 1986

Access Services Tariff Section 9 Original Sheet 3

#### ACCESS SERVICES

9. DIRECTORY ASSISTANCE SERVICE-(Continued)

DEC 20 cm

- 9.2 Undertaking of the Telephone Company-(Continued)
- E. (Continued)
  - 3. Directory Transport-(Continued)

Directory Transport is a two-way voice frequency transmission path composed of facilities determined by the Telephone Company. The two-way voice frequency path transports calls in the terminating direction (from the IC terminal location to the DA location). The voice frequency transmission path may be comprised of any form or configuration of plant capable of and typically used in the telecommunications industry for the transmission of voice and associated telephone signals within the frequency band width of approximately 300 to 3000 Hz.

The Telephone Company will determine whether the Directory Access Service is to be routed directly to a Directory Assistance Service access location or through an appropriately equipped access tandem switch when such an access tandem switch is available. If the IC desires the traffic routing to be other than that selected by the Telephone Company, it may, for additional charges as set forth in Paragraph 9.6, following, specify the traffic routing to be used in lieu of the Telephone Company-selected traffic routing.

When Directory Transport is provided using a direct route to the DA location, no address signaling is provided. When Directory Transport is provided with the use of an access tandem switch, wink start-start pulsing signaling is provided at the access tandem switch. The IC will be notified by the Telephone Company when access tandem routing is provided, and the IC shall address each call to the DA location using NPA + 555-1212. Only NPA codes handled by the DA Service access location served by the access tandem switch.

served by the access tandem super file processed. The number of Directory Transport transmission paths provided is based on the IC's order and is determined by the Telephone Company in a manner similar to Local Transport transmission paths as set forth in Section 6, Paragraph 6.5.5, preceding of C 5 the

BY Directory Transport maysulate the provided for both interstate and intrastate communications. When the IC requests such mixed access, the intrastate Directory Transport charges will be determined by the Telephone Company using the reports furnished by the IC as set forth in Section 2, Paragraph 2.3.14, preceding.

Effective:

Issued: DE

DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Public Service Contraison St. Louis, Missouri

JAN 0 1 1984 N - 1 1884

Access Services Tariff Section 9 1st Revised Sheet 4 **Replacing Original Sheet 4** 

#### ACCESS SERVICES

#### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

- 9.2 Undertaking of the Telephone Company-(Continued)
- E. (Continued)
  - 3. Directory Transport-(Continued)

Except as set forth in Paragraph 9.4, A., following, Directory Transport provided under a Special Order is subject to the order conditions as set forth in Section 5, preceding.

- AT) Directory Transport is provided with a Local Transport Interface Group as set forth in Section 6, Paragraph 6.1.2, A., preceding. Only Local Transport Interface Groups 2 through 10 will be provided.
  - 4. Special Facilities Routing

An IC may request that Directory Access Service be provided via Special Facilities Routing. The regulations, rates and charges for Special Facility Routing (Avoidance, Diversity and Cable Only) are as set forth in Section II, following.

- 5. Design Layout Report
- The Telephone Company will provide to the customer the makeup of the facilities and services provided under this section as Directory Access Service. This information will be provided in the form of a Design
- Layout Report similar to that as set forth in Paragraph 6.1.4., preceding. Design Layout Reports for Directory Access Service will be provided only when specifically requested by the customer. The Design Layout Report will be provided to the customer at no charge and will be reissued or updated whenever the facilities provided for the customer's use are materially changed.
  - 6. Transmission Performance
- (CT)Directory Access Service is provided with either Type A or B Transmission Performance. The parameters associated with these performances are guaranteed to the DA location, whether routed directly or via an access tandem. Type B Transmission Performance is provided with Interface (RT)Groups 2 through 10 when routed direct to a DA location. Type A Transmission Performance is provided with Interface Groups 2 through 10 (AT)when routed via an access tandem switch.

Issued:	June 27, 1986	Effective:	July 1, 1986	



#### Access Services Tariff Section 9 Original Sheet 4

DEC 2 Common

IIL R

83-253

#### ACCESS SERVICES

- DIRECTORY ASSISTANCE SERVICE-(Continued) 9.
  - 9.2 Undertaking of the Telephone Company-(Continued)
  - (Continued) Ε.
    - Directory Transport-(Continued) 3.

Except as set forth in Paragraph 9.4, A., following, Directory Transport provided under a Special Order is subject to the order conditions as set forth in Section 5, preceding.

4. Special Facilities Routing

An IC may request that Directory Access Service be provided via Special Facilities Routing. The regulations, rates and charges for Special Facility Routing (Avoidance, Diversity and Cable Only) are as set forth in Section 11, following.

5. Design Layout Report

The Telephone Company will provide to the IC the makeup of the facilities and services provided under this section as Directory Access Service. This information will be provided in the form of a Design Layout Report.

6. Transmission Performance

DEC 2 9 1983

Issued:

Directory Access Service is provided with either Type A, B or C Transmission Performance. The parameters associated with these performances are guaranteed to the DA location, whether routed directly or via an access tandem. Type C Transmission Performance is provided with Interface Group 1 when routed direct to a DA location. Type B Transmission Performance is provided with Interface Groups 2 through 10 when routed direct to a DA location. Type A Transmission Performance is provided with Interface Groups 2 through 10 when routed via an access tandem.

JUL 1 1986

PUBLIC SERVICE COMMISSION

OF MISSOUR

Effective:

CHE

By R. D. BARRON, Vice President-Missouri Public Service Connel

JAN 0 1 1984

A. B and C Transmission Performance Capabilities are set forth in Section 6, Paragraph 6.4.1, preceding. GANGELLED

BY

Southwestern Bell Telephone Company St. Louis, Missouri

# ACCESS SERVICES

## 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.2 Undertaking of the Telephone Company-(Continued)

- E. (Continued)
  - 6. (Continued)

When DA Service is combined with FGD or BSA-D Switched Access Service, Type A Transmission Performance is provided. When DA Service is combined with FGB or BSA-B Switched Access Service, Type B Transmission Performance is provided for Interface Groups 2 through 10. When DA Service is combined with FGC or BSA-C Switched Access Service, Type B Transmission Performance is provided.
 A and B Transmission Performance Capabilities are set forth in

Section 6, Paragraph 6.7.1, preceding.

Issued:



ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

DIRECTORY ASSISTANCE SERVICE-(Continued)

9.

(AT)

(MT)

(CT)

9.2

Ε.

Access Services Tariff Section 9 Original Sheet 4.01

RECEIVED JUN 27 1986 Undertaking of the Telephone Company-(Continued) MISSUUKI Public Service Commission

6. (Continued)

(Continued)

When DA Service is combined with Feature Group D Switched Access Service, Type A Transmission Performance is provided. When DA Service is combined with Feature Group B Switched Access Service, Type B Transmission Performance is provided for Interface Groups 2 through 10. When DA Service is combined with Feature Group C Switched Access Service, Type B Transmission Performance is provided.

A and B Transmission Performance Capabilities are set forth in Section 6, Paragraph 6.4.1, preceding.

CANCELLED APR 11 1993 BY <u>Lot R. J.</u> BY <u>Lot R. J.</u>

RALEN 超出 1 1986 86-84 ublic Service Commission

Issued: JUN 27 1986

Effective: JUL 1 1986

# ACCESS SERVICES

## 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.2 Undertaking of the Telephone Company-(Continued)

- E. (Continued)
  - 7. Acceptance Testing and Testing Capabilities

The acceptance testing and testing capabilities for Directory Access Service traffic routed through an access tandem are the same as those for the associated FGC, FGD, BSA-C or BSA-D end office switching. The acceptance testing for Directory Access Service traffic routed directly to or routed in a separate trunk group through an access tandem to the DA location will be as set forth in Section 6, Paragraph 6.1.4, preceding. The testing capabilities for Directory Access Service traffic routed directly to or routed in a separate trunk group through an access tandem to the DA location will be as set forth for cooperative scheduled testing or manual scheduled testing in Section 13, following.

- F. Trunk-side switching is provided at the DA Service access location. The DA Service access location will provide trunk-answer and disconnect supervisory signaling.
- G. The Telephone Company will distribute the calls received over the Directory Access Services to the DA operators using the DA location access equipment.
- H. In the event that the telephone number is unavailable to the DA operator, no credit applies for the charge for the call to the DA operator.When the DA location or DA operator equipment or terminals are out of service due to a Telephone Company equipment failure or an incorrect number is provided, a credit as set forth in Paragraph 9.4, H., following, will apply.
- I. DA Service may, at the option of the IC, be provided for interstate and intrastate communications. When the IC requests such mixed access, the intrastate DA Service charges will be determined by the Telephone Company using the reports furnished by the IC as set forth in Section 2, Paragraph 2.3.12, preceding.



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Acce	ess Services Tariff Section 9
-	lst Revised Sheet 5 ng Original Sheet 5
	REGEIVED
	JUN 27 1986
	MISSUURI Public Service Commission

APR 11 1993

1986

<u>, Hit</u> 1

86-84 Public Service Commission

ACCESS SERVICES

9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.2 Undertaking of the Telephone Company-(Continued)

E. (Continued)

visory signaling.

(RT)

(AT)

(AT)

7. Acceptance Testing and Testing Capabilities

The acceptance testing and testing capabilities for Directory Access Service traffic routed through an access tandem are the same as those for the associated Feature Group C or D end office switching. The acceptance testing for Directory Access Service traffic routed directly to or routed in a separate trunk group through an access tandem to the DA location will be as set forth in Section 6. Paragraph 6.1.4, preceding. The testing capabilities for Directory Access Service traffic routed directly to or routed in a separate trunk group through an access tandem to the DA location will be as set forth for cooperative schedule ing or manual scheduled testing in Section 13, following.

Public Service Commiss MISSOURI F. Trunk-side switching is provided at the DA Service access location. The DA Service access location will provide trunk-answer and disconnect super-

- G. The Telephone Company will distribute the calls received over the Directory Access Services to the DA operators using the DA location access equipment.
- (CT) H. In the event that the telephone number is unavailable to the DA operator, no credit applies for the charge for the call to the DA operator. When the DA location or DA operator equipment or terminals are out of service due to a Telephone Company equipment failure or an incorrect number is provided, a credit as set forth in Paragraph 9.4, H., following, will apply.
  - I. DA Service may, at the option of the IC, be provided for interstate and intrastate communications. When the IC requests such mixed access, the intrastate DA Service charges will be determined by the Telephone Company using the reports furnished by the IC as set forth in Section 2, Paragraph 2.3.12, preceding.

Issued: JUN 27 1986

Effective: JUL 1 1986

#### Access Services Tariff Section 9 Original Sheet 5

#### ACCESS SERVICES

DIRECTORY ASSISTANCE SERVICE-(Continued) 9.

DEC 2

- 9.2 Undertaking of the Telephone Company-(Continued)
- Ε. (Continued)
  - 7. Acceptance Testing and Testing Capabilities

The acceptance testing and testing capabilities for Directory Access Service traffic routed through an access tandem are the same as those for the associated Feature Group B. C or D end office switching. The acceptance testing for Directory Access Service traffic routed directly to the DA location will be as set forth in Section 6, Paragraph 6.1.6, preceding. The testing capabilities for Directory Access Service traffic routed directly to the DA location will be as set forth for cooperative scheduled testing or manual scheduled testing in Section 13, following.

8. Provision Of Other Than Telephone Company-Selected Traffic Routing

This option allows the IC to specify a particular routing for trunk groups in lieu of Telephone Company-Selected traffic routing, i.e., the IC may specify that the routing be on a direct-trunk basis or via an access tandem.

- F. Trunk-side switching is provided at the DA Service access location. The DA Service access location will provide trunk-answer and disconnect supervisory signaling.
- G. The Telephone Company will distribute the calls received over the Directory Access Services to the DA operators using the DA location access equipment.
- In the event that no, or an incorrect, telephone number is provided, except Ħ. when the DA location or DA operator equipment or terminals are out of service due to a Telephone Company equipment failure, no credit applies for the charge for the call to the DA operator. When the DA location or DA operator equipment or terminals are out of service due to a Telephone Company equipment failure, a credit as set forth in Paragraph 9.4, H., following, will apply.
- DA Service may, at the operion of the IC, be provided for interstate and I. intrastate communications. When the IC requests such mixed access, the intrastate DA Service charges will be determined by the Telephone Company using the reports furnished by the IC as set forth in Section 2, Paragraph BY 12 8.5 2.3.14, preceding. PUBLIC SERVICE COMMISSION

DEC 2 9 1983 Issued:

> By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St Innie Micenneissa

OF MISSOURI

Effective:

JAN 0 1 1984

Access Services Tariff Section 9 1st Revised Sheet 6 Replacing Original Sheet 6

# ACCESS SERVICES

### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.3 Obligations of the IC

- A. The IC shall determine and order the busy-hour minutes of capacity and interface type of Directory Access Services it needs for DA Service.
- (CP) B. When DA Service is initially ordered, the IC shall order the service for at least six months. Thereafter, additional service may be ordered for a minimum of six months. Not later than three months prior to the end of the six-month period, the IC shall notify the Telephone Company if the service is to be discontinued at the end of the six-month period. If no notice is received from the IC, the Telephone Company will automatically extend the service for another six months, and all appropriate charges as set forth in Paragraph 9.6, following, for another six months will apply.
  - C. The IC facilities at the IC terminal location shall provide the necessary on-hook and off-hook supervision.
  - D. When requested by the Telephone Company, the IC shall order a separate trunk group for DA Service for each NPA. The conditions when the customer will be requested to order separate trunk groups for each NPA are set forth in Paragraph 9.2, E.l., preceding.
  - E. When the IC bills its End Users, the IC shall be responsible for all contacts and arrangements with its End Users concerning the provision and maintenance of, and the billing and collecting of charges, for DA services furnished to its End Users. When the Telephone Company bills the IC End Users at the request of the IC, contacts and arrangements with IC End Users concerning the billing and collecting of charges will be as set forth in Section 8, Paragraph 8.2, preceding.
- CT) F. The customer understands that DA Operators will respond to two (2) requests per call only and will not transfer, forward or redial the call to another location for any purpose other than the provision of DA Service.

(AT)

Issued:

June 27, 1986 Effective:



Access Services Tariff Section 9 Original Sheet 6

DEC 20 3

#### ACCESS SERVICES

- DIRECTORY ASSISTANCE SERVICE-(Continued)
  - 9.3 Obligations of the IC
  - A. The IC shall determine and order the busy-hour minutes of capacity and interface type of Directory Access Services it needs for DA Service.
  - B. When DA Service is initially ordered, the IC shall order the service for at least one year. Thereafter, additional service may be ordered for a minimum of one year. Not later than six months prior to the end of the one-year period, the IC shall notify the Telephone Company if the service is to be discontinued at the end of the one-year period. If no notice is received from the IC, the Telephone Company will automatically extend the service for another year, and all appropriate charges as set forth in Paragraph 9.6, following, for another year will apply.
  - C. The IC facilities at the IC terminal location shall provide the necessary on-hook and off-hook supervision.
  - D. When requested by the Telephone Company, the IC shall order a separate trunk group for DA Service for each NPA.
  - When the IC bills its End Users, the IC shall be responsible for all con-Ε. tacts and arrangements with its End Users concerning the provision and maintenance of, and the billing and collecting of charges, for DA services furnished to its End Users. When the Telephone Company bills the IC End Users at the request of the IC, contacts and arrangements with IC End Users concerning the billing and collecting of charges will be as set forth in Section 8, Paragraph 8.2, preceding.
  - The IC shall notify its End Users through its tariff or other appropriate F. means that DA Operators will respond to two (25) Aquests per call only and will not transfer, forward or feddal the call to another location for any purpose other than the provision of DA Service. JUL 1 1986
  - 9.4 Payment Arrangements
  - Minimum Periods Α.

BY ARS 6 PUBLIC SERVICE COMMISSION

The minimum period for which DA Service and the Directory Access Service is provided and for which charges apply is one year. A minimum period of one year applies for each additional period of service ordered or extended.

If DA Service is discontinued prior to the end of each one-year period, the minimum monthly charge applies for each remaining month and fraction of month LILES in the one-year period.

Effective:

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouright's Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

JAN 0 1 198483 - 253

# ACCESS SERVICES

### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.4. Payment Arrangements

A. Minimum Periods

The minimum period for which DA Service and the Directory Access Service is provided and for which charges apply is six months. A minimum period of six months applies for each additional period of service ordered or extended.

If DA Service is discontinued prior to the end of each six-month period, the charges that apply for the remaining months are the nonrecoverable costs. Such costs include the nonrecoverable cost of equipment and material ordered, provided or used, plus the nonrecoverable cost of installation and removal including the costs of engineering, labor, supervision, transportation, rights-of-way and other associated costs less estimated net salvage.

Issued:

June 27, 1986

Effective:

July 1, 1986



### ACCESS SERVICES

#### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

- 9.4 Payment Arrangements-(Continued)
- B. Minimum Monthly Charge

DA Service is subject to a minimum monthly charge. The minimum monthly charge consists of the following elements:

For those rate elements that are billed a flat monthly rate, i.e., Directory Transport options, the minimum monthly charge is the monthly rate as set forth in Paragraph 9.6, following.

The minimum monthly charge for Directory Assistance Service calls is the charge as set forth in Paragraph 9.6, following, for the actual usage for the month.

(RT)

Issued:

November 17, 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri



No supplement to this Access Services Tariff tariff will be issued Section 9 except for the purpose 1st Revised Sheet 7 Replacing Original Shee of canceling this tariff. ACCESS SERVICES ED DIRECTORY ASSISTANCE SERVICE-(Continued) 9.4 Payment Arrangements-(Continued) JUN 27 1986 B. Minimum Monthly Charge MISSUURI DA Service is subject to a minimum monthly charge. The Public Service Commission charge consists of the following elements: For those rate elements that are billed a flat monthly rate, i.e., Directory Transport options, the minimum monthly charge is the monthly rate as set forth in Paragraph 9.6, following. CANCELLED The minimum monthly charge for Directory Assistance Service calls is the charge as set forth in Paragraph 9.6, following, for the DEC 161994 actual usage for the month. For the Directory Transport rate element, the minimum monthly Bharse R, S, H is assessed in terms of a Minimum Monthly Usage Charge (MMUCyblic Service Commission the Directory Transport usage charges based on the the Directory Transport usage charges based on the actual usage for MISSOURI the month are greater than the MMUC, the IC will be assessed the usage charges based on actual usage. If the Directory Transport usage charges based on the actual usage for the month are lower than the MMUC, the IC will be assessed the MMUC charge. The MMUC is determined as set forth in Paragraph 9.4, C., following. Rates for actual usage are set forth in Paragraph 9.6, following. C. Minimum Monthly Usage Charge (MMUC) The Minimum Monthly Usage Charge (MMUC) for Directory Transport is a dis-(RT) tance-sensitive charge that varies by mileage band and busy-hour minutes of (AT) capacity (BHMC) provided for the customer's use to the DA location. MMUC per mileage band equals the Minimum Directory Transport charge per BHMC times the BHMC provided. The Minimum Directory Transport charge per BHMC is as follows: Mileage Minimum Directory Transport Band Charge per BHMC(1) (CR) FNED 0 to 1 \$0.7582 (RT) (RT) 1996 JJL 1 (CR) Over 1 to 25 1.6130 (CR) Over 25 to 50 6.0732 86-84 (CR) **Over** 50 9.5170 ublic Service Commission (RT) (CT) (1) BHMC is the Directory Assistance Service busy-hour minutes of capacity provided for the customer's use to the DA location. **Issued**: Effective: 1 1986 JUL JUN 27 1986 By R. D. BARRON, President-Missouri Division

Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 9 Original Sheet 7

#### ACCESS SERVICES

9. DIRFCTORY ASSISTANCE SERVICE-(Continued)

9.4 Payment Arrangements-(Continued)

DEC 20 (11)

B. Minimum Monthly Charge

DA Service is subject to a minimum monthly charge. The minimum monthly charge consists of the following elements:

For those rate elements that are billed a flat monthly rate, i.e., Directory Transport options, the minimum monthly charge is the monthly rate as set forth in Paragraph 9.6, following.

The minimum monthly charge for Directory Assistance Service calls is the charge as set forth in Paragraph 9.6, following, for the actual usage for the month.

For the Directory Transport rate element, the minimum monthly charge is assessed in terms of a Minimum Monthly Usage Charge (MMUC). If the Directory Transport usage charges based on the actual usage for the month are greater than the MMUC, the IC will be assessed the usage charges based on actual usage. If the Directory Transport usage charges based on the actual usage for the month are lower than the MMUC, the IC will be assessed the MMUC charge. The MMUC is determined as set forth in Paragraph 9.4, C., following. Rates for actual usage are set forth in Paragraph 9.6, following.

C. Minimum Monthly Usage Charge (MMUC)

The Minimum Monthly Usage Charge (MMUC) for Directory Transport is a distance-sensitive charge that varies by mileage band and Access Connection busy-hour minutes of capacity (BHMC) provided to the DA location.

MMUC per mileage band equals the Minimum Directory Transport charge per BHMC times the BHMC provided.

The Minimum Directory Transport charge per BHMC is as follows:

	Mile	eage	Minimum Directory Transport
	Ban	<u>d</u>	Charge per BHMC(1)
	Over Over Over	8 to 16 16 to 25 25 to 50 50 to 100	GANCELLSE 026 JUL 1 19861.0670 BY 2 R 5.7 2.3965 PUBLIC SERVICE COMMISS8981 OF MISSOURI
	is the Access ocation.	Connections	s busy-hour minutes of capacity provided to the
Issued:	DEC 2 9 1983		Effective: JAN 0 1 1984 8 3 - 2 5 3
	B	Southwest	RON, Vice President-Missouri Patri Section Company ern Bell Telephone Company t. Louis. Missouri

### ACCESS SERVICES

## 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

- 9.4 Payment Arrangements-(Continued)
- D. Cancellation of a Special Order

An IC may cancel a Special Order for DA Service on any date prior to notification by the Telephone Company that service is available for the IC's use. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be canceled. The verbal notice must be followed by written confirmation within ten days.

When an IC cancels a Special Order for DA Service after the order date but prior to the start of service, the appropriate cancellation charges as set forth in Section 5, preceding, apply for the Directory Access Service canceled. In addition, a charge equal to any unrecoverable capital costs incurred by the Telephone Company will apply to the IC.

E. Changes to Special Orders

When an IC requests changes to a pending order for DA Service, such changes will be undertaken if they can be accommodated by the Telephone Company. The appropriate charges as set forth in Section 5, preceding, apply for the Directory Access Service changed. In addition, a charge equal to any other costs incurred by the Telephone Company because of the change will apply.

F. Moves

A move involves a change in the physical location of the point of termination at the IC terminal location or the IC terminal location. Moves
 (FC) will be treated as set forth in Section 6, Paragraph 6.10.9, preceding, and all associated nonrecurring charges will apply. Minimum period requirements will be established at the new location as set forth in Section 6, Paragraph 6.10.9, preceding. The IC will also remain responsible for satisfying all outstanding minimum period charges for the discontinued service.
 G. DA Service Rearrangements Nonrecurring charges apply for service rearrangements. Service rearrangements are as set forth in Section 6, Paragraph 6.10.1, C.2, preceding. The Service rearrangement Charges are as set forth in

(FC) Section 6, Paragraph 6.10.1, C.2, for the type of change provided by the Telephone Company.

Effective:

T 1	1 1
Issued:	March

n 26, 1993

April 11, 1993

93



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES



(CT)

(CT)

(CT)

(CT)

9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.4 Payment Arrangements-(Continued)

D. Cancellation of a Special Order

Public Service Commission An IC may cancel a Special Order for DA Service on any data notification by the Telephone Company that service is available for the IC's use. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days.

When an IC cancels a Special Order for DA Service after the order date but prior to the start of service, the appropriate cancellation charges as set forth in Section 5, preceding, apply for the Directory Access Service cancelled. In addition, a charge equal to any unrecoverable capital costs incurred by the Telephone Company will apply to the IC.

E. Changes to Special Orders

When an IC requests changes to a pending order for DA Service, such changes will be undertaken if they can be accommodated by the Telephone Company. The appropriate charges as set forth in Section 5, preceding, apply for the Directory Access Service changed. In addition, a charge equal to CANCELLED any other costs incurred by the Telephone Company because of the will apply.

- F. Moves
- A move involves a change in the physical location of the point **BY Service Service** (CT)lished at the new location as set forth in Paragraph 6.7.7, preceding. The IC will also remain responsible for satisfying all outstanding minimum period charges for the discontinued service.
  - G. DA Service Rearrangements

Nonrecurring charges apply for service rearrangements. Service rearrangements are as set forth in Paragraph 6.7.1, C.2, preceding. The Service rearrangement Charges are as set forth in Paragraph 6.7.1, C.2, for the type of change provided by the Telephone Company.

> 相性 1 1986 86-84 ablic Service Commission

APR 11 1993 #

Access Services Tariff

Replacing Original Sheet 8

REGE

1st Revised Sheet 8

JUN 27 1986

MISSUURI

Section 9

Issued: JUN 27 1986

Effective: JUL 1 1995

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 9 Original Sheet 8

DEC 2

#### ACCESS SERVICES

9. DIRECTORY ASSISTANCE SERVICE-(Continued) -

- 9.4 Payment Arrangements-(Continued)
- D. Cancellation of a Special Order

An IC may cancel a Special Order for DA Service on any date prior to the service date. The cancellation date is the date the <u>Telephone Company</u> in receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days.

When an IC cancels a Special Order for DA Service after the order date but prior to the start of service, the appropriate charges as set forth in Section 5, preceding, apply for the DA Service ordered. In addition, a charge equal to any unrecoverable capital costs incurred by the Telephone Company will apply to the IC.

E. Changes to Special Orders

When an IC requests changes to a pending order for DA Service, such changes will be undertaken if they can be accommodated by the Telephone Company. The appropriate charges as set forth in Section 5, preceding, apply for the DA Service ordered. In addition, a charge equal to any other costs incurred by the Telephone Company because of the change will apply.

F. Moves

A move involves a change in the physical location of the point of interface at the IC terminal location or the IC terminal location. Moves will be treated as set forth in Paragraph 6.7.7, preceding, and all associated nonrecurring charges will apply. Minimum period requirements will be established at the new location as set forth in Paragraph 6.7.7, preceding. The IC will also remain responsible for satisfying all outstanding minimum period charges for the discontinued service.

G. DA Service Rearrangements

Nonrecurring charges apply for service rearrangements. Service rearrangements are as set forth in Paragraph 6.7.1, C.3, preceding. The Service rearrangement Charges are as set forth in Paragraph 6.7.1, C.3, for the type of change provided by the Telephone Company.

GANGELLED JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSUUR JAN 0 1 1984 Effective: DEC 2 9 1983 Issued: By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 9 1st Revised Sheet 9 Replacing Original Sheet 9

#### ACCESS SERVICES

#### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.4 Payment Arrangements-(Continued)

- (CP) H. Credit Allowance for DA Service
  - When the DA location or DA operator equipment or terminals are out of service due to a Telephone Company equipment failure or an incorrect number is provided and an IC DA call has been answered or forwarded to a DA operator, a credit allowance for a call answered or forwarded to the DA operator equal to the rate for a Directory Assistance Service Call as set forth in Paragraph 9.6, following, plus the rate for a Directory Transport call will be applied to the IC's charges. The rate for a Directory Transport call will be the average of the Directory Transport rates per call as set forth in Paragraph 9.6, following.
  - 2. In addition to the credit as set forth in Paragraph 1., preceding, when a DA operator or DA equipment provides an incorrect number for a call and the customer reports such occurrences to the Telephone Company, a credit allowance for such DA call will apply. The credit will be as set forth in Paragraph 3., following. When the customer reports such a call, the number requested, the number provided and the reason the number provided is incorrect, the number of calls for which a credit will apply will be developed by the Telephone Company in cooperation with the customer.
  - 3. When a DA call is not completed due to the failure of Directory Access Service to DA locations, DA access equipment or DA operator activities, a credit allowance for the Switched Access Service portion in the originating LATA of such DA call will apply. When the customer reports such a call and DA number dialed, time of the call and the date of the call, the number of calls for which a credit will apply will be developed by the Telephone Company in cooperation with the customer. The credit will be as set forth following:

Issued:

June 27, 1986

Effective:

July 1, 1986



Access Services Tariff Section 9 Original Sheet 9

DEC 2 DEC

#### ACCESS SERVICES

- 9. DIRECTORY ASSISTANCE SERVICE-(Continued)
  - 9.4 Payment Arrangements-(Continued)
  - H. Credit Allowance for DA Service
    - 1. When the DA location or DA operator equipment or terminals are out of service due to a Telephone Company equipment failure and an IC DA call has been answered or forwarded to a DA operator, a credit allowance for a call answered or forwarded to the DA operator equal to the rate for a Directory Assistance Service Call as set forth in Paragraph 9.6, following, plus the rate for a Directory Transport call will be applied to the IC's charges. The rate for a Directory Transport call will be the average of the Directory Transport rates per call as set forth in Paragraph 9.6, following.
    - 2. Credit allowances for other service interruptions will be provided as set forth in Section 2, Paragraph 2.4.4, preceding.
    - 9.5 Rate Regulations
    - A. The Directory Assistance service call charge as set forth in Paragraph 9.6, A., following, applies for each call to DA Service. A call is a call which has been answered by or forwarded to a DA operator. The charge applies whether or not the DA operator provides the requested telephone number. The number of calls answered or forwarded to DA operators will be accumulated by Telephone Company measuring equipment.
    - B. The mileage for Directory Transport is measured from the serving wire center of the IC terminal location to the DA location. Title page notwithstanding, in the case where there is no DA location in a LATA, these two wire centers may be in different LATA's. However, the IC terminal location must be in a LATA served by the DA location. The measurement will be performed as set forth in Section 6, Paragraph 6.7.13, preceding.
    - C. The charge per call for Directory Transport, as set forth in Paragraph 9.6, B., following, applies for each call to DA Service. A call is as set forth in Paragraph 9.5, A., preceding. The number of calls will be accumulated as set forth in Paragraph 9.5. A., preceding.

JUL1  1986		
PUBLIC SERVICE COMMI	SSION	
Effective:	JAN 0 1 198	4 <b>83 - 253</b>
	Effective:	PUBLIC SERVICE COMMISSION OF MISSOURI Effective: JAN 0 1 198 , Vice President-Missouri

# ACCESS SERVICES

# 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

- 9.4 Payment Arrangements-(Continued)
- H. Credit Allowance for DA Service-(Continued)
  - 3. (Continued)

No Supplement to this

except for the purpose

of canceling this tariff.

tariff will be issued

a)	Credit per call when FGA and/or B Switched Access Service is billed	\$0.0253
b)	Credit per call when BSA-A or BSA-B Switched Access Service is billed	(CR)\$0.025284
c)	Credit per call when FGC and/or D Switched Access Service is billed	0.0269
d)	Credit per call when BSA-C or BSA-D Switched Access Service is billed	(CR)\$0.025616

4. Credit allowances for other service interruptions will be provided as set forth in Section 2, Paragraph 2.4.4, preceding.

### 9.5 Rate Regulations

- A. The Directory Assistance service call charge as set forth in Paragraph 9.6, A., following, applies for each call to DA Service. A call is a call which has been answered by or forwarded to a DA operator. The charge applies whether or not the DA operator provides the requested telephone number. The number of calls answered or forwarded to DA operators will be accumulated by Telephone Company measuring equipment. A credit for the provision of an incorrect telephone number will be applied as set forth in Paragraph 9.4, H., preceding.
- B. The mileage for Directory Transport is measured from the serving wire center of the IC terminal location to the DA location. Title page not-withstanding, these two wire centers may be in different LATA's. In addition, the premises of the ordering customer must be in the LATA where DA service is requested or in the LATA where the DA location is located. However, the IC terminal location must be in a LATA served by the DA location. The measurement will be performed as set forth in Section 6, Paragraph 6.10.15, preceding.
- C. The charge per call for Directory Transport, as set forth in Paragraph 9.6, B., following, applies for each call to DA Service. A call is as set forth in Paragraph 9.5, A., preceding. The number of calls will be accumulated as set forth in Paragraph 9.5, A., preceding.



P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 9 except for the purpose 2nd Revised Sheet 9.01 of canceling this tariff. Replacing 1st Revised Sheet 9.01 ACCESS SERVICES RECEIVED 9. DIRECTORY ASSISTANCE SERVICE-(Continued) 9.4 Payment Arrangements-(Continued) DEC 23 1993 н. Credit Allowance for DA Service-(Continued) MISSOUR) Public Service Commission 3. (Continued) a) Credit per call when FGA and/or B Switched Access Service is billed \$0.0253 b) Credit per call when BSA-A or BSA-B Switched Access Service is billed (CR)\$0.025284 c) Credit per call when FGC and/or D Switched Access Service is billed 0.0269 d) Credit per call when BSA-C or BSA-D Switched Access Service is billed (CR)\$0.025616

- 4. Credit allowances for other service interruptions will be provided as set forth in Section 2, Paragraph 2.4.4, preceding.
- 9.5 Rate Regulations

ission

URDEC 2 3 1993

- A. The Directory Assistance service call charge as set forth in Paragraph 9.6, A., following, applies for each call to DA Service. A call is a call which has been answered by or forwarded to a DA operator. The charge applies whether or not the DA operator provides the requested telephone number. The number of calls answered or forwarded to DA operators will be accumulated by Telephone Company measuring equipment. A credit for the provision of an incorrect telephone number will be applied as set forth in Paragraph 9.4, H., preceding.
- B. The mileage for Directory Transport is measured from the serving wire center of the IC terminal location to the DA location. Title page notwithstanding, these two wire centers may be in different LATA's. In addition, the premises of the ordering customer must be in the LATA where DA service is requested or in the LATA where the DA location is located. However, the IC terminal location must be in a LATA served by the DA location. The measurement will be performed as set forth in Section 6, Paragraph 6.10.15, preceding.

CANCELED<sup>The</sup> charge per call for Directory Transport, as set forth in Paragraph 9.6, B., following, applies for each call to DA Service. A call is as set forth in Paragraph 9.5, A., preceding. The number of calls will be accumulated as set forth in Paragraph 9.5, A., preceding.

> \_\_\_\_\_\_MISSOURI\_\_\_\_ Public Service Commission

Effective: JAN 0 1 1994

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 9 except for the purpose 1st Revised Sheet 9.01 of canceling this tariff. Replacing Original Sheet 9.01 ACCESS SERVICES RECEIVED 9. DIRECTORY ASSISTANCE SERVICE-(Continued) MAR 29 1993 9.4 Payment Arrangements-(Continued) MISSOURI H. Credit Allowance for DA Service-(Continued) Public Service Commission 3. (Continued) a) Credit per call when FGA and/or B Switched Access Service is billed \$0.0253 (AT) b) Credit per call when BSA-A or BSA-B Switched Access Service is billed (NR)\$0.025299 c) Credit per call when FGC and/or D Switched Access Service is billed 0.0269 (NR)\$0.0268 BANCELLED (AT) d) Credit per call when BSA-C or BSA-D Switched Access Service is billed 4. Credit allowances for other service interruptions will be provided 041994 set forth in Section 2, Paragraph 2.4.4, preceding. BY 2 1 R.S. # 9.01 The Directory Assistance service call charge as set forth in Public Service Commission 9.6, A., following, applies for each call to DA Service. A call is a set for the second by the se 9.5 Rate Regulations Α. charge applies whether or not the DA operator provides the requested telephone number. The number of calls answered or forwarded to DA operators will be accumulated by Telephone Company measuring equipment. A credit for the provision of an incorrect telephone number will be applied as set forth in Paragraph 9.4, H., preceding. The mileage for Directory Transport is measured from the serving wire В. center of the IC terminal location to the DA location. Title page notwithstanding, these two wire centers may be in different LATA's. In addition, the premises of the ordering customer must be in the LATA where DA service is requested or in the LATA where the DA location is located. However, the IC terminal location must be in a LATA served by the DA location. The measurement will be performed as set forth in Section 6, (FC) Paragraph 6.10.15, preceding. C. The charge per call for Directory Transport, as set forth in Paragraph 9.6, B., following, applies for each call to DA Service. A call is as set forth in Paragraph 9.5, A., preceding. The number of calls will be accumulated as set forth in Paragraph 9.5, A., preceding. Issued: MAR 2 6 1993 Effective: APPRILE 1993

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

92-304 MO. PUBLIC SERVICE COMM.

APR 1 1 1993

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES



9.4 Payment Arrangements-(Continued)

(CP) H. Credit Allowance for DA Service-(Continued)

3. (Continued)

Credit per call when FGA and/or B Switched Access Service is billed

Credit per call when FGC and/or D Switched Access Service is billed



CANCELLED

APR219 1993 #9.01

- 4. Credit allowances for other service interruptions will **By** browned will be browned by browned by as set forth in Section 2, Paragraph 2.4.4, preceding Public Service Commission Public Service Commission
- 9.5 Rate Regulations
- A. The Directory Assistance service call charge as set forth in Paragraph 9.6, A., following, applies for each call to DA Service. A call is a call which has been answered by or forwarded to a DA operator. The charge applies whether or not the DA operator provides the requested telephone number. The number of calls answered or forwarded to DA operators will be accumulated by Telephone Company measuring equipment. A credit for the provision of an incorrect telephone number will be applied as set forth in Paragraph 9.4, H., preceding.
- B. The mileage for Directory Transport is measured from the serving wire center of the IC terminal location to the DA location. Title page notwithstanding, these two wire centers may be in different LATA's. In addition, the premises of the ordering customer must be in the LATA where DA service is requested or in the LATA where the DA location is located. However, the IC terminal location must be in a LATA served by the DA location. The measurement will be performed as set forth in Section 6, Paragraph 6.7.13, preceding.
- C. The charge per call for Directory Transport, as set forth in Paragraph 9.6, B., following, applies for each call to DA Service. A call is as set forth in Paragraph 9.5, A., preceding. The number of calls will be accumulated as set forth in Paragraph 9.5, A., preceding.

86-84 Public Service Commission

ويحسر فبرد والمراجع

Issued:

JUN 27 1986

Effective: JUL 1 1950

Access Services Tariff Section 9 4th Revised Sheet 10 Replacing 3rd Revised Sheet 10

### ACCESS SERVICES

#### 9. DIRECTORY ASSISTANCE SERVICE-(Continued)

9.6 Rates and Charges

The rates and charges are:	Rates	
A. Directory Assistance Service call, each	(CR) \$0.2975	
B. Directory Access Service	Rates and Charges	
- Directory Access Installation Charge	Rates and charges are the same as those set forth in Section 6, in Para- (graph 6.11.1, B., preceding.	
- Directory Transport	Rate Per Call	
Call Miles		
0 to 1 Over 1 to 25 Over 25 to 50 Over 50	\$0.0028 0.0060 0.0222 0.0351	

Issued: September 22, 1994

Effective:



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

9.

Access Services Tariff Section 9 4th Revised Sheet 10 Replacing 3rd Revised Sheet 10

Rates and Charges

Rate Per Call

#### ACCESS SERVICES

# RECEIVED

DEC 23 1993

MISSOURI Public Service Commission

9.6	Rates and	Charges	
	The rates	and charges are:	

DIRECTORY ASSISTANCE SERVICE-(Continued)

A. Directory Assistance Service call, each . . . . . . . . . . . . . . . . . . (CR) \$0.2975

B. Directory Access Service

- Directory Transport

Call Miles

0 to 1	\$0.0028
Over 1 to 25	0.0060
Over 25 to 50	0.0222
Over 50	0.0351

CANCELLED

1994 Public Service Commission BY MISSOURI

**.** 

FILED

JAN 11994 93-224 MISSOURL

Issued: DEC 2 3 1993

Effective: JAN 0 1 1994 blic Service Commission

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 9 except for the purpose 3rd Revised Sheet 10 Replacing 2nd Revised Sheet 10 HECEIVED of canceling this tariff. ACCESS SERVICES MAR 23 1993 9. DIRECTORY ASSISTANCE SERVICE-(Continued) MISSOURI Public Service Commission 9.6 Rates and Charges Rates The rates and charges are: **Directory Assistance** Α. Service call, each . . . . . . . . . . . . . . . \$0.4465 Rates and Charges B. Directory Access Service - Directory Access Rates and charges are the Installation Charge. . . . . . . . . . .same as those set forth in Section 6, in Para-(FC) graph 6.11.1, B., preceding. Rate Per Call - Directory Transport Call Miles 0 to 1..... \$0.0028 0.0060

CANCELLED

0.0222

0.0351

OCT 0 1 1994 UR R S, TO Public Service Commission MISSOURI

**FILED** 

APR 1 1 1993 92 - 304 MO. PUBLIC SERVICE COMM.

Issued:

Effective:

MAR 2 6 1993 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

# APR 1 1 1993

tariff vi except fo	ement to this 11 be issued or the purpose ling this tariff.	Access Services Tariff Section 9 2nd Revised Sheet 10 Replacing 1st Revised Sheet 10
	ACCESS SERVICES	RECEIVED
9. DIREC	TORY ASSISTANCE SERVICE-(Continued)	APR 25 1991
	es and Charges rates and charges are:	MISSOURI Public Service Commission
Α.	Directory Assistance Service call, each	(CR) \$0.4465
В.	Directory Access Service	Rates and Charges
	- Directory Access Installation Charge	Rates and charges are the same as those set forth in Section 6, in Paragraph 6.8.1, B., preceding.
	- Directory Transport	Rate Per Call
	Call Miles	
	0 to 1	· · · · 0.0060 · · · · 0.0222

CANCELLED

\_\_\_\_

APR 11 1993 BY <u>3<sup>MA</sup> R.S.</u> Public Service Commission MISSOURI

**FILED** 

MAY 28 1991 **9 1 - 3 5 1** Public Service Commission

Issued: APR \$ 6 1991

- -----

Effective: MAY 2 8 1991
P.S.C. Mo.-No. 36

-

- · · · ·

rana a sama ar

	tarif excep	pplement to this f will be issued t for the purpose nceling this tariff.		Access Services Tariff Section 9 lst Revised Sheet 10 Replacing Original Sheet 10
			ACCESS SERVICES	REGEIVED
	9. D	IRECTORY ASSISTANCE SER	VICE-(Continued)	
	9.6	Rates and Charges		JUN 2 7 1986
		The rates and charges	are:	MISSLAAKEs
(CR)	Α.	Directory Assistance Service call, each		Public Service Commission
	В.	Directory Access Servi	ce	Rates and Charges
(CT)		- Directory Access Installation Charge.	· · · · · · · · · · · · · ·	Rates and charges are the
(TT)				
		- Directory Transport		Rate Per Call
		Call Miles		
(CR) (RT)		0 to 1	· · · · · · · · · · · · · ·	. \$0.0028
(RT) (CR) (CR) (CR) (RT) (DR)		Over 25 to 50	• • • • • • • • • • • • • • • •	. 0.0222
			MAY 28 1991 BYJ Commission Ublic Service Commission MISSOURI	FILED FUL 1 1986 86 - 84 Public Service Commission
	Issue	ed: JUN 27 1986	Effective:	JUL 1 JUL 1 1986.
			BARRON, President-Misso thwestern Bell Telephone St. Louis, Missouri	e Company

\_\_\_\_\_

Ρ	•	S	•	С	•	Mo	:	No	•	- 36

··--- · ·

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

- - -

.-

----

Access Services Tariff Section 9 Original Sheet 10

. ....

#### ACCESS SERVICES

9. DI	RECTORY ASSISTANCE SERVICE-(Continued)			
9.6	Rates and Charges		-	DEC 201113
	The rates and charges are:			Rates
	Directory Assistance Service call, each	••••		\$0.5604
в.	Directory Access Service		Rates	and Charges
	- Switched Access Service Access Connection		same as th in Section graph 6.8 for the Ad	charges are the hose set forth n 6, in Para- .1, preceding, ccess Connec- options ordered.
	- Directory Transport		Rate	e Per Call
	Call Miles         0 to 1	· · · ·	-	\$0.0023 0.0052 0.0060 0.0064 0.0104 0.0145 0.0231
	Options	USOC	Monthly <u>Rates</u>	Nonrecurring Charges
-	Provision of other than Telephone Company-Selected Traffic Routing - Direct Trunking in lieu of Tandem Trunking			
	Per Directory Access Service Transmission Path - Tandem Trunking in lieu of Direct Trunking PANNELLE Per Directory Access Service	lrtd+	ICB rates a	nd charges apply
	Transmission Path JUL 1 1986	lrtt+	ICB rates a	nd charges apply
<u>.</u>	BY A RS. 10 PUBLIC SERVICE COMMISSION	ÚN	in il i LAL i	」に引 -1 (SE)
Issued	1: DEC 2 9 1983 Effect:	ive: JA	N 0 1 1984 8	<b>3 - 2 5 3</b>
	By R. D. BARRON, Vice Pres Southwestern Bell Teleph			

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES

### 10.1 General

This Section covers Special Access Services that are provided to an IC(l) for use only by agencies or branches of the Federal Government and other users authorized by the Federal Government. In addition, this section covers the Telecommunications Service Priority (TSP) System procedures as set forth in this section and administered by the National Communications System (NCS). Services provided to state emergency

operations centers are included. These services provide for command and control communications, including communications for national security, emergency preparedness and presidential requirements. They are required to assure continuity of Government in emergency and crisis situations and to provide for national security.

Services for command and control communications and for national security and emergency preparedness sometimes require short-notice and shortduration service provisions. These provisions are especially needed to meet presidential requirements or in response to natural, man-made or declared emergencies. Requirements of this type cannot be forecasted and are usually needed for a relatively short period. The provision of service under these conditions may require the availability of facilities, such as portable microwave equipment, which are provided on a temporary basis by the Telephone Company, IC or End User.

10.2 Emergency Conditions

These services will be provided on the date requested or as soon as possible thereafter when the emergency falls into one of the following categories:

- State of crisis declared by the National Command Authorities (includes commitments made to the National Communications System in the "National Plan for Emergencies and Major Disasters").

(1) The Federal Government may be an IC for the provision of intrastate telecommunications services for itself or for others.

Issued: July 3, 1991

Effective: August 2, 1991



By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

(AT)

Access Services Tariff Section 10 Original Sheet 1

DEC 2

#### ACCESS SERVICES

#### 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES

10.1 General

This Section covers Special Access Services that are provided to an IC(1) for use only by agencies or branches of the Federal Government and other users authorized by the Federal Government. Services provided to state emergency operations centers are included. These services provide for command and control communications, including communications for national security, emergency preparedness and presidential requirements. They are required to assure continuity of Government in emergency and crisis situations and to provide for national security.

Services for command and control communications and for national security and emergency preparedness sometimes require short-notice and shortduration service provisions. These provisions are especially needed to meet presidential requirements or in response to natural, man-made or declared emergencies. Requirements of this type cannot be forecasted and are usually needed for a relatively short period. The provision of service under these conditions may require the availability of facilities, such as portable microwave equipment, which are provided on a temporary basis by the Telephone Company, IC or End User.

10.2 Emergency Conditions

These services will be provided on the date requested or as soon as possible thereafter when the emergency falls into one of the following categories:

- State of crisis declared by the National Command Authorities (includes commitments made to the National Communications System in the "National Plan for Emergencies and Major Disasters").

			(	CANCELLED
			E Public	AUG 2 1991 BY <u>IALRS</u> #1 Service Commission MISSOURI
		ent may be an IC for the provision services for itself or for others		rastate
Issued:	DEC 2 9 1983	Effective: JA	AN 0 1 198	4 JAN - 1 (SO) 8 3 - 2 5 3
	-	R. D. BARRON, Vice President-Mis		Palette Service Commission

# ACCESS SERVICES

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.2 Emergency Conditions-(Continued)

- Efforts to protect endangered U.S. personnel or property both in the U.S. and abroad. (Includes space vehicle recovery and protection efforts.)
- Communications requirements resulting from hostile action, a major disaster or a major civil disturbance.
- The director (Cabinet level) of a Federal department, Commander of a Unified/Specified Command or head of a military department has certified that a communications requirement is so critical to the protection of life and property or to the National Defense that it must be processed immediately.
- Political unrest in foreign countries which affect the national interest.
- Presidential service.
- 10.3 Intervals to Provide Service

Certain services provided under the provisions of this Section of the Tariff are provided on an individual case basis. Orders for these services shall be placed under the Interval provisions set forth in Paragraph 5.2.1, preceding.

(L) CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Issued: July 3, 1991

Effective: August 2, 1991



P.S.C. Mo.-No. 36

\_.

		lement to this	Ac	cess Services Tariff
		will be issued		Section 10
		for the purpose eling this tariff.	Replac	lst Revised Sheet 2 ing-Original Sheet-2
	or cane	ACCESS SERVICES	Neptac	REGEIVED
	10. SP	ECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Con	tinued)	JUN 27 1986
	10.2			MISSUURI
		- Efforts to protect endangered U.S. personne U.S. and abroad. (Includes space vehicle s efforts.)	el or pr recovery	PRATHY Cethed Prothenission
		- Communications requirements resulting from disaster or a major civil disturbance.	hostile	action, a major
		- The director (Cabinet level) of a Federal Unified/Specified Command or head of a mili tified that a communications requirement is tection of life and property or to the Nati be processed immediately.	itary de s so cri	partment has cer- tical to the pro-
		- Political unrest in foreign countries which	h affect	the national interest.
		- Presidential service.		
	10.3	Intervals to Provide Service	•	Таба стала стал
CT) CT) CT)		Services provided under the provisions of the are provided on an individual case basis. On be placed under the Interval provisions set preceding.	rders fo	r these services shall
(RT)				
		CANCEL	LED	
			1091 .	
		AUG 2 BY 2ml (	5 5 2	
		BY 2 Sublic Service	Commis	
		Public Service MISSC	JURI	FALED
		1410-		<u>1111</u> 1986
				1
				<b>86-84</b> Public Service Commission
				Service of the state of the service

sued: JUN 2 7 1986

Effective: JUL 1 1986

1000

Access Services Tariff Section 10 Original Sheet 2

DEC 20123

#### ACCESS SERVICES

10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

- 10.2 Emergency Conditions-(Continued)
  - Efforts to protect endangered U.S. personnel or property both in the on U.S. and abroad. (Includes space vehicle recovery and protection efforts.)
  - Communications requirements resulting from hostile action, a major disaster or a major civil disturbance.
  - The director (Cabinet level) of a Federal department, Commander of a Unified/Specified Command or head of a military department has certified that a communications requirement is so critical to the protection of life and property or to the National Defense that it must be processed immediately.
  - Political unrest in foreign countries which affect the national interest.
  - Presidential service.
- 10.3 Intervals to Provide Service

Government requirements for short-notice and short-duration services do not fit the two categories of normal ordering options (1) Access Order and (2) Planned Facilities Order. Orders for such services may be placed under the short-notice provisions set forth in Section 5, Paragraph 5.2.1, C., preceding.

10.4 Special Facilities Routing

The regulations, rates and charges governing the provision of Special Facilities Routing are set forth in Section 11, following.

GANGELLED JUL 1 1986 PUBLIC SERVICE COMMISSION

Effective: JAN 0 1 1984

1. 1. 5.

문자가 = 1 (SB )

83 - 253

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Issued: DEC 2 9 1983

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.4 Safeguarding of Service

10.4.1 Facility Availability

In order to insure communications during periods of emergency, the Telephone company will, within the limits of good management, make available the necessary facilities to restore service in the event of damage or to provide temporary emergency service, as set forth in

Paragraphs 10.7.1, C. and 10.7.5, C., following.

In order to meet the requirements of agencies or branches of the Federal Government, the Telephone Company may utilize governmentowned facilities, when necessary to provide service.

### 10.5 Federal Government Regulations

In accordance with Federal Government Regulations, all service provided to the Federal Government as an IC will be billed in arrears. However, this provision does not apply to other customers that obtain services under the provisions of this Tariff to provide their services to the Federal Government.

### 10.6 Quotation Charges

In order to comply with Federal regulations, quotation preparation charges will not apply when submitting unsolicited quotes or when submitting quotes in response to a general Request For Proposal or Invitation to Bid from agencies or branches of the Federal government as an IC. However, a charge for quotation, as set forth in Section 2, Paragraph 2.4.3, preceding, will apply in all other cases.



Access Services Tariff Section 10 lst Revised Sheet 3 Replacing Original Sheet 3

### ACCESS SERVICES



10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.4 Safeguarding of Service

10.4.1 Facility Availability

JUN 2 7 1986 MISSUURI

Public Service Commission

In order to insure communications during periods of emergency, the Telephone Company will, within the limits of good management, make available the necessary facilities to restore service in the event of damage or to provide temporary emergency service.

In order to meet the requirements of agencies or branches of the Federal Government, the Telephone Company may utilize governmentowned facilities, when necessary to provide service.

(FC) 10.5 Federal Government Regulations

(RT)

(AT)

TFC)

In accordance with Federal Government Regulations, all service provided to the Federal Government as an IC will be billed in arrears. However, this provision does not apply to other customers that obtain services under the provisions of this Tariff to provide their services to the Federal Government.

#### 10.6 Quotation Charges

In order to comply with Federal regulations, quotation preparation charges will not apply when submitting unsolicited quotes or when submitting quotes in response to a general Request For Proposal or Invitation to Bid from agencies or branches of the Federal Government as an IC. However, a charge for quotation, as set forth in Section 2, Paragraph 2.4.3, preceding, will apply in all other cases.

		CANCELLED	
		AUG 2 1991 BY 2 2 K.S. 5.	FALED
٠		BY 2 M. N.S. Public Service Commission MISSOURI	Public Service Commission
	Issued: JUN 27 1986	Effective: JUL 1	1936
	By R. S	D. BARRON, President-Missouri Divi outhwestern Bell Telephone Company St. Louis, Missouri	lsion /

Access Services Tariff Section 10 Original Sheet 3

DEC 29 1233

#### ACCESS SERVICES

#### 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.5 Safeguarding of Service

10.5.1 Restoration Priority

The regulations and nonrecurring charges governing restoration priority are set forth in Section 13, Paragraph 13.3.2, following.

#### 10.5.2 Facility Availability

In order to insure communications during periods of emergency, the Telephone Company will, within the limits of good management, make available the necessary facilities to restore service in the event of damage or to provide temporary emergency service.

In order to meet the requirements of agencies or branches of the Federal Government, the Telephone Company may utilize governmentowned facilities, when necessary to provide service.

#### 10.6 Federal Government Regulations

Government Procurement Regulations, Defense Acquisition Regulations, Federal Procurement Regulations, Basic Agreements, Federal Agency Directives, as well as Presidential Directives will be followed when providing service to the Federal Government as an IC.

In accordance with Federal Government Regulations, all service provided to the Federal Government as an IC will be billed in arrears.

#### 10.7 Quotation Charges

In order to comply with Federal regulations, quotation preparation charges will not apply when submitting unsolicited quotes or when submitting quotes in response to a general Request For Proposal or Invitation to Bid from agencies or branches of the Federal Government as an IC. However, a charge for quotation, as set forth in Section 2, Paragraph 2.4.3, preceding, will apply in all other cases.

JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

4 UAM = 1 (53.) 8 3 - 2 5 3 Public Service Commission

# ACCESS SERVICES

### 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.7 Service Offerings To The Federal Government

The following unique services are provided to an IC for use only by agencies or branches of the Federal Government, other authorized users and state emergency operations centers. The rates and charges for certain services shall be developed on an individual case basis and shall be consistent with the rates and charges for services offered in other sections of this tariff.

- 10.7.1 Type and Description
  - A. Voice Grade Special Access Service
    - 1. Voice Grade Secure Communications Type I

Approximate bandwidth of 10 - 50,000 Hz. Furnished for twopoint secure communications on two-wire or four-wire metallic facilities between an IC terminal location and an End User's premises. Services are conditioned as follows:

T-3 Conditioning - The absolute loss (referenced to one milliwatt) with respect to frequency shall not exceed:

> 15 dB at 10 Hz 13 dB at 100 Hz 9 dB at 1,000 Hz 20 dB at 10.000 Hz 30 dB at 50,000 Hz

Issued: July 3, 1991



No supplement to this
tariff will be issued
except for the purpose
of canceling this tariff.

Access	Services	s Tariff
	Sec	tion 10
lst	Revised	Sheet 4
Replacing (	Driginal	Sheet 4

REGEIVED

JUN 27 1986

ACCESS SERVICES

10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued

- (FC) 10.7 Service Offerings To The Federal Government
- (CT) The following unique services are provided to an IG for use only by agencies or branches of the Federal Government, other authorized users.
   (AT) and state emergency operations centers. The rates and charges for ministration services shall be developed on an individual case basis and shall be consistent with the rates and charges for services offered in other
- (FC)

(C)

(C)

10.7.1 Type and Description

sections of this Tariff.

- A. Voice Grade Special Access Service
  - 1. Voice Grade Secure Communications Type I

Approximate bandwidth of 10 - 50,000 Hz. Furnished for twopoint secure communications on two-wire or four-wire metallic facilities between an IC terminal location and an End User's premises. Services are conditioned as follows:

T-3 Conditioning - The absolute loss (referenced to one milliwatt) with respect to frequency shall not exceed:

15 dB at 10 Hz 13 dB at 100 Hz 9 dB at 1,000 Hz 20 dB at 10,000 Hz 30 dB at 50,000 Hz

CANCELLED

AUG 2 1991 BY 2 R.S 4 Public Service Commission MISSOURI



Issued: JUN 27 1985

Effective: JUL 1 1986

Access Services Tariff Section 10 Original Sheet 4

#### ACCESS SERVICES

SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

DEC 2 9 1223

10.8 Service Offerings To The Federal Government

The following services are provided to an IC only for agencies or branches of the Federal Government, other authorized users and state emergency operations centers:

10.8.1 Type and Description

A. Voice Grade Special Access Service

1. Voice Grade Secure Communications Type I

Approximate bandwidth of 10 - 50,000 Hertz. Furnished for twopoint secure communications on two-wire or four-wire metallic facilities between an IC terminal location and an End User's premises. Services are conditioned as follows:

T-3 Conditioning - The absolute loss (referenced to one milliwatt) with respect to frequency shall not exceed:

15 db at 10 Hz 13 db at 100 Hz 9 db at 1,000 Hz 20 db at 10,000 Hz 30 db at 50,000 Hz

GANGELLED JUL 1 1986

OF MISSUUR

JAN 0 1 1984 J

1 fed

258

PUBLIC SERVICE CON

Effective:

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company

Access Services Tariff Section 10 1st Revised Sheet 5 Replacing Original Sheet 5

# ACCESS SERVICES

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

- (FC) 10.7 Service Offerings to the Federal Government-(Continued)
- (FC) 10.7.1 Type and Description-(Continued)
  - A. Voice Grade Special Access Services-(Continued)
  - 1. Voice Grade Secure Communications Type I-(Continued)

Additional conditioning (available in one or two directions on four-wire facilities only) to provide the following characteristics:

The absolute loss (referenced to one milliwatt) with respect to frequency shall not exceed:

0 dB at 1,000Hz <u>+</u> 1 dB between 1,000 Hz and 40,000 Hz <u>+</u> 2 dB between 10 Hz and 50,000 Hz (+ means more loss)

The net loss of the conditioned service (with or without additional conditioning) shall not vary by more than 4 dB at 1,000 Hz from the levels specified above. Voice frequency signaling or supervisory tones can be transmitted.

2. Voice Grade Secure Communications Type II

Approximate bandwidth 10 - 50,000 Hz. Furnished on four-wire metallic facilities for duplex operation for two-point secure communication between an IC terminal on an End User's premises and an End User's premises. Services are conditioned as follows:

G-1 Conditioning. - The absolute loss with respect to frequency and the net loss variation shall be the same as Voice Grade Secure Communications Type I services without additional conditioning. Voice frequency signaling or supervisory tones can be transmitted.

Issued:	June 27, 1986	Effective:	July 1, 1986	

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Ρ.	.s.	С.	Mo.	-No.	36
----	-----	----	-----	------	----

Access Services Tariff Section 10 Original Sheet 5

#### ACCESS SERVICES

	CIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)
	3.1 Type and Description-(Continued)
Α.	Voice Grade Special Access Services-(Continued)
	1. Voice Grade Secure Communications Type I-(Continued)
	Additional conditioning (available in one or two directions on four-wire facilities only) to provide the following characteristics:
	The absolute loss (referenced to one milliwatt) with respect to frequency shall not exceed:
	0 db at 1,000 Hz + 1 db between 1,000 Hz and 40,000 Hz + 2 db between 10 Hz and 50,000 Hz (+ means more loss)
	The net loss of the conditioned service (with or without additional conditioning) shall not vary by more than 4 db at 1,000 Hz from the levels specified above. Voice frequency signaling or super-visory tones can be transmitted.
	2. Voice Grade Secure Communications Type II
	Approximate bandwidth 10 - 50,000 Hz. Furnished on four-wire metalli facilities for duplex operation for two-point secure communication between an IC terminal on an End User's premises and an End User's premises. Services are conditioned as follows:
	G-1 Conditioning - The absolute loss with respect to frequency and the net loss variation shall be the same as Voice Grade Secure Communications Type I services without additional conditioning. Voice frequency signaling or supervisory tones, can be transmitted.
	GANGELLED
	JUL 1 1986
	PUBLIC SERVICE AND THE SE
Issued:	DEC 2 9 1983 Effective: JAN 0 1 1984 83 - 253

By R. D. BARRON, Vice President-Missouri Public Service Campaission Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 10 1st Revised Sheet 6 **Replacing Original Sheet 6** 

# ACCESS SERVICES

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

- (FC) 10.7 Service Offerings to the Federal Government-(Continued)
- (FC) 10.7.1 Type and Description-(Continued)
  - A. Voice Grade Special Access Services-(Continued)
    - 3. Voice Grade Secure Communications Type III

Approximate bandwidth 10-50,000 Hz. Furnished on four-wire metallic facilities for duplex operation for two-point secure communication between an IC terminal location switch and an End User's premises. Services are conditioned as follows:

G-2 Conditioning - The absolute loss with respect to frequency and the net loss variation from the switch to an End User's premises shall be the same as Voice Grade Secure Communications Type I services without additional conditioning: from an End User's premises to the switch shall be the same as Voice Grade Secure Communications Type I services with additional conditioning. Voice frequency signaling or supervisory tones can be transmitted.

4. Voice Grade Secure Communications Type IV

Approximate bandwidth 10 - 50,000 Hz. Furnished on four-wire metallic facilities for duplex operation for two-point secure communication between two IC terminal location switches. Services are conditioned as follows:

G-3 Conditioning - The absolute loss with respect to frequency and the net loss variation shall be the same in both directions of transmission as Voice Grade Secure Communications Type I services with additional conditioning. Voice frequency signaling or supervisory tones can be transmitted.

Issued:

June 27, 1986



Access Services Tariff Section 10 Original Sheet 6

DEC 25 1003

#### ACCESS SERVICES

10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.8 Service Offerings To The Federal Government-(Continued)

10.8.1 Type and Description-(Continued)

- A. Voice Grade Special Access Services-(Continued)
  - 3. Voice Grade Secure Communications Type III

Approximate bandwidth 10 - 50,000 Hz. Furnished on four-wire metallic facilities for duplex operation for two-point secure communication between an IC terminal location switch and an End User's premises. Services are conditioned as follows:

G-2 Conditioning - The absolute loss with respect to frequency and the net loss variation from the switch to an End User's premises shall be the same as Voice Grade Secure Communications Type I services without additional conditioning; from an End User's premises to the switch shall be the same as Voice Grade Secure Communications Type I services with additional conditioning. Voice frequency signaling or supervisory tones can be transmitted.

4. Voice Grade Secure Communications Type IV

Approximate bandwidth 10 - 50,000 Hz. Furnished on four-wire metallic facilities for duplex operation for two-point secure communication between two IC terminal location switches. Ser-vices are conditioned as follows:

G-3 Conditioning - The absolute loss with respect to frequency and the net loss variation shall be the same in both directions of transmission as Voice Grade Secure Communications Type I services with additional conditioning. Voice frequency signaling or supervisory tones can be transmitted.

加加加加

JUL1 19986 PUBLIC SERVICE COMMISSION

OF MISSOURI

JAN 0 / 1984 JACI

1 1934 - 2 5 3

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

Effective:

### ACCESS SERVICES 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

# 10.7 Service Offerings to the Federal Government-(Continued)

- 10.7.1 Type and Description-(Continued)
  - B. Wideband Digital Special Access Service

Service arrangement for secured communications to accommodate the transmission of binary digital baseband signals in a random polar format.

- 1. Wideband Secure Communications Type I For transmission at the rate of 18,750 bits per second.
- 2. Wideband Secure Communications Type II For transmission at the rate of 50,000 bits per second.
- 3. Wideband Secure Communications Type III

To accommodate the transmission of restored polar two-level facsimile signals with a minimum signal element width of twenty microseconds at a rate of 50,000 bits per second.

# C. <u>Telecommunications Service Priority (TSP) System</u>

1. The TSP System is a service, developed to meet the requirements of
the Federal Government, which provides the regulatory,
administrative, and operational framework for the priority
installation and/or restoration of National Security Emergency
Preparedness (NSEP) telecommunications services. The TSP System
is available for both Switched and Special Access Services. The
TSP System applies only to NSEP telecommunications services, and
requires and authorizes priority action by the Telephone Company
providing such services.

The TSP System applicability is limited to Access Services that the Telephone Company can discretely identify for priority provisioning and restoring.

In addition, TSP System service shall be provided in accordance with the "Telecommunications Service Priority (TSP) System for National Security Emergency Preparedness (NSEP) Service" and the "Telecommunications Service Priority (TSP) System for National Security Emergency Preparedness (NSEP) Service."

### Issued: J

July 3, 1991

#### Effective: August 2, 1991

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



(AT)

P.S.C. Mo.-No. 36

.

\_

.

\_ \_

No supplement to this tariff will be issued except for the purpose of canceling this tariff.	Access Services Tariff Section 10 2nd Revised Sheet 7 Replacing 1st Revised Sheet 7
ACCESS SERVICES	RECEIVED
10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Co	ontinued) SEP 2.5 1989
10.7 Service Offerings to the Federal Government	
10.7.1 Type and Description-(Continued)	Public Service Commission
B. Wideband Digital Special Access Service	
Service arrangement for secured communication transmission of binary digital baseband signation	
1. Wideband Secure Communications Type I	
For transmission at the rate of 18,750 bi	its per second.
2. Wideband Secure Communications Type II	CANCELLED
For transmission at the rate of 50,000 bi	Its per second AUG 2 1991 BY $3^{4}$ R.S. $\frac{4}{7}$
3. Wideband Secure Communications Type III	Public Service Commission
To accommodate the transmission of restor signals with a minimum signal element wice a rate of 50,000 bits per second.	red polar two-level facsimile
10.7.2 Mileage Application	
Mileage, when used for rate application centers, shall be determined by the V & forth in Section 2, Paragraph 2.7, pred (F() forth in Section 7, Paragraph 7.3.6, pr	& H Coordinates Method as set ceding and administered as set
10.7.3 Moves	
(FC) The provisions set forth in 7.3.5 apply provided in this Section.	y to moves involving services
10.7.4 Order Cancellation	
A customer may cancel an order for Spec Service at any time prior to notificat that service is available for the custo	ion by the Telephone Company
The regulations and charges set forth the cancellation of Special Federal Gov	
Issued: SEP 2 5 1989 Effective	
By R. D. BARRON, President-Misso Southwestern Bell Telephone St. Louis, Missouri	Company Public Service Commission

P.S.C. Mo.-No. 36

	tariff except	lement to this will be issued for the purpose eling this tariff. ACCESS SERVICES	lst Rev Replacing Orig	And a state of the
	10. SP	ACCESS SERVICES ECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Conti	nued)	EGEIVED
(EC):	10.7	Service Offerings to the Federal Government-(C	ontinued)	JUN 27 1986
(FC)	10.	7.1 Type and Description-(Continued)		MISSUIRI
	B. Wi	deband Digital Special Access Service	Publi	c Service Commission
	th	rvice arrangements for secured communications t e transmission of binary digital baseband signa lar format.		1
	1.	Wideband Secure Communications Type I		
(°C')		For transmission at the rate of 18,750 bits p	er second.	CANCELLED
	2.	Wideband Secure Communications Type II		
		For transmission at the rate of 50,000 bits p	er second.	OCT 1 1989 Big RS # 7
_	3.	Wideband Secure Communications Type III	Philo	the Constant Color Philasters
		To accommodate the transmission of restored p facsimile signals with a minimum signal eleme twenty microseconds at a rate of 50,000 bits	olar two-leve: nt width of	
		To accommodate the transmission of binary dig signals in a random polar format at the rate per second.		3
(FC)	10.7	.2 Mileage Application		
(AT)		Mileage, when used for rate application bet centers, shall be determined by the V & H C set forth in Section 2, Paragraph 2.7, prec as set forth in Section 7, Paragraph 7.4.6,	oordinates Met eding and adm:	thod as
(141)	10.7	.3 Moves		ISAN ISM
		The provisions set forth in 7.4.5 apply to provided in this Section.	moves involvin	ng services
(AT)	10.7	.4 Order Cancellation		86-84
lacksquare		A customer may cancel an order for Special Service at any time prior to notification b that service is available for the customer'	Federal Govern y the Telephon	
		The regulations and charges set forth in 5. the cancellation of Special Federal Governm		
	Issued:	JUN 27 1986 Effective:	JUL 1 1986	
		By R. D. BARRON, President-Missou Southwestern Bell Telephone St. Louis, Missouri		

Contra C

× .

5 A.S.S.

101

10 A.

Access Services Tariff Section 10 Original Sheet 7

DEC 2 9 (233

#### ACCESS SERVICES

10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continuted)

10.8 Service Offerings to the Federal Government-(Continued)

10.8.1 Type and Description-(Continued)

B. Wideband Digital Special Access Service

Service arrangements for secured communications to accommodate the transmission of binary digital baseband signals in a random polar format.

1. Wideband Secure Communications Type I

For transmissions at the rate of 18,750 bits per second.

2. Wideband Secure Communications Type II

For transmission at the rate of 50,000 bits per second.

3. Wideband Secure Communications Type III

To accommodate the transmission of restored polar two-level facsimile signals with a minimum signal element width of twenty microseconds at a rate of 50,000 bits per second.

To accommodate the transmission of binary digital baseband signals in a random polar format at the rate of 50,000 bits per second.

#### 10.8.2 Mileage Application

Mileage, when used for rate application between two serving wire centers, shall be determined by the V & H Coordinates Method as set forth in Section 2, Paragraph 2.7, preceding.

GANGELLED JUL 1 1986 RY ARST JAN 0 1984 JAN - 1 150 Issued: DEC 2 9 1983 Effective: 83 - 253 By R. D. BARRON, Vice President-Missouri Public Service Commission Southwestern Bell Telephone Company

St. Louis, Missouri

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.7 Service Offerings to the Federal Government-(Continued)

10.7.1 Type and Description-(Continued)

- C. <u>Telecommunications Service Priority (TSP) System</u>-(Continued)
  - Some of the rate elements required for the TSP System are included in other sections of this tariff as general service offerings. This section makes reference to them for regulations, rates, and charges in order to reflect the complete TSP System.
  - 3. The customer for TSP System Service also must be the same customer for the Access Service with which it is associated.
  - 4. Under certain conditions it may be necessary to preempt one or more customer services with a lower or no restoration priority in order to install or restore NSEP telecommunications service(s) of a higher priority. If such preemption is necessary, and if circumstances permit, the Telephone Company will make reasonable effort to notify the preempted service customer of the action being taken. Credit allowance for such service preemption shall be made in accordance with the provisions set forth in Section 2, Paragraph 2.4.4, E., preceding, concerning Temporary Surrender of a Service.
  - 5. The customer, in obtaining TSP System service, acknowledges and consents to the provision of certain customer service record information by the Telephone Company to the National Communications System (NCS) in order for the NCS to maintain and administer its overall TSP System. This customer service record information will include only TSP Authorization Code and Telephone Company Circuit/Service ID.
  - 6. When TSP is revoked, or discontinued, and the associated Access Service is continued in service, no charge applies for such a discontinuance.
  - 7. Credit allowance for service interruption for Telecommunications Service Priority shall be the same as for the Access Service with which it is associated as set forth in Section 2, Paragraph 2.4.4, preceding.

Issued:

(AT)

July 3, 1991

Effective: August 2, 1991



# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

- 10.7 Service Offerings to the Federal Government-(Continued)
  - 10.7.1 Type and Description-(Continued)
- C. <u>Telecommunications Service Priority (TSP) System-(Continued)</u>
  - 8. Certain activities associated with the TSP System performed by the Telephone Company are included in the following rate elements:
    - a. Priority Installation (Provisioning) The act of supplying telecommunications service to a customer, including all associated transmission, wiring, and equipment, if provided by the Telephone Company, at an earlier time than standard order intervals would allow.
    - b. Priority Restoration Level Implementation (Assignment) The act of designating the priority level for the restoration of a particular NSEP telecommunications service.
    - c. Priority Restoration Level Change The act of changing the priority level assignment for an NSEP telecommunications service. This includes any extension of an existing priority level assignment to an expanded NSEP service.
    - d. Priority Restoration Administration and Maintenance The act of administering and maintaining the TSP system in such a manner that it corresponds to the National Communications System data base.
  - 9. When performing Priority Restoration (Repair), and/or Priority Installation, of an Access Service, the Telephone Company will attempt to notify the customer regarding certain Access Services where additional labor charges apply, as set forth in Section 13, Paragraph 13.2.6, following, before the required additional labor is undertaken. The customer, in obtaining a Priority Restoration, recognizes that quoting charges and obtaining permission to proceed with the restoration of certain Access Services may cause unnecessary delays and, as a result, would be contrary to the aforementioned Rules and Regulations.

In subscribing to TSP Service, the customer recognizes this condition and grants the Telephone Company the right to quote charges after the restoration or installation has been completed.

July 3, 1991

Effective: August 2, 1991





# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.7 Service Offerings to the Federal Government-(Continued)

### (MT) 10.7.2 Mileage Application

Mileage, when used for rate application between two serving wire centers, shall be determined by the V & H Coordinates Method as set forth in Section 2, Paragraph 2.7, preceding, and administered as set forth in Section 7, Paragraph 7.3.6, preceding.

#### 10.7.3 Moves

The provisions set forth in Section 7, Paragraph 7.3.5, preceding, apply to moves involving services provided in this Section.

10.7.4 Order Cancellation

A customer may cancel an order for Special Federal Government Access Service at any time prior to notification by the Telephone Company that service is available for the customer's use.

The regulations and charges set forth in Section 5, Paragraph 5.2.3, preceding, apply for the cancellation of Special Federal Government Access Services.

### 10.7.5 Rates and Charges

### A. Voice Grade Special Access Service

The provision of T-3 and G conditioned services contemplates station and tandem switching operations, using IC- or End User-provided equipment, as well as special access service. Separate narrowband or voice grade services, where required by the IC- or End Userprovided equipment or switching operation, are furnished in accordance with the applicable sections of this Tariff.

٢		Voice Grade Secure			Monthly	Nonrecurring
CANC		<u>Communications</u>	USOC	· · · · · · · · · · · · · · · · · · ·	Rates	Charges
		Type I, each T-3 Conditioning	GCA		ICB 1	ates and charges apply.
		Additional Conditioning, per service termination.	GTO		I CB	rates and charges apply.
(MT)		Type II, each G-l Conditioning	GCB		ICB r	ates and charges apply.
× / _	Issued:	July 3, 1991		Effective:	August 2	2, 1991

July 3, 1991 Effective: August 2, 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



(MT)

P.S.C. Mo.-No. 36

Access Services Tariff No supplement to this Section 10 tariff will be issued 1st Revised Sheet 8 except for the purpose of canceling this tariff. Replacing Original Sheet 8 REGEIVED ACCESS SERVICES SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued) 10. JUN 27 1986 10.7 Service Offerings to the Federal Government-(Continued) (FC) MISSUURI 10.7.5 Rates and Charges (FC)Public Service Commission A. Voice Grade Special Access Service The provision of T-3 and G conditioned services contemplates station and tandem switching operations, using IC- or End User-provided equipment, as well as special access service. Separate narrowband (C): or voice grade services, where required by the IC- or End Userprovided equipment or switching operation, are furnished in accordance with the applicable sections of this Tariff. Voice Grade Secure Monthly Nonrecurring (RT) Communications USOC Rates Charges Type I, each T-3 Conditioning . GCA ICB rates and charges apply. Additional Conditioning, per service termination. **GTO** ICB rates and charges apply. Type II, each G-1 Conditioning. . . . GCB ICB rates and charges apply. Type III, each ICB rates and charges apply. G-2 Conditioning. . . . GCC Additional Conditioning, per service termination. G20 ICB rates and charges apply. CANCELLED AUG 2 1991 BY 2nd RS, #8 ISALIEN Public Service Commission MISSOURI 册1 1986 86-84 DIC Service Commission JUL Issued: JUN 27 1986 Effective: By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 10 Original Sheet 8

#### ACCESS SERVICES

10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

DEC 29 1233

Public Service Commission

10.8 Service Offerings to the Federal Government-(Continued)

10.8.3 Rates and Charges

A. Voice Grade Special Access Service

The provision of T-3 and G conditioned services contemplates station and tandem switching operations, using IC- or End User-provided equipment, as well as special access service. Separate narrowband or voice grade services, where required by the IC- or End Userprovided equipment or switching operation, are furnished in accordance with the applicable sections of this Tariff.

Voice Grade Secure Communications	USOC	Monthly Rates	Nonrec Chai	rring	Termination Charges
Type I, each T-3 Conditioning	GCA	ICB	rates and	charges	apply.
Additional Conditioning, per service termination.	GTO	ICB	rates and	charges	apply.
Type II, each G-l Conditioning	GCB	ICB	rates and	charges	apply.
Type III, each G-2 Conditioning	GCC	ICB	rates and	charges	apply.
Additional Conditioning, per service termination.	<b>G2</b> 0, ´	ICB	rates and	charges	apply.

GANGELLED

.1111 1 1986 PUBLIC - E P V €£ THE WARDALISE

Southwestern Bell Telephone Company

St. Louis, Missouri

Issued: DEC 2 9 1983 Effective: JAN 0 1 1984 JAM - 1 193 By R. D. BARRON, Vice President-Missouri 83-253

# ACCESS SERVICES

### 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.7 Service Offerings to the Federal Government-(Continued)

10.7.5 Rates and Charges-(Continued)

A. Voice Grade Special Access Service-(Continued)

Voice Grade Secure <u>Communications</u>	<u>USOC</u>	MonthlyNonrecurringRatesCharges
Type III, each G-2 Conditioning	GCC	ICB rates and charges apply.
Additional Conditioning, per service termination.	G20	ICB rates and charges apply.
Type IV, each G-3 Conditioning	GCD	ICB rates and charges apply.
Additional Conditioning, per service termination	G30	ICB rates and charges apply.
B. Wideband Digital Special	Access Se	ervice
Wideband Secure Communications	<u>USOC</u>	Monthly Nonrecurring <u>Rates</u> <u>Charges</u>

Type I, each . . .GWlICB rates and charges apply.Type II, each . . .GW2ICB rates and charges apply.Type III, each . . .GW3CB rates and charges apply.

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Issued: July 3, 1994



Tariff tion 10 Sheet 9 Sheet 9 IVED 7 1986 JURI Commission
7 1986 )//Ri
7 1986 )//Ri
JURI
JURI Commission
Commission
1986
34 ammission
-

4 N 4 18 28

Access Services Tariff Section 10 Original Sheet 9

DEC 29 1933

#### ACCESS SERVICES

10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.8 Service Offerings to the Federal Government-(Continued)

10.8.3 Rates and Charges-(Continued)

A. Voice Grade Special Access Service-(Continued)

Voice Grade Secure Communications	USOC	Monthly Rates	Nonrecurring Charges	Termination Charges
Type IV, each G-3 Conditioning	GCD	ICB rate	s and charges	apply.
Additional Conditioning, per service termination	G30	ICB rate	s and charges	apply.

B. Wideband Digital Special Access Service

Wideband Secure Communications	USOC	MonthlyNonrecurringTerminationRatesChargesCharges
Type I, each	GW1	ICB rates and charges apply.
Type II, each	GW2	ICB rates and charges apply.
Type III, each	GW3	ICB rates and charges apply.

C. Move Charges

When services as set forth in A., and B., preceding, are moved to a different building, the nonrecurring charge applies; when moved to a new location in the same building, a charge of one-half the non-recurring charge applies.

GANCELLED 1111 1 1986 BY PUBLIC SERVICE COMMISSION OF MISSOURI JAN 0 1 1984 DEC 2 9 1983 Effective: Issued: By R. D. BARRON, Vice President-Missouri Ca Southwestern Bell Telephone Company St. Louis, Missouri

### 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

10.7 Service Offerings to the Federal Government-(Continued)

10.7.5 Rates and Charges-(Continued)

#### C. Telecommunications Service Priority (TSP) System

The following rates and charges are in addition to all other rates and charges that may be applicable for other services furnished under the provisions of this tariff which operate in conjunction with the TSP System. This includes, but is not limited to, Testing Services as set forth in Section 13, Paragraph 13.3.5, following.

Access Ordering Charges as set forth in Section 5, preceding, apply to TSP.

The Priority Installation (PI) charge and the Priority Restoration (PR) charge are applied per circuit for Special Access Service; per line or trunk for Switched Access Service. PI and PR apply only one time per circuit or line except when there is subsequent order activity to add legs or lines to the service. One PI or PR will apply each time leg(s) or line(s) are added to an existing TSP service. No TSP charges apply to subsequent order activity that does not affect the TSP assignment.

The Priority Level Change charge (PR8) applies when the only TSP order activity is that of changing priority levels. It applies each time the level is changed to a higher or a lower level.

The Administration and Maintenance charge (PR9) applies per line or trunk for Switched Access Service and per circuit for Special Access Service. Each leg of a multipoint service will be treated as a

separate circuit and the PR9 will apply on a per leg basis.

(AT)

Issued: July 3, 1991

Effective:

ve: August 2, 1991



P	• S	С	•	Mo.	-No	. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(R,T)

(RT)

Access Services Tariff Section 10 1st Revised Sheet 10 Replacing Original Sheet 10



CANCELLED

AUG 2 1991 BY <u>2~1 R5<sup>4</sup>1</u>0 Public Service Commission MISSOURI



Access Services Tariff Section 10 Original Sheet 10

#### ACCESS SERVICES

#### 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

DEC 2 8 1000

10.8 Service Offerings to the Federal Government-(Continued)

10.8.3 Rates and Charges-(Continued)

C. Move Charges-(Continued)

When any service, the rates and charges for which include a Termination Charge, is moved and is installed at a new location, the IC may elect:

- 1. To pay the unexpired portion of the Termination Charge for the service, if any, with the application of a nonrecurring charge and the establishment of a new Termination Charge for such service at the new location, or
- 2. To continue service subject to the unexpired portion of the Termination Charge, if any, and pay the estimated costs of moving such service, provided that the IC requests these charges be quoted prior to ordering the service move. Charges for moving such service will be based on estimated costs attributable to the move.

Move charges include the estimated costs of removal, restoration of services or facilities necessitated by the move, transportation, storage, reinstallation, engineering, labor, supervision, materials, administration and any other specific items of cost directly attributable to the move.



Access Services Tariff Section 10 Original Sheet 11

# ACCESS SERVICES

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

# 10.7 Service Offerings to the Federal Government-(Continued)

### 10.7.5 Rates and Charges-(Continued)

### C. <u>Telecommunications Service Priority (TSP) System</u>-(Continued)

	<u>USOC</u>	Monthly <u>Rates</u>	Nonrecurring Charges
1. Priority Installation (PI) of an Access Service (Per Circuit, Line, or Trunk)(1)			
Prime Service			
Vendor(2)	P1APX		\$50.00
Subcontractor(3)	P1ASX	X None	\$50.00
a. Expedited (Emer-			
gency or Essen-			
<ol> <li>Priority Installation (PI) of an Access Service (Per Circuit, Line, or Trunk)(1)</li> <li>Prime Service Vendor(2) Subcontractor(3)</li> <li>a. Expedited (Emer- gency or Essen- tial)</li> <li>b. Utilizing Specially Constructed Facilities</li> </ol>		the same as tho Section 5, Parag preceding, for t	graph 5.2.2, D.,
b. Utilizing			
Specially			
Constructed		Regulations, rat	tes and charges are
Facilities		the same as tho	e
ជុ		Section 14, foll	owing, for Special
			the facilities for
		Switched or Sp	ecial Access Service
		for which PI is	
(1) When an Access Service is ordered	with both PI and PR,	the nonrecurring	g

- charge for PR Implementation applies.
- (2) Prime Service Vendor denotes status of the Telephone Company when it contracts directly with a TSP end-user, or the end-user's authorized agent, to provide all, or a portion, of a TSP service.
- (3) Subcontractor denotes status of the Telephone Company when it contracts with
- (AT) a Prime Service Vendor to provide a portion of a TSP service to a TSP end-user.

Issued: July 3, 1991

991 Effective: August 2, 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

(AT)

# 10. SPECIAL FEDERAL GOVERNMENT ACCESS SERVICES-(Continued)

### 10.7 Service Offerings to the Federal Government-(Continued)

# 10.7.5 Rates and Charges-(Continued)

### C. <u>Telecommunications Service Priority (TSP) System</u>-(Continued)

(AT)	T) C. <u>Telecommunications Service Priority (TSP) System</u> -(Continued)						
03-0247		<u>USOC</u>	Monthly <u>Rates</u>	Nonrecurring <u>Charges</u>			
003 - IN-20	2. Priority Restoration (PR) of an Access Service (Per Circuit, Line, or Trunk)						
- 02/16/2	a. PR level implementation(1)						
ommission	Prime Service Vendor Subcontractor	PR5PX PR5SX	None None	\$51.00 \$51.00			
c Service Co	b. PR level change ONLY on existing TSP Access Service						
ssouri Publi — — —	Prime Service Vendor Subcontractor	PR8PX PR8SX	None None	\$50.00 \$50.00			
CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247	3. Administration and maintenance of PR (Per Circuit, Line, or Trunk)(2)						
CAN	Prime Service Vendor Subcontractor	PR9PX PR9SX	\$4.10 \$3.35	None None			
     (AT)	<ol> <li>When an Access Service is ordered with b charge for PR Implementation applies.</li> <li>Each leg of a Special Access multipoint se separate circuit and charges apply as descr thru 3, preceding.</li> </ol>	ervice will be trea	ated as a				

Issued: July 3, 1991

Effective: August 2, 1991 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff Section 11 2nd Revised Sheet 1 Replacing 1st Revised Sheet 1

# ACCESS SERVICES

# 11. SPECIAL FACILITIES ROUTING OF ACCESS SERVICES

(AT) The following list matches the Telephone Company's Basic Service Element(BSE) names to the industry standard names for each BSE.

Telephone Company Names

Generic Name of ONA Service

Diversity

Route Diversity

Issued: March 26, 1993

Effective: April 11, 1993



P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 11 1st Revised Sheet 1 Replacing Original Sheet 1

ACCESS SERVICES



SPECIAL FACILITIES ROUTING OF ACCESS SERVICES

11.1 Description of Special Facilities Routing of Access Services JUN 2.7 1985

The services provided under this Tariff are provided over such muces and facilities as the Telephone Company may elect. Special Facilities RoutinesSion is involved when, in order to comply with requirements spe or End User, the Telephone Company provides Switched Access Service, Special Access Service or Special Federal Government Access Service in a manner which includes one or more of the following conditions:

# 11.1.1 Diversity

Two or more services must be provided over not more than two celfeed physical routes. APR 11 1993 - #

11.1.2 Avoidance

A service must be provided on a route which avoids specesting recorrections.

# 11.1.3 Cable-Only Facilities

Certain Voice Grade services are provided on Cable-Only Facilities to meet the particular needs of an IC or End User.

Service is provided subject to the availability of Cable-Only Facilities. In the event of service failure, restoration will be made through the use of any available facilities as selected by the Telephone Company.



(CT)

Avoidance and Diversity are available on Switched Access Service as set forth in Section 6, preceding; Special Access Services as set forth in Section 7, Paragraphs 7.2.1, 7.2.3 and 7.2.5, preceding, and Special Federal. Government Access Services as set forth in Section 10, preceding. Cable-Only Facilities are available for Switched Access Service as set forth in Section 6, preceding; Voice Grade Special Access Services as set forth in Section 7, Paragraph 7.2.3, preceding, and Special Federal Government Access Services as set forth in Section 10, preceding.

FILED 班 1 1986 86-84 ice Commission Issued: JUN 27 1986 JUL Effective: 1 1980 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company

St. Louis, Missouri
Access Services Tariff Section 11 Original Sheet 1

DEC 2 113

#### ACCESS SERVICES

#### 11. SPECIAL FACILITIES ROUTING OF ACCESS SERVICES

11.1 Description of Special Facilities Routing of Access Services

The services provided under this Tariff are provided over such routes and facilities as the Telephone Company may elect. Special Facilities Routing is involved when, in order to comply with requirements specified by the IC or End User, the Telephone Company provides Switched Access Service, Special Access Service or Special Federal Government Access Service in a manner which includes one or more of the following conditions:

11.1.1 Diversity

Two or more services must be provided over not more than two different physical routes.

11.1.2 Avoidance

A service must be provided on a route which avoids specified geographical locations.

#### 11.1.3 Cable-Only Facilities

Certain Voice Grade services are provided on Cable-Only Facilities to meet the particular needs of an IC or End User.

Service is provided subject to the availability of Cable-Only Facilities. In the event of service failure, restoration will be made through the use of any available facilities as selected by the Telephone Company.

Avoidance and Diversity are available on Switched Access Service as set forth in Section 6, preceding; Narrowband, Voice Grade and Wideband Analog Special Access Services as set forth in Section 7, Paragraphs 7.2.1, A., B. and E., preceding, and Special Federal Government Access Services as set forth in Section 10, preceding. Cable-Only Facilities are available for Switched Access Service as set forth in Section 6, preceding; Voice Grade Special Access Services as set forth in Section 7, Paragraph 7.2.1, B., preceding, and Special Federal Government Access Services as set forth in Section 10, preceding

JUL 1 1980 SEPVICE COMMISSION OF MISSOUR JAN 0 1 1984 JAN Issued: Effective: DEC 2 9 1983 83 - 253 By R. D. BARRON, Vice President-Missouri I Public Service Contribe Southwestern Bell Telephone Company St. Louis, Missouri

## ACCESS SERVICES

# (MT) 11. SPECIAL FACILITIES ROUTING OF ACCESS SERVICES-(Continued)

11.1 Description of Special Facilities Routing of Access Services

The services provided under this Tariff are provided over such routes and facilities as the Telephone Company may elect. Special Facilities Routing is involved when, in order to comply with requirements specified by the IC or End User, the Telephone Company provides Switched Access Service, Special Access Service or Special Federal Government Access Service in a manner which includes one or more of the following conditions:

## (AT) 11.1.1 Diversity BSE

Two or more services must be provided over not more than two different physical routes.

## 11.1.2 Avoidance

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

(MT)

A service must be provided on a route which avoids specified geographical locations.

11.1.3 Cable-Only Facilities

Certain Voice Grade services are provided on Cable-Only Facilities to meet the particular needs of an IC or End User.

Service is provided subject to the availability of Cable-Only Facilities. In the event of service failure, restoration will be made through the use of any available facilities as selected by the Telephone Company.

Avoidance and Diversity are available on Switched Access Service as set forth in Section 6, preceding, Special Access Services as set forth in Section 7, Paragraphs 7.2.1, 7.2.3 and 7.2.5, preceding, and Special Federal Government Access Services as set forth in Section 10, preceding. Cable-Only Facilities are available for Switched Access Service as set forth in Section 6, preceding; Voice Grade Special Access Services as set forth in Section 7, Paragraph 7.2.3, preceding, and Special Federal Converse Access Services as set forth in Section 7, Paragraph 7.2.3, preceding, and Special Federal

Government Access Services as set forth in Section 10, preceding.



## ACCESS SERVICES

# 11. SPECIAL FACILITIES ROUTING OF ACCESS SERVICES-(Continued)

11.1 Description of Special Facilities Routing of Access Services-(Continued)

In order to avoid the compromise of special routing information, the Telephone Company will provide the required routing information for each specially routed service to only the ordering IC. If requested by the IC, this information will be provided when service is installed and prior to any subsequent changes in routing.

The offering of Special Facilities Routing of Access Services contemplates the use of existing facilities. Should facilities not be available, it may be necessary to construct such facilities, either as (1) normal or (2) Special Construction. If Special Construction is involved, the regulations, as set forth in Section 14, following apply.

The rates and charges for Special Facilities Routing of Access Services as set forth in Paragraph 11.2, following, are in addition to all other rates and charges that may be applicable for services provided under other sections of this Tariff.

AT)  $\frac{1}{5}$  11.2 Rates and Charges for Special Facilities Routing of Access Service

The rates and charges for Special Facilities Routing of Access Services are as follows:

11.2.1 Diversity

For each service provided in accordance with 11.1.1, preceding, the rates and charges will be developed on an individual case basis and filed following:

## <u>USOC</u>

## SYD++

11.2.2 Avoidance

For each service provided in accordance with Paragraph 11.1.2, preceding, the rates and charges will be developed on an individual case basis and filed following:

## <u>USOC</u>

## SYA++

Issued: June 27, 1986

Effective: July 1, 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Access Services Tariff Section 11 Original Sheet 2

DEC 2.2 (203

#### ACCESS SERVICES

11. SPECIAL FACILITIES ROUTING OF ACCESS SERVICES-(Continued)

11.1 Description of Special Facilities Routing of Access Services-(Continued)

In order to avoid the compromise of special routing information, the Telephone Company will provide the required routing information for each specially routed service to only the ordering IC. If requested by the IC, this information will be provided when service is installed and prior to any subsequent changes in routing.

The rates and charges for Special Facilities Routing of Access Services will be determined on an individual-case basis and are in addition to all other rates and charges that may be applicable for services provided under other sections of this Tariff.

			GANCELLED	
•			JUL 1 1986 BY R.S. 2 PUBLIC SERVICE COMMISSION OF MISSOURI	
Ţ	Issued:	DEC 2 9 1983	Effective: JAN 0 1	17 1150:
		Ву	R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri	83-259 Public Service Com

## ACCESS SERVICES

### 11. SPECIAL FACILITIES ROUTING OF ACCESS SERVICES-(Continued)

- 11.2 Rates and Charges for Special Facilities Routing of Access Service-(Continued)
  - 11.2.3 Diversity and Avoidance Combined

For each service provided in accordance with Paragraph 11.1.1 and Paragraph 11.1.2, preceding, combined, the rates and charges will be developed on an individual case basis and filed following:

#### <u>USOC</u>

#### SYB++

11.2.4 Cable-Only Facilities

For each service provided in accordance with Paragraph 11.1.3, preceding, the rates and charges will be developed on an individual case basis and filed following:

<u>USOC</u>

SYC++





## ACCESS SERVICES

## 12. SPECIALIZED SERVICE OR ARRANGEMENTS

12.1 General

Specialized Service or Arrangements may be provided by the Telephone Company, at the request of an IC, on an individual-case basis if such service or arrangements meet the following criteria:

- The requested service or arrangements are not offered under other sections of this Tariff.
- The facilities utilized to provide the requested service or arrangements are of a type normally used by the Telephone Company in furnishing its other services.
- The requested service or arrangements are provided within a LATA.
- The requested service or arrangements are compatible with other Telephone Company services, facilities and its engineering and maintenance practices.
- This offering is subject to the availability of the necessary Telephone Company personnel and capital resources.

## (AT) 12.2 Rates and Charges

Rates and charges and additional regulations, if applicable, for Specialized Service or Arrangements are provided on an individual-case basis.

Issued: June 27, 1986

Effective: July 1, 1986



Access Services Tariff Section 12 Original Sheet 1

DEC 29 223

#### ACCESS SERVICES

12. SPECIALIZED SERVICE OR ARRANGEMENTS

12.1 General

Specialized Service or Arrangements may be provided by the Telephone Company, at the request of an IC, on an individual-case basis if suchservice or arrangements meet the following criteria:

- The requested service or arrangements are not offered under other sections of this Tariff.
- The facilities utilized to provide the requested service or arrangements are of a type normally used by the Telephone Company in furnishing its other services.
- The requested service or arrangements are provided within a LATA.
- The requested service or arrangements are compatible with other Telephone Company services, facilities and its engineering and maintenance practices.
- This offering is subject to the availability of the necessary Telephone Company personnel and capital resources.

	BANGELLED JULI 1986 BY MR S.I PUBLIC SERVICE COMMISSION OF MISSOUR
	FILE
Issued: <b>DEC</b> 29 1983 By R. D.	Effective: JAN 0 1 984 83-258 BARRON, Vice President-Missouri Public Service Commission

St. Louis, Missouri

### ACCESS SERVICES

### 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES

For the proposes of Section 13 of the Southwestern Bell Telephone Access Tariff P.S.C. Mo.-No. 36 the terms "Basic Time", "Overtime" and "Premium Time" are defined as follows:

Basic Time - Work related efforts of the Telephone Company performed during normally scheduled working hours.

Overtime - Work related efforts of the Telephone Company performed outside of a normally scheduled work day.

Premium Time - Work related efforts of the Telephone Company performed outside of a normally scheduled work week.

13.1 Additional Engineering

Additional Engineering will be provided by the Telephone Company at the request of the customer only when:

- A. A customer requests additional technical information after the Telephone Company has already provided the technical information normally included on the Design Layout Report (DLR) as set forth in 6.1.4 and 7.1.6, preceding.
- B. Additional engineering time is incurred by the Telephone Company to engineer a customer's request for a customized service as set forth in 7.2, preceding.

The Telephone Company will notify the customer that additional engineering charges, as set forth in 13.1.1 following, will apply before any additional engineering is undertaken.

If more than one engineer is involved in the same additional engineering project, the total amount of time for all engineers involved will be aggregated prior to the distribution of time between the "First Half Hour or Fraction Thereof" and "Each Additional Half Hour or Fraction Thereof" rate categories.

### 13.1.1 Charges for Additional Engineering

The charges for additional engineering are as follows:

Effective: July 18, 1988



----

----

والالجام والمحاصور

			r.s.c. MoNo. 30	
tari exce	iff w ept f	ement to this ill be issued or the purpose ling this tariff.		Access Services Tariff Section 13 1st Revised Sheet 1 Replacing Original Sheet 1
			ACCESS SERVICES	REGEIVED
13.	ADD	ITIONAL ENGINEERING, AL	DITIONAL LABOR AND MIS	CELLANEOUS SERVICES
(AT)	Acc	the proposes of Section ess Tariff P.S.C. MON emium Time" are defined	lo. 36 the terms "Basic	JUN 2 7 1986 rn Bell Telephone Time", "Overtime" MISSUURI Public Service Commission
			ated efforts of the Te mally scheduled working	
			ed efforts of the Tele; a normally scheduled w	
			elated efforts of the a normally scheduled w	
(CT) 13	3.1	Additional Engineering		
		Additional Engineering request of the customer		e Telephone Company at the
		Company has already pro	wided the technical in	rmation after the Telephone formation normally included h in 6.1.3 and 7.1.6, preceding.
		Additional engineering engineer a customer's 1 7.2, preceding.		e Telephone Company to d service as set forth in
				r that additional engineering 11 apply before any additional NCELLED
(RT)				L 18 1988 2nd R.S. #1
(FC)	13.1	.1 Charges for Additio	Public Ser	vice Commission ISSOURI
		The charges for add	litional engineering ar	同日和
				Public Service Commission
Iss	ued:,	JUN 2 7 1986	Effective:	JUL 1 1986
			BARRON, President-Misso nwestern Bell Telephone St. Louis, Missouri	Company

Access Services Tariff Section 13 Original Sheet 1

#### ACCESS SERVICES

### 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES

13.1 Additional Engineering

DEC 2 S 1983

Additional Engineering is that engineering or engineering consultation requested by the IC as set forth in Paragraphs 13.1.1 through Paragraph 13.1.3, following. The Telephone Company will notify the IC that additional engineering charges as set forth in Paragraph 13.1.4, following, will apply before any additional engineering is undertaken.

13.1.1 Engineering Consultation

Engineering Consultation involves technical advice from the Telephone Company to the IC not in connection with a specific order. Engineering Consultation also includes situations in which the IC requests the Telephone Company to provide information or to perform a function which will entail additional engineering by the Telephone Company. Inquiries of a short duration in which no significant engineering time is required and inquiries associated with IC service forecasts are not included.

13.1.2 Engineering of Connections with Other Telephone Companies

Engineering of connections with other telephone companies, if not Concurring Carriers, is the engineering activity of contacting, coordinating and designing with another telephone company, portions of facilities which connect to facilities provided by the other telephone company.

13.1.3 Charges for Additional Engineering

The charges for additional engineering are as follows:

GANCELLED

JUL 1 1986

1Rt PUBLIC SERVICE COMMISSION

OF MISSOURT

JAN 01 1984

Issued: DEC 29 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

Effective:

### ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.1 Additional Engineering-(Continued)

13.1.1 Charges for Additional Engineering-(Continued)

Additional Engineering Periods	<u>USOC</u>	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
Basic Time, per engineer	AEH	\$75.99	\$21.40
Overtime, per engineer	AEH	79.33	74.24

## 13.2 Additional Labor

Additional Labor is that labor requested by the customer on a given service and agreed to by the Telephone Company.

The Telephone Company will notify the customer that Additional Labor Charges will apply before any additional labor is undertaken. Additional Labor Charges apply on a first and additional basis for each half hour or fraction thereof. If more than one technician is involved in the same Additional Labor Project, the total amount of time for all technicians involved will be aggregated prior to the distribution of time between the "First Half Hour or Fraction Thereof" and "Each Additional Half Hour or Fraction Thereof" rate categories.

A call-out of a Telephone Company employee for Additional Labor at a time not consecutive with the Telephone Company's Business Day is subject to a minimum charge of four hours, i.e., when Overtime and/or Premium Time charges apply.

For Testing and Maintenance Services, if the customer elects not to release a circuit during the Telephone Company's Business Day, the Telephone Company will work with the customer to reach a mutually agreed upon time.



	No supplement tariff will except for of canceling	be issued			1	ss Services Tariff Section 13 st Revised Sheet 2 g Original Sheet 2
_			ACCESS SI	ERVICES	i	
	13. ADDITIC (Contin	ONAL ENGINEERING, A	DDITIONAL 1	LABOR AND	MISCELLANEOU	
	13.1 Add:	itional Engineering	;-(Continued	1)		JUN 2 7 1986
(FC)	13.1.1	Charges for Additi	onal Engin	ering-(Co		MISSUURI
					First Half	Public Service Commission
					Hour or	Half Hour or
		Additional Enginee	ring		Fraction	Fraction
		Periods		USOC	Thereof	Thereof
(RT)		Basic Time,				
(CR)		per engineer	• • •	AEH	\$75.99	\$21.40
(RT)		Overtime,				
(CR)		per engineer		AEH	79.33	24.74

#### 13.2 Additional Labor

Additional labor is that labor requested by the IC on a given service and agreed to by the Telephone Company as set forth in Paragraphs 13.2.1 through 13.2.5, following. The Telephone Company will notify the IC that additional labor charges as set forth in Paragraph 13.2.6, following, will apply before any additional labor is undertaken.

13.2.1 Overtime Installation

Overtime installation is that Telephone Company installation effort outside of regularly scheduled working hours.

#### 13.2.2 Overtime Repair

Overtime repair is that Telephone Company maintenance effort performed outside of regularly scheduled working hours.



FILED 111 1 1980 86-84 Public Service Commission

Issued: JUN 27 1986

Effective: JUL 1 1986

Access Services Tariff Section 13 Original Sheet 2

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.1 Additional Engineering-(Continued)

13.1.3 Charges for Additional Engineering-(Continued)

Additional Engineering Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
Basic Time, regularly scheduled working hours, per engineer	AEH	\$74.52	\$44.93
Overtime, outside of regularly scheduled working hours, per engineer	AEH	82.89	53.29

#### 13.2 Additional Labor

Additional labor is that labor requested by the IC on a given service and agreed to by the Telephone Company as set forth in Paragraphs 13.2.1 through 13.2.5, following. The Telephone Company will notify the IC that additional labor charges as set forth in Paragraph 13.2.6, following, will apply before any additional labor is undertaken.

13.2.1 Overtime Installation

Overtime installation is that Telephone Company installation effort outside of regularly scheduled working hours.

13.2.2 Overtime Repair

Overtime repair is that Telephone Company maintenance effort performed outside of regularly scheduled working hours

1986 .1111 1 PUBLIC SERVICE COMMISSION OF MISSOURI Issued: **DEC** 2 9 1983 JAN 011 1984 Effective: Public Ser By R. D. BARRON, Vice President-Missour Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 2.1

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - NOV 17 1994

MO. PUBLIC SERVICE COMM.

13.2 Additional Labor-(Contintued)

13.2.1 Overtime Installation (M,T)

13.2.2

Overtime Installation is that Telephone Company installation effort outside of normally scheduled working hours.

- IN-2003-024 (RT)

CANCELLED - Missouri Public Service Commission - 02/16/2003

(MT)

(RT)

				FILED
Issued:	NOV 1 7 1994	Effective:	DEC 1	<sup>7</sup> 1994 DEC 1 7 1994
	By HO	RACE WILKINS, JR., President-Mi	issouri	
		Southwestern Bell Telephone S. Louis, Missouri		MISSOURI Public Service Commission

Access Services Tariff Section 13 3rd Revised Sheet 3 Replacing 2nd Revised Sheet 3

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.2 Additional Labor-(Continued)

13.2.3 Stand By

Stand by includes all time in excess of one-half (1/2) hour during which Telephone Company personnel stand by to make installation acceptance tests or cooperative tests with an IC to verify facility repair on a given service.

13.2.4 Testing and Maintenance with Other Telephone Companies

Additional testing, maintenance or repair of facilities which connect to facilities of other telephone companies, which is in addition to normal effort required to test, maintain or repair facilities provided solely by the Telephone Company.

13.2.5 Other Labor

Other labor is that additional labor not included in Paragraphs 13.2.1 through 13.2.4, preceding including, but not limited to labor incurred to extend the Point of Termination as set forth in 2.1.4, preceding, and labor incurred to accommodate a specific customer request that involves only labor which is not covered by any other section of this tariff.

If more than one technician is involved in the same additional labor project, the total amount of time for all technicians involved will be aggregated prior to the distribution of time between the "First Half Hour or Fraction Thereof" and "Each Additional Half Hour or Fraction Thereof" rate categories.



<sup>13.2.6</sup> Charges for Additional Labor

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 2nd Revised Sheet 3 Replacing 1st Revised Sheet 3

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES (VE)
  - 13.2 Additional Labor-(Continued)

13.2.3 Stand By

NOV 26 1986

Stand by includes all time in excess of one-half (1/2) hericulting Ston which Telephone Company personnel stand by to make installation acceptance tests or cooperative tests with an IC to verify facility repair on a given service.

13.2.4 Testing and Maintenance with Other Telephone Companies

Additional testing, maintenance or repair of facilities which connect to facilities of other telephone companies, which is in addition to normal effort required to test, maintain or repair facilities provided solely by the Telephone Company.

13.2.5 Other Labor

(RT)

Other labor is that additional labor not included in Paragraphs 13.2.1 through 13.2.4, preceding including, but not limited to labor incurred to extend the Point of Termination as set forth in 2.1.4, preceding, and labor incurred to accommodate a specific customer request that involves only labor which is not covered by any other section of this tariff.

	CANCELL	ED		
	JUL 18 198 BY <u>3rf R.S</u> Public Service Cor MISSOUR	<u>.₩3</u> n <b>missio</b> n		JAN - 1 1987
				A BAR AND A MANAGER
Issued:	DEC 01 1986	Effective:	IAN 011	987

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 1st Revised Sheet 3

Replacing Original Sheet 3

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES ENVICES
  - 13.2 Additional Labor-(Continued)

13.2.3 Stand By

JUN 27 1986

Stand by includes all time in excess of one-half (1/3), hour during which Telephone Company personnel stand by to make installation acceptance tests or cooperative tests with an IC to verify facility repair on a given service.

13.2.4 Testing and Maintenance with Other Telephone Companies

(RT)

Additional testing, maintenance or repair of facilities which connect to facilities of other telephone companies, which is in addition to normal effort required to test, maintain or repair facilities provided solely by the Telephone Company.



13.2.5 Other Labor

Other labor is that additional labor not included in Paragraphs 13.2.1 through 13.2.4, preceding including, but not limited to labor incurred for the installation of inside wire, used to extend the Point of Termination as set forth in 2.1.4, preceding, and labor incurred to accommodate a specific customer request that involves only labor which is not covered by any other section of this tariff.

ED	ARBELL					
		]]				
1	JAN 1 198					
# 3 MISSICN	JAN 1 191					
	i i					
1 1985						
Bervice Commission	JUL PU198	Effective:	· <u> </u>	ใช่ออ	JUN 27 1	Issued:
9 ji	uri Division		R. D. BARRON		JUN 27 1	Issued:

St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 3

DEC 2 C tom

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.2 Additional Labor-(Continued)

13.2.3 Stand By

Stand by includes all time in excess of one-half (1/2) hour during which Telephone Company personnel stand by to make installation acceptance tests or cooperative tests with an IC to verify facility repair on a given service.

13.2.4 Testing and Maintenance with Other Telephone Companies

Additional testing, maintenance or repair of facilities which connect to facilities of other telephone companies, if they are not Concurring Carriers, which is in addition to normal effort required to test, maintain or repair facilities provided solely by the Telephone Company.

13.2.5 Other Labor

Other labor is that additional labor not included in Paragraphs 13.2.1 through 13.2.4, preceding.

GANCELLED

.111 1 1986

PUBLIC SERVICE COMMISSION OF WESSOURI

JAN 0 1 1984

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Public Service Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

Access Services Tariff Section 13 3rd Revised Sheet 4 Replacing 2nd Revised Sheet 4

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.2 Additional Labor-(Continued)

13.2.6 Charges for Additional Labor-(Continued)

The charges for additional labor are as follows:

Additional Labor Periods	<u>USOC</u>	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction <u>Thereof</u>
Installation			
- Overtime, per technician	ALH	\$58.01(1)	\$ 3.42(1)
- Premium Time, per technician	ALH	61.42(1)	6.83(1)
Stand by			
- Basic Time, per technician	ALT	None	18.49

(1) A call-out of a Telephone Company employee at a time not consecutive with

(CT) the Telephone Company's Business Day is subject to a minimum charge of four

(AT) hours when overtime and/or premium time charges apply.

Issued: November 17, 1994

Effective: December 17, 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri



(L) CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Access Services Tariff No supplement to this tariff will be issued Section 13 2nd Revised Sheet 4 except for the purpose Replacing 1st Revised Sheet 4 of canceling this tariff. ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) RECEIVED 13.2 Additional Labor-(Continued) JUN 17 1988 13.2.6 Charges for Additional Labor-(Continued) MISSOURI The charges for additional labor are as follo Bublic Service Commission First Half Each Additional Half Hour or Hour or Additional Labor Fraction Fraction Periods USOC Thereof Thereof Installation or Repair - Overtime. ALH per technician . . . \$58.01(1) \$ 3.42(1) - Premium Time, per technician . . . ALH 61.42(1) 6.83(1) Stand by - Basic Time, per technician . . . ALT None 18.49 CANCELLED Public Servic

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.
FILED

JUL 18-1988

Issued: JUN 17 1988

(MT)

(FC)

Effective: JUL 1 8 1988 Service Commission

ex			Access Services Tariff Section 13 lst Revised Sheet 4 Replacing Original Sheet 4			
•	(Contin 13.2 Addi	tional Labor-(Continued) Charges for Additional La The charges for additional	bor	as follows: First Half Pu Hour or	REGEIVED JUN 27 1986 MISSUURI	
		Additional Labor Periods	USOC	Fraction Thereof	Fraction Thereof	
(RT) (CR)		<pre>Installation or Repair - Overtime,     per technician</pre>	ALH	\$58.01(1)	\$ 3.42(1)	
(RT) (CR)		- Premium Time, per technician Stand by	ALH	61.42(1)	6.83(1)	
(RT) (CR)		- Basic Time, per technician	ALT	None	18.49	
٠	Pi	JUL 18 1988 BY <u>and R.S.#4</u> ublic Service Commission				
	l) A call- the emp hours.	MISSOURI out of a Telephone Company loyee's scheduled work per	employee at iod is subje	t a time not cor act to a minimum	Elil Eli psecutive with charge of four 8.6-84	
	ssued: JUI	127 1986 Br P. D. PARRON R	Effectiv		)86	
		By R. D. BARRON, P Southwestern B St. Lo		ne Company		

1

.

ł

Access Services Tariff Section 13 Original Sheet 4

#### ACCESS SERVICES

13. ADDITIONAL FNGINFERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.2 Additional Labor-(Continued)
  - 13.2.6 Charges for Additional Labor

The charges for additional labor are as follows  $\{ \frac{r}{2} \}$ 

Additional Labor Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
Installation or Repair		·	
- Overtime, outside of regularly scheduled working hours, on a scheduled work day, per technician	. ALH	\$33.09(1)	\$ 3.50(1)
- Premium Time, outside of scheduled work day,			
per technician	. ALH	36.61(1)	7.01(1)
Stand by			
- Basic Time, regularly scheduled working hours, per technician		None	20.22
	GANCE	lled	
	JUL 1 BY ARC PUBLIC SERVICE	5.4	
(1) A call-out of a Telephone Compa the employee's scheduled work hours.	any employee a	t a time not co	
<u></u>			
Issued: DEC 2 9 1983	Effectiv	ve: JAN 0 1 <u>19</u>	)84 JAAN - 1 (88.)

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company

Effective:	JAN O	

St. Louis, Missouri

83-2	53
Public Service	Commission

DEC 20 113

Access Services Tariff Section 13 1st Revised Sheet 5 Replacing Original Sheet 5

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.2 Additional Labor-(Continued)

13.2.6 Charges for Additional Labor-(Continued)

The charges for additional labor are as follows:

Additional Labor Periods	<u>USOC</u>	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
- Overtime, per technician	ALT	None	\$21.91(1)
- Premium time, per technician	ALT	None	25.32(1)

(l) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.

Issued: June 27, 1986

Effective: July 1, 1986



Access Services Tariff Section 13 Original Sheet 5

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 201013

- 13.2 Additional Labor-(Continued)
  - 13.2.6 Charges for Additional Labor-(Continued)

The charges for additional labor are as follows:

Additional Labor Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
- Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician	ALT	None	\$23.72(1)
<ul> <li>Premium time, outside</li> <li>of scheduled work day,</li> <li>per technician</li> </ul>	ALT	None	27.23(1)

GANGELLED

JUL 1 1986 SERVICE COM OF WESSLAUR

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouril Public Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

**Fffective:** 

JAN 0 1 1984

Access Services Tariff Section 13 1st Revised Sheet 6 Replacing Original Sheet 6

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.2 Additional Labor-(Continued)

13.2.6 Charges for Additional Labor-(Continued)

The charges for additional labor are as follows:

Additional Labor Periods	<u>USOC</u>	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
Testing and maintenance with other telephone companies, or other labor			
- Basic time, per technician	ALK	\$73.08	\$18.49
- Overtime, per technician	ALK	76.50(1)	21.91(1)
- Premium time, per technician	ALK	79.91(1)	25.32(1)

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.

Issued: June 27, 1986

Effective: July 1, 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Access Services Tariff Section 13 Original Sheet 6

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 2.0 (...)
  - 13.2 Additional Labor-(Continued)
    - 13.2.6 Charges for Additional Labor-(Continued)

Public Scottine (Comparenter)

The charges for additional labor are as follows:

	Additional Labor Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
	Other Labor (or testing and maintenance with other telephone companies if they are not Concurring Carriers)	5		
	- Basic time, regularly scheduled working hours, per technician	ALK	\$49.82	\$20.22
	- Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician	ALK	53.32(1)	23.72(1)
	<ul> <li>Premium time, outside of scheduled work day, per technician</li> </ul>	ALK	56.83(1)	27.23(1)
		GANG	ELLED	
		JUL	1 1986 <u>R.S.L</u>	
the e	l-out of a Telephone Company mployee's scheduled work per: hours.	amn love at a	time not co	onsecutive with um charge of 了山口民民
Issued:	DEC 2 9 1983	Effecti		1984 JAN 71 1927 1984 83 - 253
	Southwestern	Vice Prest Bell Teleph Duis, Misso	one Company 🐃	Public Service Commis

Access Services Tariff Section 13 2nd Revised Sheet 7 Replacing 1st Revised Sheet 7

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

### 13.3 Miscellaneous Services

- 13.3.1 Customer Owned Equipment Trouble Isolation Charge
  - A. When a customer reports trouble to the Telephone Company for clearance and no trouble is found in the Telephone Company's facilities, the customer shall be responsible for payment of a Customer Owned Equipment Trouble Isolation Charge for the period of time from when Telephone Company personnel are dispatched to the customer's premises to when the work is completed. Failure of Telephone Company personnel to find trouble in Telephone Company facilities will result in no charge if the trouble is actually in those facilities, but not discovered at the time.
  - B. The customer shall be responsible for payment of a Customer Owned Equipment Trouble Isolation Charge when the Telephone Company dispatches personnel to the customer's premises, and the trouble is in equipment or communications systems provided by other than the Telephone Company or in detariffed CPE provided by the Telephone Company.

In either Paragraphs 13.3.1., A or B preceding, the Customer Owned Equipment Trouble Isolation Charge includes all personnel dispatched, including technicians dispatched to another location(s), when necessary for the purpose of testing with a technician(s) dispatched to the customer's premises.

(AT)



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 1st Revised Sheet 7 Replacing Original Sheet 7

Public Service Commission

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES 27 1986 (Continued) MISSUURI

13.3 Miscellaneous Services

(CT)

(CT)

(MT)

(AT)

- 13.3.1 Customer Owned Equipment Trouble Isolation Charge
  - A. When a customer reports trouble to the Telephone Company for clearance and no trouble is found in the Telephone Company's facilities, the customer shall be responsible for payment of a Customer Owned Equipment Trouble Isolation Charge for the period of time from when Telephone Company personnel are dispatched to the customer's premises to when the work is completed. Failure of Telephone Company personnel to find trouble in Telephone Company facilities will result in no charge if the trouble is actually in those facilities, but not discovered at the time.
  - B. The customer shall be responsible for payment of a Customer Owned Equipment Trouble Isolation Charge when the Telephone Company dispatches personnel to the customer's premises, and the trouble is in equipment or communications systems provided by other than the Telephone Company or in detariffed CPE provided by the Telephone Company.
    - In either Paragraphs (A) or (B) preceding, no credit allowance will be applicable for the interruption involved if the Customer Owned Equipment Trouble Isolation Charge applies.

}		FEB 8 1988 BY J. S. FT7 Public Service Commission MISSOURI		E 1 86- 10 Service	目的 1986 <b>8 4</b> Commissio	
Issued:	JUN 27 1900	Effective:	JUL 11	1986		•
		). BARRON, President-Missouri Ithwestern Bell Telephone Com St. Louis, Missouri				

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 Original Sheet 7

DEC 20 LLJ

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services
    - 13.3.1 Maintenance of Service
      - A. When an IC reports a trouble to the Telephone Company for clearance, the IC shall be responsible for payment of a Maintenance of Service Charge when:
        - 1. The trouble is observed to be in the equipment or communications systems provided by other than the Telephone Company, or
        - 2. No trouble is found in the Telephone Company's facilities.

In either case, no credit allowance will be applicable for the interruption involved.

- B. The Telephone Company will advise the IC that it may be responsible for payment of a Maintenance of Service Charge should either of the conditions in Paragraph A, preceding, apply.
- C. The charges for Maintenance of Service are as follows:

	Maintenance of Service Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
	Pasic Time, regularly scheduled working hours, per technician	MVV	\$49.82	\$20.22
	Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician	Gance	LED <sub>53.32(1)</sub>	23.72(1)
	Premium Time, outside of scheduled work day, per technician	BYINGS	1986 5.7 () A MS. () 56.83(1)	27.23(1)
the	<pre>ll-out of a Telephone Compa employee's scheduled work p hours.</pre>			
Issued:	DEC 2 9 1983	Effective	≥: JAN 0 1 1984	83-258
		N, Vice Preside n Bell Telephor		

St. Louis, Missouri

Access Services Tariff Section 13 2nd Revised Sheet 7.01 Replacing 1st Revised Sheet 7.01

### ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

## 13.3.1 Customer Owned Equipment Trouble Isolation-(Continued)

C. The charges for Customer Owned Equipment Trouble Isolation are as follows:

Customer Owned Equipment Trouble Isolation	<u>USOC</u>	First Half Hour or Fraction <u>Thereof</u>	Each Additional Half Hour or Fraction <u>Thereof</u>
Basic Time, per technical	MVV	\$73.08	\$18.49
Overtime, per technician	MVV	76.50(1)	21.91(1)
Premium Time, per technician	MVV	79.91(1)	25.32(1)

13.3.2 Restoration Priority(2)

(1) A call-out of a Telephone Company employee at a time not consecutive with the Telephone Company's Business Day is subject to a minimum charge of four hours when overtime and/or premium time charges apply.

(2) Restoration Priority has been superseded by Telecommunications Service Priority (TSP) System as set forth in Section 10, Paragraph 10.7.1, C., preceding. Prior to March 10, 1993, RP and TSP will both be in effect, but no new assignments of RP will be allowed. The existing RP services will be honored and maintained during the interim period between the Initial Operating Capability (IOC) and Full Operating Capability (FOC) of TSP. No order for changes to existing RP circuits will be accepted without a TSP Authorization Code or a definitive statement to remove the RP assignment. On March 10, 1993, the Telephone Company will remove all RP assignments that have not been converted to TSP or discontinued free of charge.

Issued: November 17, 1994 Effective: December 17, 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 1st Revised Sheet 7.01 Replacing Original Sheet 7.01

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES N 28 1991 (Continued)
  - 13.3 Miscellaneous Services-(Continued)

MISSOURI

RECEIVED

Public Service Commission

- 13.3.1 Customer Owned Equipment Trouble Isolation-(Continued)
  - C. The charges for Customer Owned Equipment Trouble Isolation are as follows:

Customer Owned Equipment Trouble Isolation	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
Basic Time, per technicial	MVV	\$73.08	\$18.49
Overtime, per technician	MVV	76.50(1)	21.91(1)
Premium Time, per technician	MVV	79.91(1)	25.32(1)
3.3.2 Restoration Priority(2)	)	CAN	VCELLED

(AT)13.3.2 Restoration Priority(2)

(RT)

(RT)

- DEC 161994 Public Service Commission MISSOURI
- (1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.
- (AT) (2) Restoration Priority has been superseded by Telecommunications Service Priority (TSP) System as set forth in Section 10, Paragraph 10.7.1, C., preceding. Prior to March 10, 1993, RP and TSP will both be in effect, but no new assignments of RP will be allowed. The existing RP services will be honored and maintained during the interim period between the Initial Operating Capability (IOC) and Full Operating Capability (FOC) of TSP. No order for changes to existing RP circuits will be accepted without a TSP Authorization Code or a definitive statement to remove the RP assignment. On March 10, 1993, the Telephone Company will remove all RP assignments that AT) have not been converted to TSP or discontinued free of charge.

Issued:	1111 0		·	·	Effective	:				
	JUL - ^	591				- ADS -	2 1991		1991	
		By R. D.	BARRON,	Preside	nt-Missouri	Divisio	n	AUG 2	1991	
		Sout		n Bell To Louis, 1	elephone Co				Commiss	

\_

-----

- - - -

-----

. .

•	No supplement to this tariff will be issued except for the purpose of canceling this tariff.		Access Services Sec Original She	tion 13 et 7.01
	ACCESS SEI			
	13. ADDITIONAL ENGINEERING, ADDITIONAL LA (Continued)	ABOR AND MISCEL	LANEOUS SERVICES-	7 1986
	13.3 Miscellaneous Services-(Continued)	)	MISSI Public Service	
(CT)	13.3.1 Customer Owned Equipment Troub	ole Isolation-(		
(MT)	C. The charges for Customer Owned B follows:	Equipment Troub	le Isolation are	as
(CT)     (CT)	Equipment Trouble	First Half Hour or Fraction Thereof	Each Additiona Half Hour or Fraction Thereof	.1
(RT) (CR)		177	\$73.08	\$18.49
(RT) (CR)	•	AAA	76.50(1)	21.91(1)
(RT) (CR)	•	MVV	79.91(1)	25.32(1)
(MT) (AT)	The Telephone Company will ar Restoration Priority on receiv with the Federal Communication A charge applies when a reques Priority is received either so	pt of certifica ns Commission's st to provide o ubsequent to th on of the Speci	tion in conforman Rules and Regula r change a Restor e issuance of an al Access Service	ace ations. ation Access
		Nonrec	urring Charge	
(CR)	AUG هر By	2 1991 <u>FRS# 9.01</u> ce Commission SQUARAT a time	8 6 - 8 notiiconsecutive	WITELSSION
•	Issued: JUN 27 1986 By R. D. BARRON, Preside Southwestern Bell To St. Louis, M	elephone Compan	vision	

Access Services Tariff Section 13 7th Revised Sheet 8 Replacing 6th Revised Sheet 8

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

### 13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing

### A. GENERAL DESCRIPTION

Easy Access Dialing is an arrangement whereby an end user customer (herein referred to as customer) of the Telephone Company or a local service provider that resells services of the Telephone Company for Telephone Exchange Service (either lines or trunks), FGA and BSA-A lines, WATS Access Lines and Plexar lines may select and designate to the Telephone Company an Interexchange Carrier (IC) to access, without dialing an access code. These are referred to as the customer's Primary Interexchange Carrier (PIC) for interLATA toll calls and the intraLATA Primary Interexchange Carrier (LPIC) for intraLATA toll calls. The customer

may designate an IC for PIC and a different IC for LPIC.

The end user or customer is identified as the person on the account who is responsible for payment of the account or any person contractually or otherwise lawfully authorized to change

telecommunications services and/or represent the end user customer.

Each IC will have one or more access codes assigned to it for its various types of service. When a customer or agent selects an IC as its PIC and/or LPIC, only one access code of that IC may be incorporated into the switching system of the Telephone Company permitting access to that IC by the customer or agent without dialing an access code except "1+" or "0+".

Should a customer or agent wish to use other services of the same IC or services of a different IC it will be necessary for the customer or agent to dial the necessary access code(s) to reach that IC's other service(s).

At the time a customer or agent advises the Telephone Company of its PIC and/or LPIC, the Telephone Company will use the IC access code for that IC from the designated carrier list unless the customer at that time directs the Telephone Company to use a different IC access code assigned to the customer's chosen IC. The PIC and LPIC may either be the same IC or a different IC.

Interstate intraLATA toll calls, service between points in different states which are in the same

Local Access and Transport Area, will be routed to the LPIC selected by the customer.

Issued: November 30, 2000



(T – ) (L – ) CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 6th Revised Sheet 8 Replacing 5th Revised Sheet 8

# ACCESS SERVICES

customer) of Telephone C BSA-A lines Company an referred to as the intraLATT)may designatT)Each IC willT)Each IC willT)When a custo IC may be in access to thaT)Should a cus IC it will be that IC's otheT)At the time a Code assigneT)At the time a IC it will be that IC's otheT)At the time a Code assigneT)Interstate int ICT)Interstate int CT)The code accessT)The code access	99	Effective:	JUL 2 2 1999
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>C) Easy Access customer) of Telephone C BSA-A lines Company an referred to as the intraLAT</li> <li>T) Each IC will</li> <li>F) At the time a Telephone C unless the cu code assigne</li> <li>F) At the time a</li> <li>F) At the time a</li> <li>F) At the time a</li> <li>F) Therestate int</li> <li>F) Interstate int</li> <li>F) Interstate int</li> </ul>	blic Service Commis MISSOURI		
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C BSA-A lines Company an referred to as the intraLAT</li> <li>Each IC will</li> <li>Each IC will</li> <li>When a custor IC may be in access to that</li> <li>Should a custor IC it will be that IC's other</li> <li>At the time a Telephone C unless the cu code assigne a different IC</li> <li>Interstate int Local Access</li> </ul>	Ly 745 R 5 8 blic Service Commis	ssion	-11(2)
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT</li> <li>Each IC will</li> <li>Each IC will</li> <li>When a custor</li> <li>IC may be in access to that</li> <li>Should a custor</li> <li>IC it will be that IC's other</li> <li>At the time a Telephone C unless the cu code assigne a different IC</li> <li>Interstate int Local Access</li> </ul>	DEC 3 0 2000		Service
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT</li> <li>may designat</li> <li>Each IC will</li> <li>When a custor</li> <li>IC may be in access to that</li> <li>Should a custor</li> <li>IC it will be that IC's other</li> <li>At the time a Telephone C unless the cu code assigned a different IC</li> <li>Interstate int</li> </ul>	UANCELLED		Missouri Fubliq
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT</li> <li>may designat</li> <li>Each IC will</li> <li>When a custor</li> <li>IC may be in access to that</li> <li>Should a custor</li> <li>IC it will be that IC's other</li> <li>At the time a Telephone C unless the cu code assigned</li> <li>a different IC</li> <li>Interstate int</li> </ul>	ind Transport Area, will be CANCELLED	e routed to the LPIC s	elected by the customer.
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT</li> <li>may designat</li> <li>Each IC will</li> <li>When a custor</li> <li>IC may be in access to that</li> <li>Should a custor</li> <li>IC it will be that IC's other</li> <li>At the time a Telephone C</li> <li>unless the cu code assigned</li> </ul>		-	different states which are in the sa
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT may designat</li> <li>Each IC will</li> <li>When a custor</li> <li>IC may be in access to that</li> <li>Should a custor</li> <li>IC it will be that IC's other</li> <li>At the time a Telephone C</li> <li>unless the cut</li> </ul>		en IC. The PIC and	I LPIC may either be the same IC o
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT</li> <li>may designat</li> <li>Each IC will</li> <li>When a custo IC may be in access to tha</li> <li>Should a custo IC it will be that IC's other</li> </ul>	ompany will use the IC a stomer at that time direc	access code for that ts the Telephone Co	IC from the designated carrier list ompany to use a different IC access
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT may designat</li> <li>Each IC will When a custo IC may be in access to tha</li> <li>Should a cuss IC it will be</li> </ul>		es the Telenhone C	ompany of its PIC and/or LPIC, the
13.3.3 Easy Access A. GENERAL E Easy Access customer) of Telephone C BSA-A lines Company an referred to as the intraLAT may designat Each IC will When a custo IC may be in	ecessary for the custom		the same IC or services of a difference of a difference of a difference of the necessary access code(s) to reac
13.3.3 Easy Access A. GENERAL E Easy Access customer) of Telephone C BSA-A lines Company an referred to as the intraLAT may designat Each IC will When a custo			g an access code except "1+" or "0
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines Company an referred to as the intraLAT</li> <li>may designat</li> </ul>	mer or agent selects an	IC as its PIC and/or	it for its various types of service. LPIC, only one access code of that Telephone Company permitting
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines</li> <li>Company an referred to as the intraLAT</li> </ul>	e an IC for PIC and a di		
<ul> <li>13.3.3 Easy Access</li> <li>A. GENERAL E</li> <li>Easy Access customer) of Telephone C</li> <li>BSA-A lines</li> </ul>	the customer's Primary A Primary Interexchang	Interexchange Carr e Carrier (LPIC) for	ier (PIC) for interLATA toll calls a r intraLATA toll calls. The custom
13.3.3 Easy Access A. GENERAL D Easy Access customer) of	WATS Access Lines ar	nd Plexar lines may	select and designate to the Telepho ut dialing an access code. These ar
13.3.3 Easy Access A. GENERAL E	he Telephone Company	or a local service p	user customer (herein referred to as provider that resells services of the ither lines or trunks), FGA and
13.3 Miscellaneous S	Dialing		MU. PUBLIC SERVICE CUI
	rvices-(Continued)		JUN 2 2 1999
(Continued)			RECEIVED

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

همتها أيجر

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 5th Revised Sheet 8 Replacing 4th Revised Sheet 8

ACCESS SERVICES

- ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES 13. (Continued) MAR 29 1993
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.3 Easy Access Dialing

MISSOURI **Public Service Commission** 

Easy Access Dialing is an arrangement whereby a customer for Telephone Α. Exchange Service lines and/or trunks, FGA lines, BSA-A lines, and Centrex lines, herein referred to as customer, or the person or persons who have legal authority to give the Telephone Company permission to place public and semipublic pay telephones on their premise(s) and who control access to or usage of the public or semipublic pay telephones, herein referred to as agent, may select and designate to the Telephone Company an IC to access, without an access code, for interLATA intrastate calls. This IC is referred to as the customer's or agent's primary IC and must be the same for both interLATA interstate and interLATA intrastate calls.

Each IC will have one or more access codes assigned to it for its various types of service. When a customer or agent selects an IC as its primary IC, only one access code of that IC may be incorporated into the switching system of the Telephone Company permitting access to that IC by the customer or agent without dialing an access code.

Should a customer or agent wish to use other services of the same IC it will be necessary for the customer or agent to dial the necessary access code(s) to reach that IC's other service(s).

(AT)

CANCELLED At the time a customer or agent advises the Telephone Company of its JUL 2 2 1999 the Telephone Company which IC access code, if the IC has more than one access code, the customer or agent will predesignate. By LORS #8

Public Service CommissionAccess Dialing for Telephone Exchange Service lines and for MISSOUR trunks, Feature Group A lines and Centrex lines is furnished in accordance with the detailed provisions of the Federal Communications Commission's Allocation Plan as set forth in Appendix B of its Memorandum Opinion and Order in CC Docket No. 83-1145, Phase I, adopted May 31, 1985, and released June 12, 1985. Easy Access Dialing for the Circuit Switched - Line Side Basic Serving Arrangement is furnished in accordance with the provisions for FGA as set forth above. This Order with all Appendices will be available for inspection in the Public Reference Room of the Tariff Division at the main building Efithe Commission and can also be obtained from the Commission's commercial contractor.

APR 11 1993 92 - 304MO. PUBLIC SERVICE COMM.



(AT)

(AT)

(MT)

Issued:

Effective:

1988 11 11 11 11993

MAR 2 6 1993 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 4th Revised Sheet 8 Replacing 3rd Revised Sheet 8

ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)
- (CP) 13.3.3 Basy Access Dialing

FEB 17 1989

- MISSOURI
- A. Basy Access Dialing is an arrangement whereby a customer for Telephone Exchange Service lines and/or trunks, Feature Group A lines and Centrex lines, herein referred to as customer, or the person or persons who have legal authority to give the Telephone Company permission to place public and semipublic pay telephones on their premise(s) and who control access to or usage of the public or semipublic pay telephones, herein referred to as agent, may select and designate to the Telephone Company an IC to access, without an access code, for interLATA intrastate calls. This IC is referred to as the customer's or repet's primary IC and must be the same for both interLATA interstate calls.

Each IC will have one or more access codes assigned to it for itsh as commission various types of service. When a customer or agent selects and to as commission its primary IC, only one access code of that IC may be incorporated with into the switching system of the Telephone Company permitting access to that IC by the customer or agent without dialing an access code.

Should a customer or agent wish to use other services of the same IC it will be necessary for the customer or agent to dial the necessary access code(s) to reach that IC's other service(s).

At the time a customer or agent advises the Telephone Company of its primary IC, it will be necessary for the customer or agent to inform the Telephone Company which IC access code, if the IC has more than one access code, the customer or agent will predesignate.

B. Easy Access Dialing for Telephone Exchange Service lines and for trunks, Feature Group A lines and Centrex lines is furnished in accordance with the detailed provisions of the Federal Communications Commission's Allocation Plan as set forth in Appendix B of its Memorandum Opinion and Order in CC Docket No. 83-1145, Phase I, adopted May 31, 1985 and released June 12, 1985. This Order with all Appendices will be available for inspection in the Public Reference Room of the Tariff Division at the main building of the Commission and can also be obtained from the Commission's commercial contractor.

Basy Access Dialing for agents of public and semipublic pay telephones is furnished using guidelines as detailed in the Federal Computications Commission's Allocation Plan set forth above, except the six month notification provided to interexchange carriers of endMotifices 1989 converting to equal access prior to June 1, 1989.

Issued: FEB 27 1989

Public Service Commission Effective: MAR \$0 1989
Access Services Tariff Section 13 3rd Revised Sheet 8 Replacing 2nd Revised Sheet 8

## RECEIVED

#### ACCESS SERVICES

DEC 29 1987

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVESSOURI (Continued) Public Service Commission
  - 13.3 Miscellaneous Services-(Continued)

#### (CP) 13.3.3 Easy Access Dialing

A. Easy Access Dialing is an arrangement whereby an End User may select and designate to the Telephone Company an IC to access, without an access code, for interLATA intrastate calls. This IC is referred to as the End User's primary IC and must be the same for both interLATA interstate and interLATA intrastate calls.

Each IC will have one or more access codes assigned to it for its various types of service. When a customer selects an IC as its primary IC, only one access code of that IC may be incorporated into the switching system of the Telephone Company permitting access to that IC by the customer without dialing an access code.

Should the same customer wish to use other services of the same IC it will be necessary for the customer to dial the necessary access code(s) to reach that IC's other service(s).

At the time a customer advises the Telephone Company of its primary IC, it will be necessary for the customer to inform the Telephone Company which IC access code, if the IC has more than one access code, the customer will predesignate.

B. Easy Access Dialing is furnished in accordance with the detailed provisions of the Federal Communications Commission's Allocation Plan as set forth in Appendix B of its Memorandum Opinion and Order in CC Docket No. 83-1145, Phase I, adopted May 31, 1985 and released June 12, 1985. This Order with all Appendices will be available for inspection in the Public Reference Room of the Tariff Division at the main building of the Commission and can also be obtained from the Commission's commercial contractor.



(CP)

Ac	cess	Service	s Tari:	££
			tion	13
	2nd	Revised	Sheet	8
Replacing	lst	Revised	Sheet	8

ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS (Continued) 13.3 Miscellaneous Services-(Continued) 13.3.3 Easy Access Dialing MISSUURI Public Service Commission

- A. Easy Access Dialing is an arrangement whereby an End Üser may select and designate to the Telephone Company an IC to access, without an access code, for interLATA intrastate calls. This IC is referred to as the End User's primary IC and must be the same for both interLATA interstate and interLATA intrastate calls.
- B. Prior to the introduction of Easy Access Dialing in a serving end office, the existing End Users have Access Service to AT&T as their primary IC. With the introduction of Easy Access Dialing in a serving end office, End Users will be asked to select one of the following options at no charge:
  - Designate an IC as their primary IC from a list of ICs offering service in the End Users serving end office and dial 10XXX or other access codes to reach other ICs.
  - Designate that they do not want to have a primary IC and choose to dial 10XXX or other access codes for all calls for all ICs.

If the End User does not choose one of the preceding options prior to the Easy Access conversion date, an IC will be randomly assigned to them using the allocation process mandated by the Federal Communications Commission in the Interstate Access Tariff, FCC No. 68. The End User will then have (6) months after the conversion date in which they can choose one of the preceding options at no charge. Thereafter, a nonrecurring charge as set forth in Paragraph 13.3.3, D., following, applies for any change in Easy Access Dialing.

CANCELLED JUH 1 1986 Public Service Commission 86-84 Public Service Commission . MISSOURI 1 1986 JUL Issued: Effective: JUN 27 1986 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 8 Replacing Original Sheet 8

AUG 1 4 1984

MISSOURI

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANE AS ERVECTS [] (Continued)
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.2 Restoration Priority

The Telephone Company will arrange a Special Access Service Commission Restoration Priority on receipt of certification in conformance with the Federal Communications Commission's Rules and Regulations. A charge applies when a request to provide or change a Restoration Priority is received subsequent to the issuance of an Access Order to install the service. No charge applies when a Restoration Priority is discontinued.

Nonrecurring Charge

Restoration Priority, per service arranged

\$117.45

13.3.3 Presubscription

(CT)

- A. Presubscription is an arrangement whereby an End User may select and designate to the Telephone Company an IC to access, without an access code, for interLATA intrastate calls. This IC is referred to as the End User's predesignated IC and must be the same for both interLATA interstate and interLATA intrastate calls.
- B. On the effective date of this Tariff, all existing End Users will have Access Service to AT&T as their predesignated IC. Within six (6) months after the introduction of Feature Group D in a serving end office, End Users may select one of the following options at no charge. Thereafter a nonrecurring charge as set forth in Paragraph 13.3.3, D., following, applies for any change in Presubscription.
  - Designate an IC as predesignated IC and dial 10XX or other access codes to reach other IC's.
  - Designate that they do not want to be presubscribed to any IC and choose to dial 10XX or other as real for all calls for all IC's, including AT&T.

		J <b>UL</b> 1 1986	FILED
		BY 2ml RS.8	SEP 14 1984
		PUBLIC SERVICE COMMISSION	
	······································	OF MISSOUR	Public Service Commission
Issued:	AUG 1 5 1984	Effective: SEP	1 4 1984
	By	R. D. BARRON, President-Missouri	Division
	Southwestern Bell Telephone Company		
	St. Louis, Missouri		

Access Services Tariff Section 13 Original Sheet 8

#### ACCESS SERVICES

ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-13. (Continued)

13.3 Miscellaneous Services-(Continued)

DEC 2

13.3.2 Restoration Priority

101 The Telephone Company will arrange a Special Access Service for Restoration Priority on receipt of certification in conformance with the Federal Communications Commission's Rules and Regulations. A charge applies when a request to provide or change a Restoration Priority is received subsequent to the issuance of an Access Order to install the service. No charge applies when a Restoration Priority is discontinued. ANGELLED



\$117.45

Nonrecurring Charge

JAN 0 1 1984

83-253

#### 13.3.3 Presubscription

- Presubscription BY BY SERVICE COMMISSION Presubscription is an Farthangementer Whereby an End User may select and A. . designate to the Telephone Company an IC to access, without an access code, for interLATA intrastate calls. This IC is referred to as the End User's predesignated IC and must be the same for both interLATA interstate and interLATA intrastate calls.
- On the effective date of this Tariff, all existing End Users will have Β. Access Service to AT&T as their predesignated IC. Within 90 days after the introduction of Feature Group D in a serving end office, End Users may select one of the following options at no charge. Thereafter a nonrecurring charge as set forth in Paragraph 13.3.3, D., following, applies for any change in Presubscription.
  - Designate an IC as predesignated IC and dial 10XX or other access codes to reach other IC's.
  - Designate that they do not want to be presubscribed to any IC and choose to dial 10XX or other access codes for all calls for all IC's, including AT&T.

Effective:

DEC 2 9 1983 Issued:

> By R. D. BARRON, Vice President-Missouri Public So: Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 3rd Revised Sheet 8.01 Replacing 2nd Revised Sheet 8.01

### ACCESS SERVICES

### 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

B. PIC and LPIC Assignments

The Telephone Company will make changes in the customer's PIC and LPIC assignment pursuant to (1) through (2) following:

1. <u>New Service Requests</u>

New customers will be asked to select a PIC and LPIC when they place an order with the Telephone Company for Telephone Exchange Service, FGA, BSA-A, WATS Access Line Service and Plexar Service.

For Telephone Exchange service, FGA, BSA-A, and WATS Access Line Service, customers may select a separate PIC and LPIC for each line requested.

For Plexar service, customers may select their PIC and LPIC for a single Plexar line, multiple Plexar Lines or the Plexar Group. A Plexar Group defines a group of Plexar facilities (i.e., line, trunks, attendant features, etc.) with a common dialing pattern. A customer may have multiple Plexar Groups.

If a new customer fails to make a selection when placing an order for Telephone Exchange Service, FGA or BSA-A lines, WATS Access Line Service or Plexar Service, the customer will be allowed a six month period from installation of their service to select a PIC and LPIC at no charge. Any changes made after the six month period has expired or any change subsequent to the initial PIC and LPIC selection during the six month period will incur Easy Access Dialing Change charges as set forth in Section 13.4.3, following, specific to the line or lines being changed.

New customers may also designate that they do not want a PIC and/or LPIC. This choice (No-PIC and No-LPIC) is considered a valid PIC and LPIC selection. Any subsequent changes will incur Easy Access Dialing Change charges as set forth in Section 13.4.3, following, specific to the line or lines being changed.

On all requests for new service, the Telephone Company will notify the customer of the PIC and LPIC assigned to the relevant line(s).

Issued: JUNE 22, 1999

Effective: JULY 22, 1999

FILED MO PSO

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

(ĊT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

		Acce	ess Servi	ic <b>es</b> Ta	ariff
				Sectio	on 13
		2nd	Revised	Sheet	8.01
	Replacing	1st	Revised	Sheet	8.01
C					

ACCESS SERVICES

- ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES 13. (Continued)
  - MAR 29 1993

13.3.3 Easy Access Dialing-(Continued)

13.3 Miscellaneous Services-(Continued)

MISSOURI Public Service Commission

- (MT) (MT)
- Easy Access Dialing for agents of public and semipublic pay telephones is furnished using guidelines as detailed in the Federal Communications Commission's Allocation Plan set forth above, except the six month notification provided to interexchange carriers of end offices converting to equal access prior to June 1, 1989.
- C. Principal Provisions of the Allocation Plan:
  - 1. The Telephone Company will notify customers and agents of the availability of equal access and Easy Access Dialing in their particular areas through the mailing of an Equal Access Ballot. The mailing of initial ballots will occur no later than 85 days prior to the conversion of the end office serving the customers and agents. Customers and agents will be asked to return their respective ballot within 30 days of its receipt.

Customers and agents may select one of the following options at no charge:

- indicate a primary IC for all of its lines
- indicate a different IC for each of its lines

Only one carrier may be selected for each particular line.

The Telephone Company will accept IC provided lists of customers 2. and agents that have made individual arrangements with a specific IC to designate that IC as their primary long distance carrier. CANCELLED However, an IC must also submit a Limited Blanket Agency Agreement affirming that it does have authorization to act on behalf of the customer or agent to establish Easy Access Dialing Service. When JUL 2 2 1999 customer or agent choice discrepancies occur, it may be necessary for the IC to furnish evidence of proper authorization upon request JY 3 RS # 8.01 -ublic Service Commission the Telephone Company.

MISSOURI

3. The Telephone Company will tabulate the initial ballots and the lists of customers and agents provided by ICs to determine the percentage of customers and agents that selected each IC. ICs that wish to receive allocated traffic must meet and maintain the ordered allocation eligibility criteria for the two year period following the equal access conversion date. If an IC wishes to change any of the agreed upon criteria within the two year period,

(MT)

Issued:

MAR'2 6 1993

Effective:

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

MO PUBLIC SERVICE COMM.

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 1st Revised Sheet 8.01 Replacing Original Sheet 8.01

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES VED (Continued)
  - 13.3 Miscellaneous Services-(Continued)
- (CP) 13.3.3 Easy Access Dialing-(Continued)

MISSOURI Public Service Commission

BY 2miR.S

FEB 17 1989

- C. Principal provisions of the Allocation Plan:
  - 1. The Telephone Company will notify customers and agents of the availability of equal access and Easy Access Dialing in their particular areas through the mailing of an Equal Access Ballot. The mailing of initial ballots will occur no later than 85 days prior to the conversion of the end office serving the customers and agents. Customers and agents will be asked to return their respective ballot within 30 days of its receipt.

Customers and agents may select one of the following CANCELLED charge: APR 11 1993 # 8.01

- indicate a primary IC for all of its lines
- indicate a different IC for each of its lines

Only one carrier may be selected for each part Public Service Commission MISSOURI

- The Telephone Company will accept IC provided lists of customers 2. and agents that have made individual arrangements with a specific IC to designate that IC as their primary long distance carrier. However, an IC must also submit a Limited Blanket Agency Agreement affirming that it does have authorization to act on behalf of the customer or agent to establish Basy Access Dialing Service. When customer or agent choice discrepancies occur, it may be necessary for the IC to furnish evidence of proper authorization upon request by the Telephone Company.
- 3. The Telephone Company will tabulate the initial ballots and the lists of customers and agents provided by ICs to determine the percentage of customers and agents that selected each IC. IC's that wish to receive allocated traffic must meet and maintain the ordered allocation eligibility criteria for the two year period following the equal access conversion date. If an IC vishes to change any of the agreed upon criteria within the two year period, it must notify the telephone company and its allocated customers and agents, of those changes 30 days before the changes are to take place. ICs participating in the allocation process will have customers and agents that have not selected a primary IC assigned, at random, in proportion to the results of the firsting to response for a particular serving end office.

Issued: FEB 27 1989

MAR 3 0 1089 Effective: MAR 90 1989 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 8.01

#### ACCESS SERVICES

## RECEIVED

ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-13. (Continued)

**MISSOURI** Public Service Commission

BY LOARS # 8.01

13.3 Miscellaneous Services-(Continued)

(CP) 13.3.3 Easy Access Dialing-(Continued)

C. Principal provisions of the Allocation Plan:

The Telephone Company will notify customers of the availability 1. of equal access and Easy Access Dialing in their particular areas through the mailing of an Equal Access Ballot. The mailing of initial ballots will occur no later than 85 days prior to the conversion of the end office serving the customers. Customers will be asked to return their respective ballot within 30 days of Customers may select one of the following options at no charge: - indicate a primary IC for all of its lines MAR 30 1989 - indicate a different IC for cash its receipt.

Only one carrier may be selected for each particulation

- 2. The Telephone Company will accept IC provided lists of customers that have made individual arrangements with a specific IC to designate that IC as their primary long distance carrier. However, an IC must also submit a Limited Blanket Agency Agreement affirming that it does have authorization to act on behalf of the customer to establish Easy Access Dialing Service. When customer choice discrepancies occur, it may be necessary for the IC to furnish evidence of proper authorization upon request by the Telephone Company.
- The Telephone Company will tabulate the initial ballots and the 3. lists of customers provided by ICs to determine the percentage of customers that selected each IC. ICs that wish to receive allocated traffic must meet and maintain the ordered allocation eligibility criteria for the two year period following the equal access conversion date. If an IC wishes to change any of the agreed upon criteria within the two year period, it must notify the telephone company and its allocated customers of those changes 30 days before the changes are to take place. ICs participaring in the allocation process will have customers that have not selected a primary IC assigned, at random, in proportion to the results of the first ballot response for a particular serving end office 0 1988

Issued: JAN UB 1900

Effective: FEB 0 8 1988 Ublic Service Commission

By R. D. BARRON, President-Missouri Division . Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 4th Revised Sheet 8.02 Replacing 3rd Revised Sheet 8.02

## ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Easy Access Dialing-(Continued)
    - B. PIC and LPIC Assignments-(Continued)
      - 2. Existing Service Requests
        - a. Customers of existing Telephone Exchange Service, FGA or BSA-A lines, WATS Access Line Service and Plexar Service may request changes to their PIC and LPIC according to the selection options set forth in B., 1., preceding (e.g., separate PIC and LPIC, No-PIC and No-LPIC). For these services, the Telephone Company will make PIC and LPIC changes based upon the following:
          - 1. For customer initiated requests, the Telephone Company will make PIC and LPIC changes upon notification by the customer of the change desired.
          - 2. For IC initiated changes of customer PIC and/or LPIC assignments, the Telephone Company will make PIC and/or LPIC changes pursuant to an IC-provided list of customers accepted by the Telephone Company under a Limited Blanket Agency Agreement. When an IC relies on a Letter of Authorization (LOA) to make a change it must obtain from the customer a signed LOA authorizing the change. LOAs or other valid forms of authorization based on Missouri and/or Federal regulations, must not be more than 45 days old when such changes are submitted to the Telephone Company.

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247 (RT)

Issued: November 30, 2000

Effective: December 30, 2000



(CT)

Access Services Tariff Section 13 3rd Revised Sheet 8.02 Replacing 2nd Revised Sheet 8.02

JUN 2 2 1999

NO. PUBLIC SERVICE COMM

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) RECEIVED
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.3 Easy Access Dialing-(Continued)
    - PIC and LPIC Assignments-(Continued) Β.

2. **Existing Service Requests** 

- a. Customers of existing Telephone Exchange Service, FGA or BSA-A lines, WATS Access Line Service and Plexar Service may request changes to their PIC and LPIC according to the selection options set forth in B., 1., preceding (e.g., separate PIC and LPIC, No-PIC and No-LPIC). For these services, the Telephone Company will make PIC and LPIC changes based upon the following:
  - 1. For customer initiated requests, the Telephone Company will make PIC and LPIC changes upon notification by the customer of the change desired.
  - 2. For IC initiated changes of customer PIC and/or LPIC assignments, the Telephone Company will make PIC and/or LPIC changes pursuant to an IC-provided list of customers accepted by the Telephone Company under a Limited Blanket Agency Agreement. When an IC relies on a Letter of Authorization (LOA) to make a change it must obtain from the customer a signed LOA authorizing the change. LOAs or other valid forms of authorization based on Missouri and/or Federal regulations, must not be more than 45 days old when such changes are submitted to the Telephone Company.
- b. Unauthorized PIC and LPIC Charge

An unauthorized PIC and/or LPIC change occurs when a customer's PIC and LPIC is changed without proper authorization by the customer. When a customer notifies the Telephone Company within 12 months of the date that its PIC and/or LPIC has been changed without its consent, the Telephone Company will notify the disputed PIC and/or LPIC of the dispute and revert the customer to its previous PIC and/or LPIC at no charge to the customer except as set forth in this section. The Telephone Company will also credit the customer's account for the disputed PIC and/or LPIC change.

The Telephone Company will request evidence of proper customer authorization from the disputed IC or agent that requested the PIC and/or LPIC change. If the IC or agent does not provide a LOA or other valid forms of authorization based on Missouri and/or Federal regulations, with a customer authorization date less than or equal to 45 days old from the date the change was submitted to the Telephone Company, Easy Access Dialing Change charges as set forth in Section 13.4.3, following, will be assessed to the disputed IC or agent. If the IC or agent provides a LOA or other valid forms of authorization based on Missouri and/or Federal regulations, appropriately signed and dated as described above, the Telephone Company will assess applicable PIC and/or LPIC Easy Access Dialing Change charges to the customer. Sanice

In lieu of the above LOA investigation, the IC may choose to participate in PIC and/or ~[[]]] : 2 / 1449 LPIC Switchback.

Issued:

(ĊT)

0 2000

DEC

JUN 2 2 1999

Effective:

JUL 2 2 1999

		· · · · · · · · · · · · · · · · · · ·
No supplement to the tariff will be issue except for the purp of canceling this t	led bose	Access Services Tariff Section 13 2nd Revised Sheet 8.02 lacing 1st Revised Sheet 8.02
13. ADDITIONAL ENG (Continued)	INEERING, ADDITIONAL LABOR AND MIS	SCELLANEOUS SERVICES- RECEIVED
13.3 Miscellaneo	ous Services-(Continued)	MAR 29 1993
-	cess Dialing-(Continued)	NACES ALL ST
C. Principal	Provisions of the Allocation Plan	n:-(Continued)Sanios Commission
3. (Cont	inued)	
and a place custo at ra	ast notify the Telephone Company an agents, of those changes 30 days be e. ICs participating in the allocation omers and agents that have not select andom, in proportion to the results onse for a particular serving end of	efore the changes are to take ation process will have ected a primary IC assigned, s of the first ballot
sent than vill	omers and agents not responding to a second ballot for the selection 40 days prior to end office conve indicate the primary IC that has to respond to the second ballot.	of a primary IC no earlier rsion. This second ballot
in the lines	an IC listed on the initial ballo ne allocation process, the nonresp s will be allocated to the remaining neir initial results.	onding customers' and agents'
	rate allocation processes will be ness lines, and public and semipub	
desig	a discrepancy is determined regar gnation of a primary IC, the follo upon the situation encountered:	
	nen a customer or agent indicates ine on a ballot, or returns an ill ompany will contact the customer o	egible ballot, the Telephone
by 3 RS # 8.67 as rublic Service Commission na MISSOURI Co	nen both a ballot and an entry on gent list are received for one cus ated primary IC does not match on ompany will process the ballot as noice.	tomer or agent and the desig- both documents, the Telephone
Te	nen two or more ICs provide custom elephone Company indicating that a as designated them as its primary	a particular customer or agent
Issued: MAR 2 6 19	Effective:	
	By R. D. BARRON, President-Missou Southwestern Bell Telephone C St. Louis, Missouri	Company

**...** 

MO. PUBLIC SERVICE COMM.

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 1st Revised Sheet 8.02 Replacing Original Sheet 8.02

#### ACCESS SERVICES

 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVECEIVED (Continued)
 13.3 Miscellaneous Services-(Continued)ANCELLED
 (CP)
 13.3.3 Easy Access Dialing-(Continued APR 11 1993, MISSOURI C. Principal Provisions of the AllExtrem Provision of the AllExtrem Provision of the AllExtrem Provision of the AllExtrem Provision of the AllExtrem Public Service Commission MISSOURI MISSOURI
 3. (Continued)

> Customers and agents not responding to the initial ballot will be sent a second ballot for the selection of a primary IC no earlier than 40 days prior to end office conversion. This second ballot will indicate the primary IC that has been assigned to them if they fail to respond to the second ballot.

> When an IC listed on the initial ballot elects not to participate in the allocation process, the nonresponding customers' and agents' lines will be allocated to the remaining IC's in relative proportion to their initial results.

Separate allocation processes will be used for residence and business lines, and public and semipublic pay telephone lines.

- 4. When a discrepancy is determined regarding a customer's or agent's designation of a primary IC, the following conditions apply depending upon the situation encountered:
  - When a customer or agent indicates more than one IC choice per line on a ballot, or returns an illegible ballot, the Telephone Company will contact the customer or agent for clarification.
  - When both a ballot and an entry on an IC provided customer and agent list are received for one customer or agent and the designated primary IC does not match on both documents, the Telephone Company will process the ballot as the customer's or agent's choice.
  - When two or more IC's provide customer and agent lists to the Telephone Company indicating that a particular customer or agent has designated them as its primary IC, the customer or agent will be allocated along with nonrespondents to the initial ballot. In such instances, the Telephone Company will morify the customer or agent and the involved IC's of the conflict.

MAR 3 0 1989

Public Service Commission

Issued: FEB 27 1989

(MT)

## Bffective: MAR 30 1989

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 8.02

#### ACCESS SERVICES

ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICE CEIVED (Continued)

13.3 Miscellaneous Services-(Continued)

DEC 29 1987

(CP) 13.3.3 Easy Access Dialing-(Continued)

- MISSOURI Public Service Commission
- C. Principal Provisions of the Allocation Plan:-(Continued)
  - 3. (Continued)

Customers not responding to the initial ballot will be sent a second ballot for the selection of a primary IC no earlier than 40 days prior to end office conversion. This second ballot will indicate the primary IC that has been assigned to them if they fail to respond to the second ballot.

When an IC listed on the initial ballot elects not to participate in the allocation process, the nonresponding customers' lines will be allocated to the remaining ICs in relative proporticELLEDir initial results.

£.

Separate allocation processes will be used for restance and BY 104 R.S.#8.02 business customer lines.

- 4. When a discrepancy is determined regarding a current of a primary IC. the following conditions of a primary IC, the following conditions apply depending opon the situation encountered:
  - When a customer indicates more than one IC choice per line on a ballot, or returns an illegible ballot, the Telephone Company will contact the customer for clarification.
  - When both a ballot and an entry on an IC provided customer list are received for one customer and the designated primary IC does not match on both documents, the Telephone Company will process the ballot as the customer's choice.
  - When two or more ICs provide customer lists to the Telephone Company indicating that a particular customer has designated them as its primary IC, the customer will be allocated along with nonrespondents to the initial ballot. In such instances, the Telephone Company will notify the customer and the involved ICs of the conflict.
- 5. Customers served by an end office that was converted to equal access prior to May 31, 1985, and who have not selected a being y IC, will be subject to the provisions of the Allocation Plan on a retroactive basis. FEB 8 1988

Effective: FEB 0.8 198 Public Service Commission JAN 08 1988 Issued: By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 5th Revised Sheet 8.03 Replacing 4th Revised Sheet 8.03

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Easy Access Dialing-(Continued)

## C. Rate Regulation

1. Unless otherwise specified, Easy Access Dialing Change charges apply to each PIC and LPIC selection on Telephone Exchange Service, FGA and BSA-A lines, WATS Access lines and Plexar lines, including changes to or from a No-PIC and No-LPIC designation.

Easy Access Dialing PIC and LPIC Change charges are set forth in Section 13.4.3 following. Since Easy Access Dialing PIC Change charges exist in the Telephone Company Intrastate Access Service Tariff and the Interstate Access Service Tariff, the total charge to the customer will be calculated based on one half of the intrastate charge and one half of the interstate charge.

2. Initial and Subsequent Selection

Existing customers may select an LPIC within six months from the implementation of intraLATA Dialing Parity in Missouri July 22, 1999 – January 22, 2000 without charge (No-LPIC is also a valid selection). Selection of an LPIC after the six-month period or any change to the customer's initial LPIC selection will result in LPIC Change Charges as set forth in 13.4.3, following.

Issued: November 30, 2000

Effective: December 30, 2000

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

(CT)

(CT)

Access Services Tariff Section 13 4th Revised Sheet 8.03 Replacing 3rd Revised Sheet 8.03

## ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)	RECEIVED JUN 2 2 1999	
13.3.3 Easy Access Dialing-(Continued)		
B. PIC and LPIC Assignments-(Continued)		
2. Existing Service Requests-(Continued)	MO. PUBLIC SERVICE CUMM	
c. PIC and LPIC Switchback		

PIC and LPIC Switchback is an option whereby the Telephone Company does not investigate the validity of a PIC and/or LPIC change if a business or residence customer disputes a PIC and/or LPIC change as set forth in Section 13.3.3.C.7.

C. Rate Regulation

1. Unless otherwise specified, Easy Access Dialing Change charges apply to each PIC and LPIC selection on Telephone Exchange Service, FGA and BSA-A lines, WATS Access lines and Plexar lines, including changes to or from a No-PIC and No-LPIC designation.

Easy Access Dialing PIC and LPIC Change charges are set forth in Section 13.4.3 following. Since Easy Access Dialing PIC Change charges exist in the Telephone Company Intrastate Access Service Tariff and the Interstate Access Service Tariff, the total charge to the customer will be calculated based on one half of the intrastate charge and one half of the interstate charge.

2. Initial and Subsequent Selection

Existing customers may select an LPIC within six months from the implementation of intraLATA Dialing Parity in Missouri July 22, 1999 – January 22, 2000 without charge (No-LPIC is also a valid selection). Selection of an LPIC after the six-month period or any change to the customer's initial LPIC selection will result in LPIC Change Charges as set forth in 13.4.3, following.

# CANCELLED

DEC 3 0 2000 By 5th R 5 8.03 Public Service Commission MISSOURI

Missouri Fublic

-11EP 2 2 1994

Issued: JUN 2 2 1999

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 3rd Revised Sheet 8.03 Replacing 2nd Revised Sheet 8.03

ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) RECEIVED
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.3 Easy Access Dialing-(Continued)

JUN 23 1994

MISSOURI

- C. Principal Provisions of the Allocation Plan:-(Convicte Commission
  - 4. (Continued)

will be allocated along with nonrespondents to the initial ballot. In such instances, the Telephone Company will notify the customer or agent and the involved ICs of the conflict.

5. Customers of Telephone Exchange Service lines and/or trunks, Feature Group A lines and Centrex lines served by an end office that was converted to equal access prior to May 31, 1985, and who have not selected a primary IC, will be subject to the provisions of the Allocation Plan on a retroactive basis.

Agents of public and semipublic pay telephones served by an end office that was converted to equal access prior to June 1, 1989, and who have not selected a primary IC, will be subject to the provisions of the Allocation Plan on a retroactive basis. The mailing of the initial ballots will begin on or before January 1, 1989, and be completed by May 1, 1989.

6. If an IC elects to discontinue its FGD or BSA-D service offering in the converting end office prior to the conversion date or within two years after the introduction of FGD or BSA-D in the converting end office, the IC is obligated to do the following:

Notify the Telephone Company of the cancellation of its FGD or CANCELLED BSA-D order.

JUL 2 2 1999 Contact all customers and agents in writing that have been allocated to or selected the cancelling IC as their primary IC, By Y R S # 8.03 and inform these customers and agents of the cancellation and **Public Service Commission** request the customers and agents to select a new primary IC. and inform these customers and agents of the cancellation and MISSOURI

> The canceling IC must also state to its customers and agents that it will pay for any change charge that may apply.

FILED AUC 1 7 1994 95 MO. PUBLIC SERVICE COMM JUN 2 4 1994 Issued: Effective: AUG 1 7 1996 By M. H. SCHULTEIS, Executive Director-External Affairs Southwestern Bell Telephone St. Louis, Missouri

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 13 except for the purpose 2nd Revised Sheet 8.03 of canceling this tariff. Replacing 1st Revised Sheet 8.03 ACCESS SERVICES ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-13. (Continued) RECEIVED 13.3 Miscellaneous Services-(Continued) MAR 29 1993 13.3.3 Easy Access Dialing-(Continued) MISSOURI C. Principal Provisions of the Allocation Plan:-(Continued)Service Commission 4. (Continued) will be allocated along with nonrespondents to the initial ballot. (MT) In such instances, the Telephone Company will notify the customer (MT) or agent and the involved ICs of the conflict. 5. Customers of Telephone Exchange Service lines and/or trunks, Feature Group A lines and Centrex lines served by an end office that was converted to equal access prior to May 31, 1985, and who have not selected a primary IC, will be subject to the provisions of the Allocation Plan on a retroactive basis. Agents of public and semipublic pay telephones served by an end office that was converted to equal access prior to June 1, 1989, and who have not selected a primary IC, will be subject to the provisions of the Allocation Plan on a retroactive basis. The mailing of the initial ballots will begin on or before January 1, 1989, and be completed by May 1, 1989. (AT) 6. If an IC elects to discontinue its FGD or BSA-D service offering in the converting end office prior to the conversion date or within two years after the introduction of FGD or BSA-D in the converting (AT) end office, the IC is obligated to do the following: (AT) Notify the Telephone Company of the cancellation of its FGD or CANCELLET BSA-D order. Contact all customers and agents in writing that have been allocated to or selected the cancelling IC as their primary IC, Mand inform these customers and agents of the cancellation and request the customers and agents to select a new primary IC. The canceling IC must also state to its customers and agents that it will pay for any change charge that may apply. PUD **Change Charges** D. 1. Easy Access Dialing change charges, as set forth in Haragraph 13.3.3, F., will be applied as follows: (MT) ≀ 1 1 1993 **Issued:** Effective: MAR 2 6 1993 MO. PUBLIC SERVICE COMM. APR 1 1 1993 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(MT)

(MT)

Access Services Tariff Section 13 1st Revised Sheet 8.03 Replacing Original Sheet 8.03

APR 11 1993 # 03

ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-CANCELLED (Continued) RECEIVED
  - 13.3 Miscellaneous Services-(Continued)
- (CP) 13.3.3 Easy Access Dialing-(Continued)

FEB 1 7 1989

- 3.3.3 Easy Access Dialing-(Continued) APR 11 15 (MISSOURI MISSOURI
  C. Principal Provisions of the Allocation Plant (Dorfhund Dic Service Commissic)
  5. Customers of Telephone Exchange Stavice Using A lines and Centrex lines served by an end office that vas converted to equal access prior to May 31, 1985, and who have not selected a primary IC, will be subject to the provisions of the Allocation Plan on a retroactive basis.

Agents of public and semipublic pay telephones served by an end office that was converted to equal access prior to June 1, 1989, and who have not selected a primary IC, will be subject to the provisions of the Allocation Plan on a retroactive basis. The mailing of the initial ballots will begin on or before January 1, 1989 and be completed by May 1, 1989.

- 6. If an IC elects to discontinue its Feature Group D service offering in the converting end office prior to the conversion date or within two years after the introduction of Feature Group D in the converting end office, the IC is obligated to do the following:
  - Notify the Telephone Company of the cancellation of its Feature Group D order.
  - Contact all customers and agents in writing that have been allocated to or selected the cancelling IC as their primary IC, and inform these customers and agents of the cancellation and request the customers and agents to select a new primary IC.
  - The cancelling IC must also state to its customers and agents that it will pay for any change charge that may apply.
- D. **Change Charges** 
  - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. vill be applied as follows:
    - An Basy Access Dialing change charge does not apply to any changes made prior to the end office equal access coave Dion a. date.

MAR 3 0 1989

Public Service Commission

Issued: FEB 27 1989

(MT)

Effective: MAR 30 1989 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 8.03

#### ACCESS SERVICES

3. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICESDEC 291987 (Continued)

13.3 Miscellaneous Services-(Continued)

## MISSOURI Public Service Commission

RECEIVED

- (CP) 13.3.3 Easy Access Dialing-(Continued)
  - C. Principal Provisions of the Allocation Plan:-(Continued)
    - 6. If an IC elects to discontinue its Feature Group D service offering in the converting end office prior to the conversion date or within two years after the introduction of Feature Group D in the converting end office, the IC is obligated to do the following:
      - Notify the Telephone Company of the cancellation of its Feature Group D order.
      - Contact all customers in writing that have been allocated to or selected the cancelling IC as their primary IC, and inform these customers of the cancellation and request the customers to , select a new primary IC.
      - The cancelling IC must also state to its customers that it will pay for any change charge that may apply.
  - D. Change Charges
    - 1. Easy Access Dialing change charges, as set forth i CANGELLED 13.3.3., F. will be applied as follows:
      - a. An Easy Access Dialing change charge does not apply to A SH89 any changes made prior to the end office equal Brees Commission conversion date.

Public Source of the equal access not designate a primary IC prior to the equal access conversion date, the customer may make an initial primary IC selection at no charge during the six month period following the equal access conversion date. Customers will incur an Easy Access Dialing change charge for any subsequent changes. An Easy Access Dialing change charge will apply for any change made to the customer's primary IC after the six-month period has expired.

b. The Easy Access Dialing change charge does apply if an IC discontinues its Feature Group D offering within two years after the introduction of Feature Group D in the converting end office. The IC will be billed this change charge for each end user that is designated to the IC.

Issued: JAN V & 1300

Effective: FEB 0 8 1988 Service Commission

Access Services Tariff Section 13 1st Revised Sheet 8.0301 Replacing Original Sheet 8.0301

### ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
- 13.3.3 Easy Access Dialing-(Continued)
- C. Rate Regulations-(Continued)
  - 3. Telephone Exchange Service lines, FGA and BSA-A lines, and WATS Access lines

For a single PIC or LPIC change, the per PIC or LPIC change charge applies. For multiple PIC and/or LPIC changes for a single end user customer requested on a single order, the per PIC and/or LPIC change charge applies to the first PIC or LPIC changed and the per supplemental change charge applies to each additional PIC and/or LPIC changes. For PIC and/or LPIC changes requested on multiple access lines on a single order, the per PIC or LPIC change charge for a single end user customer applies to the first PIC and/or LPIC change charge for a single end user customer applies to the first PIC and/or LPIC changed and the per supplemental change charge applies to each additional PIC and/or LPIC changed and the per supplemental change charge applies to each additional PIC and/or LPIC.

4. Plexar lines and Plexar Groups

Easy Access Dialing Change charges are applied for each PIC and/or LPIC change on a per PIC and/or LPIC, per supplemental PIC and/or LPIC, per group and per additional group basis.

For PIC and/or LPIC changes to a single Plexar line, the per PIC and/or LPIC Change charge applies. For PIC and/or LPIC changes to multiple Plexar lines on a single order, the per PIC and/or LPIC change charge applies to the first PIC and/or LPIC changed and the supplemental PIC and/or LPIC Change charge applies to each additional PIC and/or LPIC changed.

For PIC and/or LPIC changes to a Plexar Group, the per group change charge and the per supplemental group supplemental line change charge applies as follows:

- a. For PIC and/or LPIC changes on only one Plexar Group, the per group change charge applies as well as the per group supplemental line change charge for each line in the Plexar Group that does not have an individual PIC and/or LPIC assigned to it.
- b. When a PIC and/or LPIC changes for multiple Plexar Groups, the per group change charge applies to each group changed and the per group supplemental line change charge applies to each line in the Plexar Group that does not have an individual PIC and/or LPIC assigned to it.

Issued: JUNE 22, 1999

```
Effective: JULY 22, 1999
```



(CT)

(CT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(AT)

(TA)

(MT)

(MT)

Access Services Tariff Section 13 Original Sheet 8.03.01

RECEIVED

MISSOURI

Public Service Commission

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES23 1994 (Continued)
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.3 Easy Access Dialing-(Continued)

If the presubscribed 0+ IC designated as the primary IC for 0+ and 00- traffic elects not to submit an order for its 1+ interLATA sent-paid traffic or fails to select a designated subcontractor (secondary service provider) to handle its 1+ interLATA sent-paid calls from the Telephone Company's pay telephones, the 1+ interLATA coin sent-paid traffic for that presubscribed 0+ carrier will continue to be routed to the existing default 1+ carrier (provided such carrier continues to accept it) until the presubscribed 0+ carrier is ready to handle the 1+ interLATA sent-paid traffic, or makes arrangements with another IC to handle the traffic, as set forth in Section 22.

- D. Change Charges
  - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3, F., will be applied as follows:

CANCELLED

JUL 2 2 1999

rublic Service Commission MISSOURI



MO. PUBLIC SERVICE CO

Issued: JUN 2 4 1994

Effective: AUG 1 7 1999 - D 1 1001

By M. H. SCHULTEIS, Executive Director-External Affairs Southwestern Bell Telephone St. Louis, Missouri

Access Services Tariff Section 13 5th Revised Sheet 8.04 Replacing 4th Revised Sheet 8.04

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Easy Access Dialing-(Continued)
    - C. Rate Regulations-(Continued)
      - 5. Pay Telephone SmartCoin<sup>sm</sup> Access Lines

If the customer's PIC and/or LPIC elects not to submit an order for its 1+ interLATA or intraLATA sent paid traffic from SmartCoin Access Lines or fails to select a designated subcontractor (secondary service provider) for such traffic, the 1+ interLATA or intraLATA carrier sent paid traffic will be routed to the existing default 1+ interLATA or intraLATA carrier (provided such carrier continues to accept the traffic) until the customers' PIC and/or LPIC is ready to handle the 1+ interLATA or intraLATA sent paid traffic, or makes arrangements with another IC to handle the traffic, as set forth in Section 21 Coin Services, following.

6. Discontinuance of FGD or BSA-D Service

If an IC discontinues its FGD or BSA-D service in an end office, Easy Access Dialing Change charges as set forth in Section 13.4.3, following, will apply to the IC discontinuing the FGD or BDA-D service. These charges will be assessed to the IC for each end user that is presubscribed to that IC.

The IC must contact in writing all affected customers for whom it serves as their PIC and/or LPIC. The IC must notify these customers that they must select a new PIC and LPIC and the IC will pay all related Easy Access Dialing Change Charges.

7. PIC and LPIC Disputes

The alleged unauthorized carrier will be billed the appropriate Easy Access Dialing Change Charge(s) for the alleged unauthorized change and the appropriate Easy Access Dialing Change Charge(s) to change the customer to their preferred IC. For multiple alleged unauthorized PIC changes, supplemental charges, as set forth in section 13.4.3 (A) following will apply.

Issued: November 30, 2000

Effective: December 30, 2000

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 4th Revised Sheet 8.04 Replacing 3rd Revised Sheet 8.04

## ACCESS SERVICES

Issued:			Effective:		
DEC 3 0 54b R Public Service (MISSO	DEC 54b Public Servid		<ul> <li>a. Unauthorized PIC and/or LPIC Option</li> <li>The following applies to ICs not participating in the Switchback Option set forth in 13.3.3.C.7.b (following): If the IC provides the Telephone Company with a LOA or other valid forms of authorization based on Missouri and/or Federal regulations, signed by the customer and dated no more than 45 days prior to the date the PIC or LPIC change was submitted to the Telephone Company, authorizing the change, the Telephone Company will bill the customer for each PIC and LPIC change. When multiple PIC and LPIC changes are required, first and supplemental charges, as set forth in 13.4.3.B., will apply.</li> </ul>		
Comin URI		If a PIC and/or LPIC dispute occurs, I assessed as follows:	Easy Access Dialing Change charges will be		
8.04 mission	7.	PIC and LPIC Disputes	Missouri Fublic Service Commission		
u		•	ected customers for whom it serves as their PIC ese customers that they must select a new PIC and Easy Access Dialing Change Charges.		
		Change charges as set forth in Section	A-D service in an end office, Easy Access Dialing n 13.4.3, following, will apply to the IC rvice. These charges will be assessed to the IC for that IC.		
	6.	Discontinuance of FGD or BSA-D Ser	rvice		
		intraLATA sent paid traffic from Smart subcontractor (secondary service provid intraLATA carrier sent paid traffic will intraLATA carrier (provided such carri customers' PIC and/or LPIC is ready to	cts not to submit an order for its 1+ interLATA or rtCoin Access Lines or fails to select a designated ider) for such traffic, the 1+ interLATA or I be routed to the existing default 1+ interLATA or ier continues to accept the traffic) until the o handle the 1+ interLATA or intraLATA sent paid other IC to handle the traffic, as set forth in Section		
	5.	Pay Telephone SmartCoin <sup>sm</sup> Access Lin			
C.	Rate	e Regulations-(Continued)	MO. PUBLIC SERVICE CUM		
13.3.3	Eas	y Access Dialing-(Continued)	JUN 2 2 1999		
13.3 Mi	scell	aneous Services-(Continued)	RECEIVE		
		-	LABOR AND MISCELLANEOUS SERVICES-		
	(Cont 13.3 Mi 13.3.3 C.	(Continued 13.3 Miscell 13.3.3 Eas C. Rat 5. 6.	<ul> <li>If the customer's PIC and/or LPIC electintraLATA sent paid traffic from Smart subcontractor (secondary service provintraLATA carrier sent paid traffic will intraLATA carrier (provided such carricustomers' PIC and/or LPIC is ready to traffic, or makes arrangements with an 21 Coin Services, following.</li> <li>6. Discontinuance of FGD or BSA-D Services as set forth in Section discontinuing the FGD or BDA-D service charges as set forth in Section discontinuing the FGD or BDA-D services and/or LPIC. The IC must contact in writing all aff and/or LPIC. The IC must notify the LPIC and the IC will prove il related IC.</li> </ul>		

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 3rd Revised Sheet 8.04 Replacing 2nd Revised Sheet 8.04

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) RECEIVED
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.3 Easy Access Dialing-(Continued)
      - D. Change Charges-(Continued)

MO. PUBLIC SERVICE COMM.

DEC 8 1994

- 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows: - (Continued)
  - a. An Easy Access Dialing change charge does not apply to any changes made prior to the end office equal access conversion date. If the customer or agent does not designate a primary IC prior to the equal access conversion date, the customer or agent may make an initial primary IC selection at no charge during the six month period following the equal access conversion date. Customers or agents will incur an Easy Access Dialing change charge for any subsequent changes. An Easy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired.
  - b. The Easy Access Dialing change charge does apply if an IC discontinues its FGD or BSA-D offering within two years after the introduction of FGD or BSA-D in the converting end office. The IC will be billed this change charge for each end user that is designated to the IC.

c... The Telephone Company will notify customers and agents that are allocated as set forth in C.3, and 5, preceding, of the date that the IC selected for allocation purposes will be activated. In the case of those customers or agents that will be retroactively allocated, an Easy Access Dialing change charge does not apply to any changes made during the 90-day period prior to by 4 RS\_#8.04 the allocation activation date. If the customer or agent does ublic Service Commission not designate a primary IC prior to the allocation activation date, the customer or agent may make an initial primary IC selection at no charge during the six-month period following the allocation activation date. An Easy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired. The Easy Access Dialing Change Charge may be billed to the end user, agent or IC.

(AT) (AT)

CANCELLED

JUL 2 2 1999

MISSOURI

DEC 0 8 1994 Issued:

Effective: JAN 0 8 1995 JAN 0 8 1995

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri

MISSOURI Public Service Commission

HLED

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(MT)

(MŤ)

(AT)

(AT)

Access Services Tariff Section 13 2nd Revised Sheet 8.04 Replacing 1st Revised Sheet 8.04

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.3 Easy Access Dialing-(Continued)
      - D. Change Charges-(Continued)

WISSOLDI Public Service Commission

MAR 29 1993

- 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
  - a. An Easy Access Dialing change charge does not apply to any changes made prior to the end office equal access conversion date. If the customer or agent does not designate a primary IC prior to the equal access conversion date, the customer or agent may make an initial primary IC selection at no charge during the six month period following the equal access conversion date. Customers or agents will incur an Easy Access Dialing change charge for any subsequent changes. An Easy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired.
    - b. The Easy Access Dialing change charge does apply if an IC discontinues its FGD or BSA-D offering within two years after the introduction of FGD or BSA-D in the converting end office. The IC will be billed this change charge for each end user that is designated to the IC.
    - c. The Telephone Company will notify customers and agents that are allocated as set forth in C.3, and 5, preceding, of the date that the IC selected for allocation purposes will be activated. In the case of those customers or agents that will be retroactively allocated, an Easy Access Dialing change charge does not apply to any changes made during the 90-day period prior to the allocation activation date. If the customer or agent does not designate a primary IC prior to the allocation activation date, the customer or agent may make an initial primary IC selection at no charge during the six-month period following the allocation activation date. An Easy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired. CANCELLED FILED

APR 11 1993 81995 JAN 92 - 304 # 8.04 Q. PUBLIC SERVICE COMM Public Science Commission Issued: MAR 2 6 1993 1 1 1993 MISSOURI By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Section 13

FEB 17 1989

No supplement to this Access Services Tariff tariff will be issued except for the purpose 1st Revised Sheet 8.04 of canceling this tariff. **Replacing Original Sheet 8.04** ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICESECEIVED CANCELLED (Continued) APR 11 1993 From MISSOURI BY Sund R.S. Supplic Service Commissi Public Service Commission All Structure Commission 13.3 Miscellaneous Services-(Continued) (CP) 13.3.3 Basy Access Dialing-(Continued) D. Change Charges-(Continued) Public Services Basy Access Dialing change charges, as Mset forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued) a. (Continued) (MT) If the customer or agent does not designate a primary IC prior to the equal access conversion date, the customer or agent may make an initial primary IC selection at no charge during the six month period following the equal access conversion date. Customers or agents will incur an Easy Access Dialing change charge for any subsequent changes. An Basy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired. b. The Easy Access Dialing change charge does apply if an IC discontinues its Feature Group D offering within two years after the introduction of Feature Group D in the converting end office. The IC will be billed this change charge for each end user that is designated to the IC. (MT) c. The Telephone Company will notify customers and agents that are allocated as set forth in (C)(3) and (5) preceding of the date that the IC selected for allocation purposes will be activated. In the case of those customers or agents that will be retroactively allocated, an Easy Access Dialing change charge does not apply to any changes made during the 90-day period prior to the allocation activation date. If the customer or agent does not designate a primary IC prior to the allocation activation date, the customer or agent may make an initial primary IC selection at no charge during the six-month period following the allocation activation date. An Basy Access Dialing change charge will apply for any change made to the customer spr agent's primary IC after the six-month period has expired. MAR 3 0 1099 (HT) Public Service Commission Issued: FEB 27 1989 Effective: MAR 30 1989

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 8.04

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SEFECTSIVED (Continued)
  - 13.3 Miscellaneous Services-(Continued)

DEC 29 1987

(CP) 13.3.3 Easy Access Dialing-(Continued)

**MISSOURI** Public Service Commission

- D. Change Charges-(Continued)
  - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
    - c. The Telephone Company will notify customers that are allocated as set forth in (C)(3) and (5) preceding of the date that the IC selected for allocation purposes will be activated. In the case of those customers that will be retroactively allocated, an Easy Access Dialing change charge does not apply to any changes made during the 90-day period prior to the allocation activation date. If the customer does not designate a primary IC prior to the allocation activation date, the customer may make an initial primary IC selection at no charge during the six-month period following the allocation activation date. An Easy Access Dialing change charge will apply for any change made to the customer's primary IC after the six-month period has expired.

d. If an allocated customer changes its long distance carrier because the IC no longer complies with the ordered allocation criteria and an Easy Access Dialing change charge applies, the IC will be billed the charge instead of the customer.

e. New customers, who are served by end office already equipped with Feature Group D, will be asked to select a primary IC at the time they place an order with the Telephone Company for Telephone Exchange Service or Feature Group A Switched Access Service. The Telephone Company will send the customer a confirmation notice identifying the primary IC selected by the customer. There will be no charge for the initial selection of a primary IC nor for the identification of an IC different from that given verbally at the time the customer placed the order if the customer contacts the Telephone Company after confirmation to change selection. An Easy Access Dialing change charge will apply to any projequent changes to the customer's primary IC. FILED MAR 3.0 1989

BYLAR SER.04

FEB 8 1988

Public Service Commission

Issued: JAN 0.8 1988

Public Starting MISSOUFEB 0 8 1988 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Lcuis, Missouri

Access Services Tariff Section 13 3rd Revised Sheet 8.05 Replacing 2nd Revised Sheet 8.05

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Easy Access Dialing-(Continued)
    - C. Rate Regulations-(Continued)

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

(RT)

Issued: November 30, 2000

Effective: December 30, 2000



No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 2nd Revised Sheet 8.05 Replacing 1st Revised Sheet 8.05

RECEIVED

JUN 2 2 1999

## ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

C. Rate Regulations-(Continued)

(CT)

MÖ. PUBLIC SERVICE COMM

7. PIC and LPIC Disputes-(Continued)

a. Unauthorized PIC and LPIC Option-(Continued)

If an IC that does not participate in the Switchback option set forth in (b) below does not provide the Telephone Company with an appropriately signed and dated LOA (as described above) or other valid forms of authorization based on Missouri and/or Federal regulations, authorizing the change, the Telephone Company will credit the customer's account for the disputed PIC or LPIC change charge. The IC will be held liable for 90 days. The IC will be billed for each change in the PIC and LPIC designation (i.e., two PIC and LPIC changes for each line) plus an Unauthorized PIC and LPIC change charge for each PIC and LPIC change. When multiple PIC and LPIC changes are required as a result of an Unauthorized PIC and LPIC Change, first and supplemental charges, as set forth in Section 1.3.4.3.B., will apply.

b. PIC and LPIC Switchback Option

The PIC and LPIC Switchback option is available to any IC that has a Limited Blanket Agency Agreement for submitting PIC and LPIC orders on file with the Telephone Company and that notifies the Telephone Company in writing of that IC's desire to participate in PIC and LPIC Switchback. Under this option, the Telephone Company neither requests, nor accepts, LOAs from the IC.

If the Telephone Company is notified that a customer disputes a PIC and/or LPIC change within 90 days from when the PIC and/or LPIC change was made, the Telephone Company will restore the customer's previous PIC and/or LPIC without charge. Additionally, the customer will be credited the PIC and/or LPIC change charge(s) assessed by the Telephone Company.

CANCELLED

DEC 3 0 2000

Public Service Commission MISSOURI

3NAR5 8.05

Missouri Public Service Commission

-11 HI BAR & X 1999

Issued

(CT)

JUN 2 2 1999

Effective: JUL

JUL 221999

Access Services Tariff Section 13 1st Revised Sheet 8.05 Replacing Original Sheet 8.05

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

RECEIVED

MAR 29 1993

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

D. Change Charges-(Continued)

MISSOURI Public Sarvice Commission

- 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3, F., will be applied as follows:-(Continued)
  - d. If an allocated customer or agent changes its long distance carrier because the IC no longer complies with the ordered allocation criteria and an Easy Access Dialing change charge applies, the IC will be billed the charge instead of the customer or agent.
  - New customers or agents, who are served by an end office e. already equipped with FGD or BSA-D will be asked to select a primary IC at the time they place an order with the Telephone Company for Telephone Exchange Service, FGA or BSA-A Switched Access Service, Centrex Service or Public or Semipublic Pay Telephone Service. The Telephone Company will send the customer or agent a confirmation notice identifying the primary IC selected by the customer or agent. There will be no charge for the initial selection of a primary IC nor for the identification of an IC different from that given verbally at the time the customer or agent placed the order if the customer or agent contacts the Telephone Company after confirmation to change selection. An Easy Access Dialing change charge will apply to any subsequent changes to the customer's or agent's primary IC.

CANCELLED

JUL 2 2 1999

By 2 R.S # 8.05 Public Service Commission MISSOURI

FILED

APR 1 1 1993 92 - 304 MO. PUBLIC SERVICE COMM.



(AT)

(AT)

Issued: MAR 2 6 1993

Effective:

APR 1 1 1993

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 8.05

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)
- (CP)

(MT)

(MT)

- 13.3.3 Basy Access Dialing-(Continued)
  - D. Change Charges-(Continued)

# FEB 17 1989 MISSOURI

RECEIVED

- Public Service Commission 1. Basy Access Dialing change charges, as set forth in Paragraph 13.3.3.,F. will be applied as follows:-(Continued)
  - d. If an allocated customer or agent changes its long distance carrier because the IC no longer complies with the ordered allocation criteria and an Easy Access Dialing change charge applies, the IC will be billed the charge instead of the customer or agent.
  - e. New customers or agents, who are served by an end office already equipped with Feature Group D, will be asked to select a primary IC at the time they place an order with the Telephone Company for Telephone Exchange Service, Feature Group A Switched Access Service, Centrex Service or Public or Semipublic Pay Telephone Service. The Telephone Company will send the customer or agent a confirmation notice identifying the primary IC selected by the customer or agent. There will be no charge for the initial selection of a primary IC nor for the identification of an IC different from that given verbally at the time the customer or agent placed the order if the customer or agent contacts the Telephone Company after confirmation to change selection. An Easy Access Dialing change charge will apply to any subsequent changes to the customer's or agent's primary IC.



## FILED

MAR 3 0 1989

Public Service Commission

Issued: FEB 27 1989

Effective: MAR 30 1989

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

## Missouri Public Service Commission

# REC'D NOV 3 0 2000

Access Services Tariff Section 13 7th Revised Sheet 9 Replacing 6th Revised Sheet 9

## ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Easy Access Dialing-(Continued)
    - C. Rate Regulations-(Continued)

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

8. On-Line Transfer Service

On-Line Transfer Service is an option under which the Telephone Company will provide for the transfer of a residential or business end user customer to the PIC and LPIC selected by the customer, when that IC participates in the On-Line Transfer Service.

> Missouri Public Service Commission

FILED LIEC 30 2000

Issued: November 30, 2000

Effective: December 30, 2000

Access Services Tariff Section 13 6th Revised Sheet 9 Replacing 5th Revised Sheet 9

RECEIVEN

JUN 2 2 1999

MO. PUBLIC SERVICE COMM

### ACCESS SERVICES

#### ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

(CT)

C. Rate Regulations-(Continued)

- 7. PIC and LPIC Disputes-(Continued)
  - b. PIC and LPIC Switchback Option-(Continued)

The IC who initiated the disputed PIC and/or LPIC change will be billed the PIC and/or LPIC Change charge(s) (previously credited to the customer's account) as well as the PIC and/or LPIC change charge(s) to change the end user back to their previous IC. ICs participating in the PIC and LPIC Switchback option will be assessed the applicable Easy Access Dialing Change charge(s), as set forth in Section 13.4.3.A.

The Switchback option does not relieve the IC of any of its legal, equitable, or regulatory obligations or requirements, whether interstate or intrastate in nature, including the Missouri Public Service Commission's requirements to verify all PIC and LPIC orders obtained by telemarketing prior to submitting orders to the Telephone Company and for instituting steps to obtain LOAs on PIC and LPIC orders submitted to the Telephone Company. In addition, the Switchback option does not affect a customer's legal, equitable, or regulatory remedies, including the option of initiating a complaint to the F.C.C. or relevant Missouri Public Service Commission concerning unauthorized PIC and LPIC changes.

8. On-Line Transfer Service

On-Line Transfer Service is an option under which the Telephone Company will provide for the transfer of a residential or business end user customer to the PIC and LPIC selected by the customer, when that IC participates in the On-Line Transfer Service.

# CANCELLED

DEC 3 0 2000 By 7<sup>th</sup> RS9 Public Service Commission MISSOURI

Missouri Fublic

1111 1 × 2 × 1990

Issued

(CT)

JUN 2 2 1999

Effective:

JUL 2 2 1999

P.S.C. Mo.-No. 36 Access Services Tariff No supplement to this tariff will be issued Section 13 except for the purpose 5th Revised Sheet 9 of canceling this tariff. Replacing 4th Revised Sheet 9 ACCESS SERVICES ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-13. (Continued) RECEIVED 13.3 Miscellaneous Services-(Continued) MAR 291993 13.3.3 Easy Access Dialing-(Continued) MISSOURI Public Service Commission D. Change Charges-(Continued) 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3, F., will be applied as follows:-(Continued) (Continued) e. If new customers are unable to make a selection when they place their orders for Telephone Exchange Service, FGA or BSA-A (AT) Switched Access Service, or Centrex Service, they will be allowed a six-month period from installation of their service to select a primary IC at no charge. New agents are required to make a selection when they place their order for public or semipublic pay telephone service. An Easy Access Dialing change charge will apply for any subsequent selection. An Easy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired. f. The Easy Access Dialing change charge will not apply to customers or agents who select a primary IC as part of the retroactive balloting process referred to in Paragraph 13.3.3, C.5, preceding. Single line customers, agents or multiline customers requesting g٠ an Easy Access Dialing change to only one line will incur the per line change charge. Multiline customers who request an Easy Access Dialing change to more than one line in a single request will incur the per line change charge for the first line and the CANCELLFD supplemental line change charge for each additional line as set forth in Paragraph 13.3.3, F., following. JUL 2 2 1999 Centrex customers have the capability to change their primary IC h. 3y 6 RS I 9 for a single Centrex line, multiple Centrex lines or the Centrex Lic Service Commission Group. A Centrex Group defines a group of Centrex facilities MISSOURI (i.e., lines, trunks, attendant features, etc.) with a common dialing pattern. A customer may have multiple Centrex Groups. ILED APR 11 1993

Issued: MAR 2 6 1993 Effective: MO. PUBLIC SERVICE COMM. APR 1 1 1993 Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 4th Revised Sheet 9

Replacing 3rd Revised Sheet 9

ACCESS SERVICES

- ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLAPBOUS SERVICES-(Continued) RECEIVED 13.
- FEB 17 1989 13.3 Miscellaneous Services-(Continued) APR 11 1993
- BYSTARS #9 Public Service Complexion MISSOURI MISSOURI (CP) 13.3.3 Easy Access Dialing-(Continued)
  - D. Change Charges-(Continued)
    - 1. Basy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
      - e. (Continued)

If new customers are unable to make a selection when they place their orders for Telephone Exchange Service, Feature Group A Switched Access Service, or Centrex Service, they will be allowed a six-month period from installation of their service to select a primary IC at no charge. New agents are required to make a selection when they place their order for public or semipublic pay telephone service. An Basy Access Dialing change charge will apply for any subsequent selection.

An Easy Access Dialing change charge will apply for any change made to the customer's or agent's primary IC after the six-month period has expired.

- f. The Easy Access Dialing change charge will not apply to customers or agents who select a primary IC as part of the retroactive balloting process referred to in Paragraph 13.3.3., C.5 preceding.
- g٠ Single line customers, agents or multiline customers requesting an Easy Access Dialing change to only one line will incur the per line change charge. Multiline customers who request an Easy Access Dialing change to more than one line in a single request will incur the per line change charge for the first line and the supplemental line change charge for each additional line as set forth in Paragraph 13.3.3., F., following.
- h. Centrex customers have the capability to change their primary IC for a single Centrex line, multiple Centrex lines or the Centrex Group. A Centrex Group defines a group of Centre, Collities (i.e., lines, trunks, attendant features, etc.) with a common dialing pattern. A customer may have multiple Acentic Schoups.

Public Service Commission

Issued: FEB 27 1989

Bffective: MAR 20 1989

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

(CP)

Access Services Tariff Section 13 3rd Revised Sheet 9 Replacing 2nd Revised Sheet 9

## RECEIVED

#### ACCESS SERVICES

DEC 29 1987

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES MISSOURI (Continued) Public Service Commission

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

- D. Change Charges-(Continued)
  - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)

(Continued) e.

> If new customers are unable to make a selection when they place their orders for Telephone Exchange Service, they will be allowed a six-month period from installation of their service to select a primary IC at no charge.

An Easy Access Dialing change charge will apply for any change made to the customer's primary IC after the six-month period has expired.

- The Easy Access Dialing change charge will not apply to end £. users who select a primary IC as part of the retroactive balloting process referred to in Paragraph 13.3.3., C.5 preceding.
- MANY A Start Store Commission charge. Multiline customers requesting an Easy Access BY A Store Commission charge. Multiline customers who request an Easy Access Dialing change to more than one line in a single request incur the per line change charge for the supplemental line in a supplemental lin Single line or multiline customers requesting an Easy Access Dialing change to only one line will incur the per line change Dialing change to more than one line in a single request will incur the per line change charge for the first line and the supplemental line change charge for each additional line as
  - Centrex customers have the capability to change their primary h. IC for a single Centrex line, multiple Centrex lines or the Centrex Group. A Centrex Group defines a group of Centrex facilities (i.e., lines, trunks, attendant features, etc.) with a common dialing pattern. A customer may have muchtiple Centrex Groups.

FEB 8 1988

Issued: JAN 0 8 1988

CANCELLED

Effective: FEB Public Service Commission

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri
No supplement to this Access Services Tariff tariff will be issued Section 13 except for the purpose 2nd Revised Sheet 9 of canceling this tariff. Replacing 1st Revised Sheet 9 ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) JUN 27 1986 13.3 Miscellaneous Services-(Continued) MISSUURI (CP)Public Service Commission 13.3.3 Easy Access Dialing-(Continued) С. New End Users will be asked to designate a primary IC at the time they place an order with the Telephone Company for Telephone Exchange Service. They may select either of the following options. There will be no additional charge for this initial selection. - Designate an IC as a primary IC and dial 10XXX or other access codes to reach other ICs. - Designate that they do not want a primary IC and choose to dial 10XXX or other access codes for all calls to all ICs. Subsequent to the installation of Telephone Exchange Service, a nonrecurring charge as set forth in Paragraph 13.3.3, D., following, applies for any change to the customer's primary IC. If new customers are unable to make a selection when they place their orders for Telephone Exchange Service, they will be allowed a six-month period from installation of their service to select a primary IC at no charge. An Easy Access Dialing change charge will apply for any change made to the customer's primary IC after the six-month period has expired. Single line or multiline customers requesting an Easy Access Dialing change to only one line will incur the per line change charge. Multiline customers who request an Easy Access Dialing change to more than one line in a single request will incur the per line change charge for the first line and the supplemental line change charge for each additional line as set forth in Paragraph 13.3.3, D., following. CANCELLED FEB 8, 1988 1926 Public Service Commission j) 1 86-84 MISSOURI Service Commission Issued: JUN 27 1988 JUL-- 1 1936 Effective: By R. D. BARRON, President-Missouri Division

Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 1st Revised Sheet 9 Replacing Original Sheet 9

AUG 1 - 1984

MISSOURI

#### ACCESS SERVICES

13.	ADDITIONAL ENGINEER	ING, ADDITIONAL	LABOR AND	MISCELLANEOR	国歌座\$¥/20)
	(Continued)			00	

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Presubscription-(Continued)
    - C. New End Users will be asked to presubscribe to a Public Senice Commission they place an order with the Telephone Company for Telephone Exchange Service. They may select either of the following options. There will be no additional charge for this initial selection.
      - Designate an IC as predesignated IC and dial 10XX or other access codes to reach other IC's.
      - Designate that they do not want to be presubscribed to any IC and choose to dial 10XX or other access codes for all calls to all IC's.

Subsequent to the installation of Telephone Exchange Service, a nonrecurring charge as set forth in Paragraph 13.3.3, D., following, applies for any change in presubscription.

D. The nonrecurring charge for presubscription is as follows:

Nonrecurring Charge

(CR)

Presubscription, per Telephone Exchange Service line or trunk. . . . . . . . .

\$5.00(1)

GANCELLED

JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSOURI

(AT) (1) This charge is billed to the End User which is the subscriber to the Telephone Exchange Service. This is the maximum charge to be applied each time a selection is changed.

SEP 14 1984

Issued: AUG 1 5 1984

Effective: SFR

SEP Public Service Commission

Access Services Tariff Section 13 Original Sheet 9

a de Calin

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.3 Presubscription-(Continued)
    - C. New End Users will be asked to presubscribe to an IC at the time they place an order with the Telephone Company for Telephone Exchange Service. They may select either of the following options. There will be no additional charge for this initial selection.
      - Designate an IC as predesignated IC and dial 10XX or other access codes to reach other IC's.
      - Designate that they do not want to be presubscribed to any IC and choose to dial 10XX or other access codes for all calls to all IC's.

Subsequent to the installation of Telephone Exchange Service, a nonrecurring charge as set forth in Paragraph 13.3.3, D., following, applies for any change in presubscription.

D. The nonrecurring charge for presubscription is as follows:

Nonrecurring Charge

Presubscription, per Telephone Exchange Service line or trunk. . . . . . . . .

\$29.59

NOTE: This charge is billed to the End User which is the subscriber to the Telephone Exchange Service.

GANGELLED EP14 1984 LIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984 JAN - 1 198

No Supplement to this

tariff will be issued

except for the purpose

of canceling this tariff.

Access Services Tariff Section 13 4th Revised Sheet 9.01 Replacing 3rd Revised Sheet 9.01

# ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)	RECEIVE
13.3.3 Easy Access Dialing-(Continued)	JUN 2 2 1999
C. Rate Regulations-(Continued)	

**On-Line Transfer Service** 8.

> On-Line Transfer Service is available to all ICs who participate in the Equal Access subscription process. The IC has the option to have either residential customers, business customers, or both residential and business customers transferred under the On-Line Transfer Service. In addition, the IC must provide a toll free number dedicated solely for On-Line Transfer Service in order for the Telephone Company to transfer the end user customer to the IC.

> A charge will be assessed to the participating IC for each customer call transferred to that IC as set forth in Section 13.4.3.C.

9. Reseller

> A reseller of the Company's Telephone Exchange Service (either lines or trunks), FGA and BSA-A lines, WATS Access lines and Plexar lines will be charged Easy Access Dialing nonrecurring charges as applicable.

Missouri Fublic

-11-11 : 221999

Issued:

JUN 2 2 1999

Effective:

JUL 2 2 1999

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247 (CT)

(CT)

# P.S.C. Mo.-No. 36

MO. PUBLIC SERVICE CONNER

Access Services Tariff Section 13 3rd Revised Sheet 9.01 Replacing 2nd Revised Sheet 9.01

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

RECEIVED

13.3 Miscellaneous Services-(Continued)

MAR 29 1993

- 13.3.3 Easy Access Dialing-(Continued)
  - D. Change Charges-(Continued)

MISSOURI Public Service Commission

- Easy Access Dialing change charges, as set forth in Paragraph 1. 13.3.3, F., will be applied as follows:-(Continued)
  - h. (Continued)

Centrex customers requesting an Easy Access Dialing change to only one line will incur the per line change charge. When a Centrex customer wishes to make an Easy Access Dialing change to multiple lines and does so with a single request, the per line change charge will apply to the first line, and the supplemental line change charge will apply to each additional line as set forth in Paragraph 13.3.3, F., following. Centrex customers who request an Easy Access Dialing change to their Centrex Group will incur the per Centrex Group change charge for each Centrex Group changed and a Centrex Group line change charge for each line in the Centrex Group that does not have an individual primary IC assigned to it.

i. The Telephone Company will make changes in the customer's or agent's primary IC assignment pursuant to an IC provided list of customers or agents accepted by the Telephone Company under a Limited Blanket Agency Agreement. Should customer or agent choice discrepancies occur, and the IC is unable to produce proper agency authorization, the IC rather than the customer or agent will be billed for any post-conversion Easy Access Dialing charge(s) that may apply for making the change and/or restoring By 4 RS # 9.01 the customer's or agent's original IC assignment.

Public Service Commission The Telephone Company will make changes in the customer's or agent's primary IC assignment resulting from an IC merger or separation into more than one entity or subsidiary. The applicable change charges will be determined on an Individual Case Basis (ICB) and billed to the IC rather than the customer or agent provided that the merger or separation does not involve discontinuance of the merged IC's FGD or BSA-D services. If FGD or BSA-D services are discontinued, then the rules and regulations specified in 13.3.3, C.6, preceding will apply.

Issued:

(AT)

(AT)

CANCELLED

JUL 2 2 1999

MISSOURI

MAR 2 6 1993 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

APR 1 1 1993

APR 1 1 1993 92-304 MO PHRIIC CEDVICE CONT

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 2nd Revised Sheet 9.01 Replacing 1st Revised Sheet 9.01

FEB 1 7 1989

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLATIOUS SERVICESEIVED (Continued)
  - 13.3 Miscellaneous Services-(Continued) BY 3MUK A THE BID MISSOURI Public Service Completic Service Commission
- (CP) 13.3.3 Easy Access Dialing-(Continued)
  - D. Change Charges-(Continued)
    - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
      - h.-(Continued)

Centrex customers requesting an Easy Access Dialing change to only one line will incur the per line change charge. When a Centrex customer wishes to make an Easy Access Dialing change to multiple lines and does so with a single request, the per line change charge will apply to the first line, and the supplemental line change charge will apply to each additional line as set forth in Paragraph 13.3.3., P., following. Centrex customers who request an Easy Access Dialing change to their Centrex Group vill incur the per Centrex Group change charge for each Centrex Group changed and a Centrex Group line change charge for each line in the Centrex Group that does not have an individual primary IC assigned to it.

APR 11 1993

BY 3MAR 5 9.01

MISSOURI

- i. The Telephone Company will make changes in the customer's or agent's primary IC assignment pursuant to an IC provided list of customers or agents accepted by the Telephone Company under a Limited Blanket Agency Agreement. Should customer or agent choice discrepancies occur, and the IC is unable to produce proper agency authorization, the IC rather than the customer or agent will be billed for any post-conversion Easy Access Dialing charge(s) that may apply for making the change and/or restoring the customer's or agent's original IC assignment.
- j. The Telephone Company vill make changes in the customer's or agent's primary IC assignment resulting from an IC merger or separation into more than one entity or subsidiary. The applicable change charges will be determined on an Individual Case Basis (ICB) and billed to the IC rather than the customer or agent provided that the merger or separation does not involve discontinuance of the merged IC's Feature Group D services. If Feature Group D services are discontinued, then the rules and regulations specified in 13.3.3 (C)(6) preceding will apply.

			_	EII ED
Issued:	FEB	27	1201	Bffective: MAR 90 1989
			Bur	P. D. BAPPON Provident Migsouri Division MAR 30 1989
			25	Southwestern Bell Telephone Company St. Louis, Missouri Public Service Commission
				St. Louis, Missouri Public Service Commission

Access Services Tariff Section 13 1st Revised Sheet 9.01 Replacing Original Sheet 9.01

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-DEC 29 1987 (Continued)

13.3 Miscellaneous Services-(Continued)

MISSOURI Public Service Commission

RECEIVED

- (CP) 13.3.3 Easy Access Dialing-(Continued)
  - D. Change Charges-(Continued)
    - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)

CANCELLER, Continued) CANCELLER, Contrex customers requesting an Easy Access Dialing change to only one line will incur the per line change charge. When a Contrex customer wishes to make an Easy Access Dialing change Contrex customer wishes to make an Easy Access Dialing change (ASMITIPLE lines and does so with a single request, the per BY CONTINUED BY CONTINUED DUDIE Service NISSOUR Supplemental line change charge will apply to each additional line as set forth in Paragraph 13.3.3.,F., following. Centrex customers who request an Easy Access Dialing change to their Centrex Group will incur the per Centrex Group change charge for each Centrex Group changed and a Centrex Group line change charge for each line in the Centrex Group that does not have an individual primary IC assigned to it.

> i. The Telephone Company will make changes in the customer's primary IC assignment pursuant to an IC provided list of customers accepted by the Telephone Company under a Limited Blanket Agency Agreement. Should customer choice discrepancies occur, and the IC is unable to produce proper agency authorization, the IC rather than the customer will be billed for any post-conversion Easy Access Dialing charge(s) that may apply for making the change and/or restoring the customer's original IC assignment.

j. The Telephone Company will make changes in the customer's primary IC assignment resulting from an IC merger. The applicable change charges will be determined on an Individual Case Basis (ICB) and billed to the IC rather than the customer provided that the merger does not involve discontinuance of the merged IC's Feature Group D services. If Feature Group D services are discontinued, then the rules and regulation specified in 13.3.3 (C)(6) preceding will apply.

FEB 8 1988

Issued: JAN 0.8 1988

Public Service Commission Effective: FEB () 8 1988

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 Original Sheet 9.01

(CP)		.3 Miscellaneous Services-(Continued) 13.3.3 Easy Access Dialing-(Continued)	MISSUURI Public Service Commission	and the second second second
7	13.	ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLAN (Continued)	eous services- JUN 27 1986	
1		ACCESS SERVICES	REGEIVED	

C. (Continued)

Centrex customers have the capability to change their primary IC for a single Centrex line, multiple Centrex lines or the Centrex Group. A Centrex Group defines a group of Centrex facilities (i.e., lines, trunks, attendant features, etc.) with a common dialing pattern. A customer may have multiple Centrex Groups.

Centrex customers requesting an Easy Access Dialing change to only one line will incur the per line change charge. When a Centrex customer wishes to make an Easy Access Dialing change to multiple lines and does so with a single request, the per line change charge will apply to the first line, and the supplemental line change charge will apply to each additional line as set forth in Paragraph 13.3.3, D., following. Centrex customers who request an Easy Access Dialing change to their Centrex Group will incur the per Centrex Group change charge for each Centrex Group changed and a Centrex Group line change charge for each line in the Centrex Group that does not have an indiED vidual primary IC assigned to it.

D. The nonrecurring charge for Easy Access Dialing is as follows 8 1988BY 124 R.S. # 9.01 Nonrecurring Chargervice Commission MISSOURI Easy Access Dialing, - per line change charge. . . . . . . . . . . . \$6.65 (1) - supplemental line change charge . . . . . 1.52 (1) - per Centrex Group change charge . . . . . 6.65(1)0.47(1)- Centrex Group line change charge. . . . . 5 [ 5 ] (1) This charge is billed to the End User which is the subscriber to the 1986 Telephone Exchange Service. If nonrecurring charges for Easy Access Dialing exist in the Telephone Company Intrastate Access Service gagifig 4

and the Interstate Access Service Tariff, the total charge to the rous Commission tomer will be calculated based on one half of the Intrastite charge and one half of the Interstate charge.

Issued: JUN 27 1986

1 1986 Effective: JUL

A

# P.S.C. Mo.-No. 36 Missouri Public Service Commission

# REC'D NOV 3 0 2000

Access Services Tariff Section 13 5th Revised Sheet 9.02 Replacing 4th Revised Sheet 9.02

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

#### 13.4 Rates and Charges-(Continued)

13.4.3 Easy Access Dialing-(Continued)

١.	Easy Access Dialing Rate Regulations-(Continued)	
		Nonrecurring Charge
	Residence/Business/Plexar Lines:	
	- per PIC change	\$6.65
	- per LPIC change	5,83
	- per supplemental PIC and/or LPIC change	1.52
	Plexar Groups:	
	- per group change	6.65
	- per group supplemental line change	0.47

 B. On-Line Transfer Service Charge
 Rate Per Call

 - per end user customer call transferred
 0.25

# Missouri Public Service Commission

# FILED UEC 30 2000



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

(RT)

(RT)

Issued: November 30, 2000

Effective: December 30, 2000

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

1

ţ

Access Services Tariff Section 13 4th Revised Sheet 9.02 Replacing 3rd Revised Sheet 9.02

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

	13.4 Rates and Charges-(Continued)	RECEIVED
	13.4.3 Easy Access Dialing-(Continued)	JUN 2 2 1999
(CT)	A. Easy Access Dialing Rate Regulations-(Continued)	MO. FUBLIC SERVICE COMM Nonrecurring Charge
	Residence/Business/Plexar Lines:	<u> </u>
	- per PIC change	\$6.65
	- per LPIC change	5.83
	<ul> <li>per supplemental PIC and/or LPIC change</li> </ul>	1.52
	Plexar Groups:	
	- per group change	6.65
	<ul> <li>per group supplemental line change</li> </ul>	0.47
	B. Easy Access Dialing Unauthorized PIC and/or LPIC	
	Residence/Business:	
	- per line PIC and/or LPIC change	17.00
	- per supplemental PIC and/or LPIC line change	6.00
ŏ	SmartCoin Access Lines:	
	<ul> <li>per line PIC and/or LPIC change</li> </ul>	63.00
	<ul> <li>per supplemental PIC and LPIC line change</li> </ul>	16.00
		Rate Per Call
	C. On-Line Transfer Service Charge	0.05
(CT)	<ul> <li>per end user customer call transfered</li> </ul>	0.25

# CANCELLED

DEC 3 0 2000 L, 5th RS 9.02 Fublic Service Commission MISSOURI

Missouri Fublic

11-11 2 2 19:30

Issued: JUN 2 2 1999

Effective:

JUI 2 2 1999

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 3rd Revised Sheet 9.02 Replacing 2nd Revised Sheet 9.02

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) **RECEIVED** 
  - 13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

D. Change Charges-(Continued)

MISSOURI Public Service Commission

OCT 04 1993

- 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
  - k. If nonrecurring charges for Easy Access Dialing exist in Telephone Company Intrastate Access Service Tariffs and the Interstate Access Service Tariff, the total charge to the customer or agent will be calculated based on one half of the Intrastate charge and one half of the Interstate charge. If nonrecurring charges for Easy Access Dialing exist only in the Telephone Company's Intrastate Access Service Tariff, 100 percent of the intrastate nonrecurring charge will apply.
- E. No Primary Carrier Selected
  - Customers may designate that they do not want a primary IC by notifying their Telephone Company business office. This choice is considered a valid selection and an Easy Access Dialing change charge will apply to any subsequent change. This no-primary IC designation is unavailable to agents of public and semipublic pay telephones.

# CANCELLED

(AT)

. The nonrecurring charge for Easy Access Dialing is as follows:

JUL 2 2 1999 By YRS # Q.02 Easy Access Dialing, Nonrecurring Charges(1) Public Service Commission \$6.65 1.52 6.65 0.47 These charges are billed to the customer who is the subscriber to the (1) Telephone Exchange Service, FGA or BSA-A lines, Centrex Service, or agent of public and semipublic pay telephones except when billed to set forth in Paragraph 13.3.3(D) preceding. Effective: NUV Issued: OCT 1 1 1993 Public Service Commission By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 2nd Revised Sheet 9.02 Replacing 1st Revised Sheet 9.02

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

RECEIVED

MAR 29 1993

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

D. Change Charges-(Continued)

MISSOURI Public Service Commission

- Easy Access Dialing change charges, as set forth in Paragraph 1. 13.3.3, F., will be applied as follows:-(Continued)
  - k. If nonrecurring charges for Easy Access Dialing exist in Telephone Company Intrastate Access Service Tariffs and the Interstate Access Service Tariff, the total charge to the customer or agent will be calculated based on one half of the Intrastate charge and one half of the Interstate charge.
- Ε. No Primary Carrier Selected
  - 1. Customers may designate that they do not want a primary IC by notifying their Telephone Company business office. This choice is considered a valid selection and an Easy Access Dialing change charge will apply to any subsequent change. This no-primary IC designation is unavailable to agents of public and semipublic pay telephones.

The nonrecurring charge for Easy Acres Dialing is as follows: F.

Easy Access Dialing	NOV 11 1995 NOV 11 1995 NOV 11 1995 NOV 11 1995 Norrecurring Charges(1) NOV 11 1995 Norrecurring Charges(1) Service Commission hargenissouri
- per line change charge.	V.3 Commission \$6.65
- per Centrex Group change	hargen 1950 6.65

(AT)

(1) These charges are billed to the customer who is the subscriber to the Telephone Exchange Service, FGA or BSA-A lines, Centrex Service, or to the agent of public and semipublic pay telephones except when billed to as set forth in Paragraph 13.3.3, D., preceding.

1 1 1993 304 MO. PUBLIC SERVICE COMM.

MAR 2 6 1993 Issued:

Ef	fe	ct	i١	/e:
----	----	----	----	-----

APR 1 1 1993

0.47

Access Services Tariff Section 13 1st Revised Sheet 9.02 Replacing Original Sheet 9.02

#### ACCESS SERVICES

13.	ADDITIONAL ENGINEERING,	ADDITIONAL LABOR	AND MISCELLANEOUS	SERVERSEIVED
	(Continued)			

13.3 Miscellaneous Services-(Continued)

(CP)

13.3.3 Easy Access Dialing-(Continued)

MISSOURI Public Service Commission

FEB 17 1989

- D. Change Charges-(Continued)
  - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
    - If nonrecurring charges for Basy Access Dialing exist in k. Telephone Company Intrastate Access Service Tariffs and the Interstate Access Service Tariff, the total charge to the customer or agent will be calculated based on one half of the Intrastate charge and one half of the Interstate charge.
- E. No Primary Carrier Selected
  - 1. Customers may designate that they do not want a primary IC by notifying their Telephone Company business office. This choice is considered a valid selection and an Basy Access Dialing change charge will apply to any subsequent change. This no-primary IC designation is unavailable to agents of public and semipublic pay telephones.

P. The nonrecurring charge for Easy Access Encling is as follows: CANCER II 1993 # 9.0 2 APR 11 1995 # 9.0 2 APR 10 APR 10

	av 2 commission	
-	per line change charge. BY & Commission supplemental line change philese MISSOURI	\$6.65
-	supplemental line change, childge and SOUT	1.52
-	per Centrex Group change charge	6.65
-	Centrex Group line change charge	0.47

(1) These charges are billed to the customer who is the subscriber to the Telephone Exchange Service, Feature Group A lines, Centrex Service, or to the agent of public and semipublic pay telephones except when billed to an IC as set forth in Paragraph 13.3.3(D) preceding. FILED

Issued: FEB 27 1889

Bffective: MAR 30 1989 Public Service Commission

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

(CP)

Access Services Tariff Section 13 Original Sheet 9.02

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICESIVED (Continued)

DEC 29 1987

MISSOURI

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued) Public Service Commission

- D. Change Charges-(Continued)
  - 1. Easy Access Dialing change charges, as set forth in Paragraph 13.3.3., F. will be applied as follows:-(Continued)
    - k. If nonrecurring charges for Easy Access Dialing exist in Telephone Company Intrastate Access Service Tariffs and the Interstate Access Service Tariff, the total charge to the customer will be calculated based on one half of the Intrastate charge and one half of the Interstate charge.
- E. No Primary Carrier Selected
  - Customers may designate that they do not want a primary IC by 1. notifying their Telephone Company business office. This choice is considered a valid selection and an Easy Access Dialing change charge will apply to any subsequent change.
- F. The nonrecurring charge for Easy Access Dialing is as follows:

## Nonrecurring Charge

Easy Access Dialing,

\$6.65 1.52 6.65 - Centrex Group line change charge. . . . . . . . 0.47



FILED

Issued: JAN 0 8 1988

Access Services Tariff Section 13 2nd Revised Sheet 9.03 Replacing 1st Revised Sheet 9.03

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

(RT)

Issued:

JUNE 22, 1999

Effective: JULY 22, 1999

FILED **MO PSC** 

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

(AT)

(AT)

Access Services Tariff Section 13 1st Revised Sheet 9.03 Replacing Original Sheet 9.03

#### ACCESS SERVICES

- ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOREGENEE 13. (Continued)
  - 13.3 Miscellaneous Services-(Continued)

OCT 17 1994

13.3.3 Easy Access Dialing-(Continued)

MO. PUBLIC SERVICE COMM.

G. CIC Code Reclamation

Nonrecurring Charges for changes in the customer's primary IC assignment will be waived when the charge is the result of the return of a Carrier Identification Code (CIC) TO Bellcore.

H. If a customer or agent choice discrepancy occurs, and the IC is unable to produce a Letter of Authorization (LOA), the customer's or agent's account will be credited for an amount equal to the nonrecurring Easy Access Dialing Change Charge. The IC will be billed the nonrecurring Easy Access Dialing Change Charges billed to the customer or agent and the nonrecurring Unauthorized PIC Change Charge(s) as set forth following for restoring the customer's or agent's original IC assignment. The Unauthorized PIC Change Charge applies to all lines, with the exception of coin which is separately identified.

Easy Access Dialing Unauthorized PIC Change Charge

Nonrecurring Charge

Residence/Business: - per line change charge - per supplemental line change charge	(NR) \$17.00 6.00
<pre>Public and Semi-Public Coin:     per line change charge     per supplemental line change     charge</pre>	63.00 (NR) 16.00

# CANCELLED

JUL 2 2 1999

By 2 RS #9.03 Public Service Commission MISSOURI



NOV 1 7 1994

MISSOURI Public Service Commission

Issued: OCT 1 7 1994

NOV 1 7 1994 By Horace Wilkins, Jr., President-Missouri Southwestern Bell Telephone St. Louis, Missouri

Effective:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(AT)

Access Services Tariff Section 13 Original Sheet 9.03

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) RECEIVED

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

G. CIC Code Reclamation

MISSOUR! Public Service Commission

APR 29 1991

Nonrecurring Charges for changes in the customer's primary IC assignment will be waived when the charge is the result of the return of a Carrier Identification Code (CIC) to Bellcore.

# CANCELLED

NOV 171994 1 at R.S. # 9.03 Public Service Commission MISSOURI

Issued: APR 3 0 1991

91 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

FILED

MAY 30 1991

Public Service CommissiC

Access Services Tariff Section 13 2nd Revised Sheet 9.04 Replacing 1st Revised Sheet 9.04

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

Issued:

(RT)

JUNE 22, 1999

Effective: JULY 22, 1999



By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

(MT)

(AT)

Access Services Tariff Section 13 1st Revised Sheet 9.04 Replacing Original Sheet 9.04

#### ACCESS SERVICES

RECEIVED

MO. PUBLIC SERVICE COMM.

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 8 1994
  - 13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

I. PIC Switchback is an option under which no investigation activities are performed by the Telephone Company when a business or residence end user customer denies requesting a PIC change previously submitted by an IC participating in PIC Switchback.

PIC Switchback is available to all ICs who participate in the Equal Access subscription process and have a Limited Blanket Agency Agreement for submitting PIC orders on file with the Telephone Company. In addition, ICs must provide written notification to the Telephone Company of their desire to participate in PIC Switchback.

When the Telephone Company is notified by a business or residence end user who denies requesting a PIC change within 90 days from when the PIC change was made and provided the IC participates in this option, the Telephone Company will restore the end user to their previous IC at no charge. Additionally, the end user will be credited the PIC change charge(s) assessed by the Telephone Company for the disputed PIC change.

The participating IC who initiated the disputed PIC change will be billed the PIC change charge(s) credited to the end user's account, as well as the PIC change charge(s) to change the end user back to their previous IC. A Letter of Authorization (LOA) will not be requested from the IC nor accepted at a later date under the PIC Switchback option. ICs participating in the PIC Switchback option will be **CANCELLED** ssessed the applicable Easy Access Dialing Change Charge(s).

JUL 2 2 1999 asy Access Dialing Unauthorized PIC Change regulations and rates will continue to apply to ICs choosing not to participate in PIC

By 2 RS #9.0 Switchback. Public Service Commission

(AT)

MISSOURI This option does not relieve the IC of the F.C.C.'s requirements for verifying all PIC orders obtained by telemarketing prior to submitting orders to the Telephone Company and for instituting steps to obtain LOAs on all PIC orders submitted to the Telephone Company.

Issued: DFC 0 8 1994

Effective: JAN U 8 1995

JAN 0 8 1995

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

MISSOURI Public Service Commission

(AT)

(AT)

Access Services Tariff Section 13 Original Sheet 9.04

#### ACCESS SERVICES

RECEIVED

MO. PUBLIC SERVICE COMM.

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) NOV 2 1994
  - 13.3 Miscellaneous Services- (Continued)

13.3.3 Easy Access Dialing-(Continued)

I. On-Line Transfer Service is an option under which the Telephone Company will provide for the transfer of a residential or business end user customer to the Primary IC selected by the customer, when that IC participates in the On-Line Transfer Service.

On-Line Transfer Service is available to all ICs who participate in the Equal Access subscription process. The IC has the option to have either residential end user customers, business end user customers, or both residential and end user customers transferred under the On-Line Transfer Service. In addition, the IC must provide a toll free number dedicated solely for On-Line Transfer Service in order for the Telephone Company to transfer the end user customer to the IC.

The following charge will be assessed to the participating IC for each end-user customer call transferred to that IC.

Rate Per Call

\$.25

On-Line Transfer Service Charge

- per end user customer call transferred

CANCELLED

LAN 81995 BY 124 R.S = 9.04Public Service Commission MISSOURI

FILED

DEC 2 3 1994

MISSOURI Public Service Commission

Issued: NOV 2 3 1994

Effective: DEC 2 3 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 9.05 **Replacing Original Sheet 9.05** 

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

 $(\mathbf{R}|\mathbf{T})$ 

Issued:

JUNE 22, 1999

Effective: JULY 22, 1999

**MO PSC** 

By JAN NEWTON, President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 9.05



ACCESS SERVICES

DEC 8 1994

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - MO. PUBLIC SERVICE COMM.

13.3 Miscellaneous Services-(Continued)

13.3.3 Easy Access Dialing-(Continued)

(MT) (FC)

(MT)

J. On-Line Transfer Service is an option under which the Telephone Company will provide for the transfer of a residential or business end user customer to the Primary IC selected by the customer, when that IC participates in the On-Line Transfer Service.

On-Line Transfer Service is available to all ICs who participate in the Equal Access subscription process. The IC has the option to have either residential end user customers, business end user customers, or both residential and end user customers transferred under the On-Line Transfer Service. In addition, the IC must provide a toll free number dedicated solely for On-Line Transfer Service in order for the Telephone Company to transfer the end user customer to the IC.

The following charge will be assessed to the participating IC for each end-user customer call transferred to that IC.

Rate Per Call

On-Line Transfer Service Charge

- per-end user customer <u>call</u>-transferred

\$.25

# CANCELLED

JUL 2 2 1999 By 1 RS # 9.05 Public Service Commission MISSOURI

HILED

JAN 0 8 1995

Issued: DEC 0 8 1994

JAN 0 8 1995 MISSOURI Public Service Commission By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

Access Services Tariff Section 13 2nd Revised Sheet 10 Replacing 1st Revised Sheet 10

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.4 Standard Jacks - Registration Program

Registered jacks are provided by the Telephone Company to connect Registered Equipment to those services that are subject to the Registration Program as set forth in Technical Reference Publication AS No. 1. Registered jacks are used to terminate services provided by the Telephone Company. For each service provided by the Telephone Company, a specific registered jack has been selected as the standard for the service. This jack will be provided and installed at no additional charge. Other services or facilities, provided either by the Telephone Company or by others, may also be terminated in any spare capacity of the standard jack. There is no charge for the use of spare capacity of the jack.

The customer may procure an additional jack (or jacks) from either the Telephone Company or another vendor for installation on the customer's side of the demarcation point.

The use of Jacks is covered in Part 68 of the F.C.C.'s Rules and Regulations. Specific jacks are described in the document on file with the F.C.C. entitled "Descriptions of Standard Registration Program Connection Configurations Supplementing Configurations described in Subpart F of Part 68 of the F.C.C.'s Rules and Regulations".

Issued:	October 11, 1993	Effective:	November 11, 1993			
	By M. H. SCHULTEIS, D	ivision Manager-Reg	ulatory & Industry Relations			
	Southwestern Bell Telephone Company					

St. Louis, Missouri



No supplement to this Access Services Tariff tariff will be issued Section 13 except for the purpose 1st Revised Sheet 10 of canceling this tariff. Replacing Original Sheet 10 ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) JUN 27 1985 13.3 Miscellaneous Services-(Continued) MISSUURI Public Service Commission 13.3.4 Standard Jacks - Registration Program Standard jacks are provided by the Telephone Company to connect Registered Equipment to those services as set forth in Section 2. Paragraph 2.5.1, preceding. The use of jacks is covered in the FCC's Rules and Regulations. Specific jacks are described in the document on file with the FCC entitled "Descriptions of Standard Registration Program Connection Configurations Supplementing Configuration Described in the FCC's Rules and Regulations." These jacks are used to terminate services provided by the Telephone Company. Other services or facilities provided by the Telephone Company or by others may also be terminated in any spare capacity of the jacks remaining after installation without additional charge for the use of such capacity. The nonrecurring charges, which include installation, for standard jacks and their typical uses are set forth following: Nonrecurring USOC Charges CANCELLED Standard Voice Jacks Α. 1. Miniature Six-position jacks for connection of terminal iommission equipment as follows: BY Public Service Single-line telephone MISSO set, surface or flush (CR) RJ11C \$6.77 mounted . . . . . b. Single-line telephone (CR) sets, wall mounted. . . . RJIIW-6-23 [SII] 1986 .用1 1 vice Commission Issued: JUN 27 1900 JUL 1 1986 Effective: By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 10

DEC 2

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.4 Standard Jacks Registration Program

Standard jacks are provided by the Telephone Company to connect Registered Equipment to those services that are subject to the Registration Program as set forth in Section 2, Paragraph 2.5, preceding. The use of jacks is covered in the FCC's Rules and Regulations. Specific jacks are described in the document on file with the FCC entitled "Descriptions of Standard Registration Program Connection Configurations Supplementing Configuration Described in the FCC's Rules and Regulations."

These jacks are used to terminate services provided by the Telephone Company. Other Services or facilities provided by the Telephone Company or by others may also be terminated in any spare capacity of the jacks remaining after installation without additional charge for the use of such capacity.

The rates and charges for standard jacks and their typical uses are set forth following:

	-					USOC	No 	nrecurring Charges	
	A.	Stand	lard Void	e Jacks					
•		t	for conne	e Six-positio ection of ten as follows:	minal				
,			set, mount	le-line telep surface or f ted le-line telep	iush RANR		D	\$12.42	
		i		, wall mounte		rjiiw 1986		12.42	
	,				BY AR . PUBLIC SERVICE	S. 10 E COMMISSIC SSOUNI			
<b>-</b>		<u></u>	,				<u>.</u>		
I	ssued:	DEC	2 9 1983		Effecti	ve: J	AN O		
			By	Southwester	N, Vice Presi n Bell Teleph Louis, Misso	one Compa		<b>83 - 25</b> Public Service Com	- ,

Access Services Tariff Section 13 3rd Revised Sheet 11 Replacing 2nd Revised Sheet 11

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)



October 11, 1993

Effective: November 11, 1993

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri



-

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 13 2nd Revised Sheet 11 Replacing 1st Revised Sheet 11

- -- --

### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS (Continued)	SERVICES-
13.3 Miscellaneous Services-(Continued)	JUN 17 1988
13.3.4 Standard Jacks - Registration Program-(Continued)	MISSOURI Service Commission
USOC	Nonrecurring Charges
A. Standard Voice Jacks-(Continued)	
1. (Continued)	
c. Two-line non-key telephone sets, surface or flush mounted RJ14C	\$ 5.21
d. Single-line bridged 4-wire exchange 2/RT, T1/R1 RJ1DC	6.77
e. Two-line non-key telephone sets, wall mounted RJ14W f. Special single-line equin-	<b>ED</b> 6.23
ment for use in hospital critical care areas RJ17C 11	1993 6.77
(DR) (FC) g. Three-line non-key tele- phone sets and ancillary devices	23.00
<pre>(NR) h. Single-line non-key telephone ancillary devices connected directly to central office lines where there is a requirement for make busy:</pre>	
- Portable Wall Mounted Equipment RJ18W (NR) - All Other RJ18C	6.26 FILEØ <sup>80</sup>
	JUL 18 1988

Issued: JUN 17 1988

Effective: JUL 18 1980 lic Service Commission

2342

	No supplemen tariff will except for t of canceling	be isa he pur	wed pose tariff.	ACCESS SERVICES	Repla	Access Services Sect lst Revised Sh acing Original Sh	ion 13 eet 11
	13. ADDITIC (Contin	NAL EN ued)		ITIONAL LABOR AN	ID MISCELLA	NEOUS BERVICES	VED
	13.3 Misc	ellane	ous Services-(	Continued)		JUN 27	1986
_	13.3.4	Standa	rd Jacks - Reg	istration Progra		Public Service (	(
	A. St	anda ra	Voice Jacks-(	Continued)	USOC	Charges	
	1.		tinued)	concruded)			
	1.	•	,	··· ··!··			
(CR)		с.	Two-line non-k sets, surface mounted	or flush	RJ14C	\$ 5.21	
('CR).		đ.	Single-line br exchange 2/RT,		RJ 1DC	6.77	
(CR)		е.	Two-line non-k sets, wall mou		RJ14W	6.23	
(CR)		f.	Special single ment for use is critical care	n hospital	RJ17C	6.77	
(CR)		g.	9DB single-lin ment with mode and mode indic leads. This j mally used in with a series	indication ation common ack is nor- association	RJ16X	4.68	-
(CR)		h.	Three-line non hone sets and devices	ancillary	RJ25C	<u>23.00</u> [京讯] []	
			CANCEL	LED			986
•		Pu	JUL 18 19 BY <u>2nd R</u> blic Service Co MISSOUF	<u>S.≠//</u> m <b>misai</b> on		<b>B 6 - 8</b> Fublic Service Co	4 mmission
	Issued: JUN	27 1	336	Effectiv	re: JUL	1 1986	
			Southweste	N, President-Mis rn Bell Telephor . Louis, Missour	ae Company	ision	

10 ST 10 ST 10 ST

5.27.2

20 St 10 2 12

P.S.C. MoNo. 3	٢.	ъ.	<b>U</b> •	MO	•-NO		20
----------------	----	----	------------	----	------	--	----

No supplement to this Access Services Tariff tariff will be issued Section 13 except for the purpose Original Sheet 11 of canceling this tariff. ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 2 7 (233 13.3 Miscellaneous Services-(Continued) 13.3.4 Standard Jacks - Registration Program-(Continued) Nonrecurring USOC Charges A. Standard Voice Jacks-(Continued) 1. (Continued) Two-line non-key telephone с. sets, surface or flush mounted . . . . . . . . . . RJ14C \$12.42 d. Single-line bridged 4-wire exchange 2/RT, T1/R1. . . RJIDC 12.42 e. Two-line non-key telephone sets, wall mounted. . . . RJ14W 12.42 f. Special single-line equipment for use in hospital critical care areas . . . . RJ17C 12.42 9DB single-line data equipg. ment with mode indication and mode indication common leads. This jack is normally used in association with a series jack. . . . 12.42 RJ16X h. Three-line non-key tele-60.97 JUL 1 1986 BY PUBLIC SERVICE COMMISSION OF MISSOURI DEC 2 9 1983 JAN 0 1 1984 J Issued: Effective: 83

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company | Public Service Commission

Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 2nd Revised Sheet 12 Replacing 1st Revised Sheet 12

# ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)



Effective: November 11, 1993

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri



	<b>UU</b> N	<u> </u>				· • •
	Issued:	2719	86	Effective: JUL	1	1986
					<b>t</b> ⊪	n an far i far far san a <del>an an a</del>
,					Ţ	86-84 Public Service Commission
						H.H. 1 1986
					rife demokratie	
CR)				premises station lines (25- line capacity) RJ	J21X	49.04
			e.	For connection to off-		
CR)			d.	For connection to 4-wire tie trunks E&M Type II signaling (6-line capacity) RJ	12HX	49.04
CR)				tie trunks E&M Type II signaling (6-line capacity) RJ	J2FX	49.04
			c.	For connection to 2-wire		
CR)				tie trunks E&M Type I signaling (8-line capacity) RJ	J2GX	49.04
CR)			Ъ		J2EX	\$ 49.04
			a.	For connection to 2-wire tie trunks E&M Type I signaling		BY Service Commission Public Service Commission MISSOURI
			nat	ing equipment and channel ivation devices as follows:		NOV 11 1993 BY 2 MARS 13
		2.		Position Miniature Ribbon for nection of multiline termi-		CANCELLED
	A.	. Sta	ndar	d Voice Jacks-(Continued)		Charges
				USOC	- 1	Public Service Commission
				Services-(Continued) acks - Registration Program-(Contin	nued	MISSOURI
	(Continu	ied)				JUN 2 7 1980
)	13. ADDITION	IAL EN	GINE	ACCESS SERVICES ERING, ADDITIONAL LABOR AND MISCELL	ANEO	REGEIVED
	of canceling			ff. Repl		st Revised Sheet 12 g Original Sheet 12
	except for th					ST KAVISAA SNAAT 17

A.

Access Services Tariff Section 13 Original Sheet 12

5. O. M. Salo

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 20 NOU

13.3 Miscellaneous Services-(Continued)

13.3.4

		ard Jacks - Registration Progra	um-(Continued	Den service Provinsional de la companya de la co Provinsional de la companya de la com	in en
			USOC	Nonrecurring Charges	(a) Control of the second s
Star	ndar	d Voice Jacks-(Continued)			-
2.	con nat	Position Miniature Ribbon for nection of multiline termi- ing equipment and channel ivation devices as follows:			
	a.	For connection to 2-wire tie trunks E&M Type I signaling (12-line capacity)	RJ2EX	\$199.85	
	Ъ.	For connection to 4-wire tie trunks E&M Type I signaling (8-line capacity).	RJ2GX	199.85	
	c.	For connection to 2-wire tie trunks E&M Type II signaling (8-line capacity).	. RJ2FX	199.85	
	d.	For connection to 4-wire tie trunks E&M Type II signaling (6-line capacity).	. RJ2HX	199.85	
	e.	For connection to off- premises station lines (25- line capacity)		199.85	
		JUL 1 RY J R PUBLIC SERVICE	COMMISSION		2:12
DEC	29	1983 Effect:	Lve: JAN O	3 <u>43</u> -1 1 1984 <b>8 3 -</b>	<del>1633</del> 258

Ry R. D. BARRON, Vice President-Missourd Public Service

Southwestern Bell Telephone Company St. Louis, Missource

Issued: DEC 2 9 1983

Access Services Tariff Section 13 3rd Revised Sheet 13 Replacing 2nd Revised Sheet 13

# ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)



Effective: November 11, 1993

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri



ti e:	ariff will xcept for	nt to this be issued the purpos og this tar	i :e			ccess Services Tariff Section 13 2nd Revised Sheet 13 1st Revised Sheet 13
			ACC	CESS SERVICES		
	3. ADDITI (Conti		EERING, ADDIT	ONAL LABOR AND	MISCELLAN	EOUS SERVICES-
	13.3 Mis	scellaneous	s Services-(Con	itinued)		RECEIVED
	13.3.4	Standard	Jacks - Regist	tration Program-	(Continue	I) JUN 1 7 1988
					USOC	MISSOURI Public Service Commission Charges
	A. St	tandard Vo	ice Jacks-(Con	tinued)		CANCELLED
(DR)	2 .	. (Contin	ued)			NOV 11 1993 # 13
(FC)		lin	connection of e bridged 4-wi F, T1/R1	re exchange	RJ2DX	NOV TTE S# 13 Y 3 M R.S. # 13 IIC Service Commission MISSOURI MISSOURI MISSOURI
(DR)						
(AT)		Non Anc dir Lin	connection of -Key Telephone illary Devices ectly to Centra es where there uirement for Ma	and connected al Office	. RJ2MB	49.22
(FC)	3	single- at loca	proof jack for line telephone tions such as	sets used	. RJ15C	41.84
(NR)		Non-Reg Private Communi Referen		Band Analog for Bell	. JM8	FILED
•						JUL 18 1988
• ·					P	ublic Service Commission
	Issued: J	UN 17 198	38	Effective:	JUL 18	1988
•			outhwestern Be	esident-Missour 11 Telephone Con is, Missouri		: :

20

. Seiter

	No suppl tariff w except f of cance	√ll↓ For t	be is he pu	ssued		Repla	Access Services Tariff Section 13 lst Revised Sheet 13 acing_Original_Sheet_13
					ACCESS SERVICES		REGEIVED
		OITIO ntin		NGINEERING, ADD	ITIONAL LABOR ANI	D MISCELL	NEOUS SERVICES- JUN 27 1986
				aeous Services-(	•		MISSUURI Public Service Commission
	1.2.4.4		o cuin	ard backs - heg	ISTICTION TOGTA		Nonrecurring Charges
	Α.	. Sta	ndaro	l Voice Jacks-(C	ontinued)		
		2.	(Cor	itinued)			
( <u>C</u> R)			f.	For use with se such as toll re (12-line capaci	stirctors	RJ71C	\$ 49.04
(R)			g.	For connection line bridged 4- 2/RT, T1/R1	wire exchange	RJ 2DX	49.04
•		3.		ies Jacks for co ninal equipment	as follows:		CANCELLED
(CR)			a.	Single-line ala devices		RJ31X	JUL 11 \$61988
			Ъ.	Series ancillar such as automat		Pt	BY <u>Zud</u> R.S.#13 Iblic Service Commission MISSOURI
(CR)				Single-line set exclusion		RJ32X	11.56
(CR)			c.	Two-line teleph with exclusion		RJ37X	11.56
(CR)		4.	sin at	therproof jack f gle-line telepho locations such a inas	ne sets used s boats and	RJ15C	4F.84 ED
¢							HH_ 1 1986 <b>86 - 84</b> Public Service Commissio
	Issued:	JUI	127		Effective N, President-Mis		
				Southweste	rn Bell Telephon Louis, Missour	e Company	

SATURA STATES

20.28

1

÷

1

**.** -

Access Services Tariff Section 13 Original Sheet 13

S 60

## ACCESS SERVICES

13	(Con .3 M			eous Services-(Cont	inued)		DEC 2 )	
	13.3.	4 S	Itand	ard Jacks - Registr	ation Program	n-(Continu		Commiss
						USOC	Nonrecurring Charges	
	A.	Sta	ndar	d Voice Jacks-(Cont	inued)	-	-	
		2.	(Co	ntinued)				
			f.	For use with serie such as toll restr (12-line capacity)	ictors	RJ71C	\$130.98	
			g.	For connection of line bridged 4-wir 2/RT, T1/R1	e exchange	RJ2DX	125.33	
		3.		ies Jacks for conne minal equipment as				
			a.	Single-line alarm devices		RJ31X	82.42	
			Ъ.	Series ancillary d such as automatic				
				Single-line sets we exclusion		RJ32X	82.42	
			c.	Two-line telephone with exclusion on		RJ37X	82.42	
		4.	sin at	therproof jack for gle-line telephone locations such as b inas	sets used	ĻĻĘD	150.17	
					JUL 1	1986		
					BY A R			· · · · ·
Issue	ed:	DEC	29	1983	Fffecti	ve: JAI	JAN 1 N O 1 1984 <b>8 3 - 2 5</b> Public Service Cr	

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company
Access Services Tariff Section 13 3rd Revised Sheet 14 Replacing 2nd Revised Sheet 14

# ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)



Effective: November 11, 1993

By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri



P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

----

. ---

Access Services Tariff Section 13 2nd Revised Sheet 14 Replacing 1st Revised Sheet 14

### ACCESS SERVICES

	17		tinu	,	ous Services-(Continued)			RECEIVED
	¥.				rd Jacks - Registration Pro	gram-(Con	tinued)	JUN 17 1988
		,		tanu	ru sacks – negistration rio			MISSOURI ic Sawiffe Commissio Charges
		В.	Sta	ndaro	Data Jacks			
·			1.	coni and	versal Data Jack for use in mecting fixed loss loop (FLL programmed (P) types of data pment (one-line capacity).	a	RJ41S	\$ 21.37
			2.	in	rammed Data Jack for use connecting programmed data pment (one-line capacity).	• •	RJ45S	17.89
DR)								
NR)			3.	Jac los (P) Thi to of	tiple Line Universal Data t for use in connecting fixe t loop (FLL) and programmed types of data equipment. types of data equipment. types of data equipment. types of data equipment types of the use the following equipment:	d		NCELLED N 11 1993 2 Nd R. St. 14 2 Nd R. St. 14
NR)				a.	Multiple Universal Data Jac Mounting Capacity 16 Circui (103A Apparatus Mounting RJ First Universal Data Jack i included	ts M2X). s	BY Public	Service Commission MISSOURI 135.60
AT)				b.	Universal Data Jack. One i required for each additiona circuit	-	RJ41S	FILED
								JUL 18 1988
							Publ	ic Service Commissio
	Jee	ued: 、	אחר	17	988 Effect	ive JU	18 1988	

Southwestern Bell Telephone Company

an saidhean a-

St. Louis, Missouri

P.S.C. Mo.-No. 36

tariff wi except fo	nent to this Li be issued t the purpose	Access Services Tariff Section 13 1st Revised Sheet 14
or cancer	ing this tariff. ACCESS SERVIO	Replacing Original Sheet 14
13. ADDI (Con	FIONAL ENGINEERING, ADDITIONAL LABO	R AND MISCELLANEOUS SERVICES 7 1985
	iscellaneous Services-(Continued)	MISSUURI Public Service Commiss
13.3.	4 Standard Jacks - Registration Pro	Nonrecurring USOCCharges
	B. Standard Data Jacks	
(CR)	<ol> <li>Universal Data Jack for us connecting fixed loss loop and programmed (P) types of equipment (one-line capac:</li> </ol>	p (FLL) of data
(CR)	<ol> <li>Programmed Data Jack for a in connecting programmed equipment (one-line capac:</li> </ol>	iata
(CR)	3. Miltiple Line Universal Da Jack for use in connecting loss loop (FLL) and progra (P) types of data equipment This jack will terminate to eight lines. The select of this jack requires the of the equipment listed for lowing:	g fixed ammed ht. up ction use ol-
) (CR)	a. Multiple Line Universa Data Jack Circuit Card	al is. One
	CANCELLED	FRLED
)	JUL 18 1988 BY <u>Znd R.S.#</u> 14 Public Service Commission MISSOURI	Public Service Commissio
Issued: J	UN 27 1986 Effe	ctive: JUL 1 1935

Access Services Tariff Section 13 Original Sheet 14

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 29 1003

13.3 Miscellaneous Services-(Continued)

13.3.4 Standard Jacks - Registration Program-(Continued) Public Schlassical Continued

				USOC	Nonrecurring Charges
В.	Stan	dard Data Jacks	-		
		and programmed (	Mack for use in 1 loss loop (FLL) (P) types of data Line capacity)	. RJ41S	\$_85.81
		Programmed Data in connecting pr equipment (one-)		. RJ45S	85.81
		loss loop (FLL) (P) types of dat This tack will to to eight lines. of this tack red of the equipment	connecting fixed and programmed ta equipment. terminate up The selection quires the use	. RJ26X	312.76
		For use with circuit card	ne Universal ircuit Cards. n RJ26X. One i per circuit	. RJ26S	99.36
			GANGE		
			JUL 1 BY A PUBLIC SEPN CE	S:14	1
			Othe and he		
Issued:	DEC 2	9 1983	Effect	0/11	83-200
		By R. D. South	BARRON, Vice Pres western Beli Telep	ident-Misso hone Compan	uri Public Service Conni

St. Louis, Missouri

Access Services Tariff Section 13 3rd Revised Sheet 15 Replacing 2nd Revised Sheet 15

# ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)



By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri



-

1

•

....

and we have a set of the

Sec. Sec. 1

No supplement to this tariff will be issued except for the purpo of canceling this ta	l se	Access Services Tariff Section 13 2nd Revised Sheet 15 cing 1st Revised Sheet 15
<b>U</b>	ACCESS SERVICES	C C C C C C C C C C C C C C C C C C C
13. ADDITIONAL ENGL (Continued)	NEERING, ADDITIONAL LABOR AND MISCE	LLANEOUS SERVICES- RECEIVED
13.3 Miscellaneou	s Services-(Continued)	JUN 17 1988
13.3.4 Standard	Jacks - Registration Program-(Cont	
B. Standard D	ata Jacks-(Continued)	
3. (Conti	nued)	CANCELLED
(DR) (NR) c. St	andard Digital Data Jacks	NOV 11 1993 BY 3 M R.S. #15 Public Service Commission MISSOURI
1.	Single Jack 1.544 mbps Digital Services Miniature 8-Position Non-Keyed Modular Jack	BY Service Commission Public Service Commission MISSOURI RJ48C \$ 57.20
2.	Sub-rate Digital (2.4, 4.8, 9.6, and 56 kbps) and Local Area Data Channel Jacks	
	a. Single Jack for Sub-Rate Digital and Local Area Data Channels. Miniature 8-Position Keyed Modular Jack	RJ48S 14.20
	<ul> <li>b. Multiple Jack for Sub-Rate Digital and Local Area Data Channels. 12 Circuit Maximum. 50 Position</li> </ul>	RJ48T 49.22
(NR)	Miniature Ribbon Jack	
_		FILED
		JUL 18 1988
		Public Service Commission
Issued: JUN 17 19	38 Effective: JUL	18 1988
By R	. D. BARRON, President-Missouri Divi Southwestern Bell Telephone Company St. Louis, Missouri	

P.S.C. Mo.-No. 36

ssued urpose	lst R	Services Tariff Section 13 Levised Sheet 15 Liginal Sheet 15
ENGINEERING, ADDITIONAL LAE neous Services-(Continued)	LI BOR AND MISCELLANEOUS S Program-(Continued)Pub)	SECENVED JUN 2 7 1986 MISSUURI ic Service Commission
	USOCCh	arges
tandard Data Jacks-(Continu	ued)	
Data Jack Mounting o	options.	
- Wall Mounting with cover	а клмзх	\$25.62
- Rack Mounting (19 inch or 23 inch).	RJM4X	17.02
JUL 18 1988	Рирг	FRILED HH. 1 1986 <b>86 - 84</b> Ic Service Commission
	ENGINEERING, ADDITIONAL LAE neous Services-(Continued) dard Jacks - Registration F standard Data Jacks-(Continue (Continued) b. Multiple Line Univer Data Jack Mounting of For use with RJ26X. required per RJ26X. - Wall Mounting with cover - Rack Mounting (19 inch or 23 inch). MUL 18 1988 Man L.S. #/S Service Commission	ssued urpose Ist F Replacing Or ACCESS SERVICES ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS S neous Services-(Continued) dard Jacks - Registration Program-(Continued) dard Jacks - Registration Program-(Continued) Nonre- USOC CH tandard Data Jacks-(Continued) b. Multiple Line Universal Data Jack Mounting options. For use with RJ26X. One required per RJ26X. - Wall Mounting with cover RJM3X - Rack Mounting (19 inch or 23 inch) RJM4X ANCELLED JUL 18 1988 2 Mall S. #/S

Issued: JUN 27 1985

I

4

Effective: JUL 1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 15

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

DEC 2 9 1033

13.3.4 Standard Jacks - Registration Program-(Continued)

USOC

Nonrecurring Charges

B. Standard Data Jacks-(Continued)

3. (Continued)

 b. Multiple Line Universal Data Jack Mounting options.
 For use with RJ26X. One required per RJ26X.

> Wall Mounting with cover . . . . . . . . . RJM3X \$56.46
>  Rack Mounting (19 inch or 23 inch). . . . . RJM4X 35.00

GANGELLED JUL 1 1986 PUBLIC REPAIRE COMMISSION OF WEAKER

Pahle

်ခ

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 16 Replacing Original Sheet 16

## ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services(Continued)

## 13.3.5 Testing Services

Testing Services offered under this section of the tariff are optional and subject to rates and charges as set forth in 13.3.5, C., following. Other testing services provided by the Telephone Company in association with Access Services are furnished at no additional charge. These other testing services are described in Paragraphs 6.1.4 and 7.1.7, preceding.

Testing services are normally provided by Telephone Company personnel at Telephone Company locations. However, provisions are made in 13.3.5, A.5 and B.2, following for a customer to request Telephone Company personnel to perform testing services at the IC terminal location or End User's premises.

The offering of Testing Services under this section of the Tariff is made subject to the availability of the necessary qualified personnel and test equipment at the various test locations mentioned in Paragraphs 13.3.5, A., B. and C., following:

A. Switched Access Service

Testing Services for Switched Access are comprised of (a) tests which performed during the installation of a Switched Access Service and (b) tests which are performed after acceptance of such access services by an IC, i.e., inservice tests. These inservice tests may be further divided into two broad categories of tests: scheduled and nonscheduled.

Scheduled tests are those tests performed by the Telephone Company on a regular basis, e.g., monthly, which result in the measurement of Switched Access Service. Scheduled tests may be done on an automatic basis (no

(CT)

Issued: June 27, 1986

Effective: July 1, 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff Section 13 Original Sheet 16

#### ACCESS SERVICES

ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-13. (Continued)

13.3 Miscellaneous Services-(Continued)

DEC 2 0 trag

13.3.5 Testing Services

The Telephone Company will, in addition to any IC-requested acceptance testing, perform such tests as it believes necessary to insure that the access services ordered by an IC are functioning properly prior to turning over such access services to the IC. In addition, the Telephone Company as part of the ongoing work to maintain the continued satisfactory performance of the access services ordered by the IC, may perform periodic tests.

Testing Services offered under this Section of the Tariff are in addition to the tests described above and will be provided when requested by the IC, except for scheduled testing of Feature Groups C and D, which is nonoptional. Testing services are normally provided by Telephone Company personnel at Telephone Company locations. However, provisions are made in Paragraphs 13.3.5, A.5. and B.2., following, for an IC to request Telephone Company personnel to perform testing services at the IC terminal location or End User's premises. In addition, the Telephone Company will, at the request of the IC, perform Acceptence Testing with the IC in accordance with the provisions set forth in Sections 6 and 7, preceding.

The offering of Testing Services under this Section of the Tariff is made subject to the availability of the necessary qualified personnel and test equipment at the various test locations mentioned in Paragraphs 13.3.5, A., B. and C., following:

A. Switched Access Service

DEC 2 9 1983

Testing Services for Switched Access are comprised of (a) tests which are performed during the installation of a Switched Access Service and (b) tests which are performed after acceptance of such access services by an IC, i.e., in-service tests. These in-service tests may be further divided into two broad categories is tests incheduled and nonscheduled.

Scheduled tests are those tests performed by the Telephone Company on a regular basis, e.g., monthly, which result in the measurement of Switched Access Service. Scheduled tests may be done on an automatic basis (no  $x = \frac{1}{2} + \frac{1}{2} +$ 1 M4 (K

PUBLIC SERVICE COMMISSION OF MISSCRUB

Effective:

8Y

Issued:

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Public Series Commission St. Louis, Missouri

JAN 0 1 1984

Access Services Tariff Section 13 1st Revised Sheet 17 Replacing Original Sheet 17

## ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - A. Switched Access Service-(Continued)

Telephone Company or IC technicians involved), on a cooperative basis [Telephone Company technician(s) involved at Telephone Company office(s) and IC technician(s) involved at IC terminal location(s)], or a manual basis [Telephone Company technician(s) involved at Telephone Company office(s) and at IC terminal location(s)].

Nonscheduled tests are performed by the Telephone Company "on demand," which result in the measurement of Switched Access Services. Non-scheduled tests may involve Telephone Company technicians at the Telephone Company offices and at the IC's terminal location.

1. Additional Cooperative Acceptance Testing

Additional Cooperative Acceptance Testing (ACAT) of Switched Access Service involves the Telephone Company provision of a technician at its office(s), and the IC provides a technician at its terminal location(s) with suitable test equipment to perform the required tests.

Additional Cooperative Acceptance Tests may, for an example, consist of the following tests:

- Impulse Noise
- Phase Jitter
- Signal to C-Notched Noise Ratio
- Intermodulation (Nonlinear) Distortion
- Frequency Shift (Offset)
- Envelope Delay Distortion
- Dial Pulse Percent Break

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff Section 13 Original Sheet 17

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.5 Testing Services-(Continued)

DEC 2 9 (203)

A. Switched Access Service-(Continued)

Telephone Company or IC technicians involved), on a cooperative basis [Telephone Company technician(s) involved at Telephone Company office(s) and IC technician(s) involved at IC terminal location(s)], or a manual basis [Telephone Company technician(s) involved at Telephone Company office(s) and at IC terminal location(s)].

Nonscheduled tests are performed by the Telephone Company "on demand," which result in the measurement of Switched Access Services. Non-scheduled tests may involve Telephone Company technicians at the Telephone Company offices and at the IC's terminal location.

1. Additional Cooperative Acceptance Testing

Additional Cooperative Acceptance Testing (ACAT) of Switched Access Service involves the Telephone Company provision of a technician at its office(s), and the IC provides a technician at its terminal location(s) with suitable test equipment to perform the required tests.

Additional Cooperative Acceptance Tests may, for an example, consist of the following tests:

JIII 1

Effective:

1986

JAN 0 1 198

- C-Notched Noise
- Impulse Noise
- Phase Jitter
- Signal to C-Notched Noise Ratio
- Intermodulation (Nonlinear) Distortion
- Frequency Shift (Offset)
- Envelope Delay Distortion
- Dial Pulse Percent Break GANGELLED

Issued: DEC 2 9 1983

١

By R. D. BARRON, Vice President-Missouri 83-253 Southwestern Bell Telephone CompanyPublic Service Commission St. Louis, Missouri

فكرعد بمحجز كنعب الملا

Access Services Tariff Section 13 2nd Revised Sheet 18 Replacing 1st Revised Sheet 18

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - A. Switched Access Service-(Continued)
      - 2. Automatic Scheduled Testing

Automatic Scheduled Testing (AST) of Switched Access Services (FGB, FGC, FGD, BSA-B, BSA-C and BSA-D), where the IC provides remote office test lines and 105 test lines with associated responders or their functional equivalent, will consist of monthly loss and C-message noise tests and annual balance test. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance tests, the IC may also order, at additional charges, gain-slope and C-notched noise testing.

The Telephone Company will provide a monthly AST report that lists the test results from each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

Issued: March 26, 1993



P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(Continued)

13.

RT)

Access Services Tariff Section 13 1st Revised Sheet 18 Replacing Original-Sheet-18 REGENVEU ACCESS SERVICES ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEQUS SERVICES 7 1986 MISSOURI 13.3 Miscellaneous Services-(Continued) Public Service Commission 13.3.5 Testing Services-(Continued)

- A. Switched Access Service-(Continued)
  - Automatic Scheduled Testing 2.

Automatic Scheduled Testing (AST) of Switched Access Services (Feature Groups B, C and D), where the IC provides remote office test lines and 105 test lines with associated responders or their functional equivalent, will consist of monthly loss and C-message noise tests and annual balance test. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/ balance tests, the IC may also order, at additional charges, gainslope and C-notched noise testing.

The Telephone Company will provide a monthly AST report that lists the test results from each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

CANCELLED APR 11 1993 BY 2 nd R.S. 18 Public Service Commission MISSOURI



Issued: JUN 27 1980

Effective:

JUL

1 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 Original Sheet 18

DEC 20 ( 1)

自由的资

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.5 Testing Services-(Continued)

A. Switched Access Service-(Continued)

2. Automatic Scheduled Testing

Automatic Scheduled Testing (AST) of Switched Access Services (Feature Groups B, C and D), where the IC provides remote office test lines and 105 test lines with associated responders or their functional equivalent, will consist of monthly loss and C-Message Noise tests and annual balance test. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/ balance tests, the IC may also order, at additional charges, gainslope and C-Notched Noise testing.

With Feature Group B Switched Access Service, AST is only provided to IC's who order it, i.e., AST is optional. However, with Feature Groups C and D, AST is nonoptional, and IC's with these Switched Access Services will be required to pay for AST, whether or not they have ordered it. With Feature Group C, where AST is not available in a Telephone Company office, Cooperative Scheduled Testing (CST) will be provided on a nonoptional basis. With Feature Group D, CST or Manual Scheduled Testing (MST) may be specified by the IC in lieu of AST. Trunks from a Telephone Company digital switch (e.g., 4E, 5E or DMS) to an IC digital switch utilizing digital facilities are excluded from mandatory routine testing.

The Telephone Company will provide a monthly AST report that lists the test results from each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

GANGELLED JUI 1 1986 PUBLIC SERVICE COMMUNICATION OF MISSCUR JAN 0 1 1984 Effective: Issued: DEC 2 9 1983 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Ubio Service Commission St. Louis, Missouri

Access Services Tariff Section 13 2nd Revised Sheet 19 Replacing 1st Revised Sheet 19

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - A. Switched Access Service-(Continued)
      - 3. Cooperative Scheduled Testing

Cooperative Scheduled Testing (CST) of Switched Access Services (FGB, FGC, FGD, BSA-B, BSA-C, BSA-D and Directory Access Service not routed through an access tandem), where the Telephone Company provides a technician at its office(s), and the IC provides a technician at its terminal location(s) or with suitable test equipment to perform the required tests, will consist of quarterly loss and C-message noise tests and annual balance tests. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance measurements, the IC may also order, at additional charges, gain-slope and C-notched noise testing.

The Telephone Company will provide, on a quarterly basis, a CST report that lists the test results for each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

Issued: March 26, 1993



P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

13.

(RT)

pplement to this f will be issued	Access Services Tariff Section 13
t for the purpose	lst Revised Sheet 19
nceling this tariff.	Replacing Original Sheet 19
ACCESS SERVICES	REGENVISIO
ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND (Continued)	
3 Miscellaneous Services-(Continued)	JUN 27 1986
	MISSUURI

Public Service Commis

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - A. Switched Access Service-(Continued)
      - 3. Cooperative Scheduled Testing

Cooperative Scheduled Testing (CST) of Switched Access Services (Feature Groups B, C, D and Directory Access Service not routed through an access tandem), where the Telephone Company provides a technician at its office(s), and the IC provides a technician at its terminal location(s) or with suitable test equipment to perform the required tests, will consist of quarterly loss and C-message noise tests and annual balance tests. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance measurements, the IC may also order, at additional charges, gain-slope and C-notched noise testing.

The Telephone Company will provide, on a quarterly basis, a CST report that lists the test results for each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

CANCELLED APR 11 1393 -BY2 mit R.S. Public Service Commission MISSOUR 1986 調料 1 mmission

JUN 27 1966 Issued:

JUL 1 1986 Effective:

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 19

DEC 20 Low

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.5 Testing Services-(Continued)

- A. Switched Access Service-(Continued)
  - 3. Cooperative Scheduled Testing

Cooperative Scheduled Testing (CST) of Switched Access Services (Feature Groups B, C, D and Directory Access Service not routed through an access tandem), where the Telephone Company provides a technician at its office(s), and the IC provides a technician at its terminal location(s) or with suitable test equipment to perform the required tests, will consist of quarterly loss and C-Message Noise tests and annual balance tests. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance measurements, the IC may also order, at additional charges, gain-slope and C-Notched Noise testing.

With Feature Group B Switched Access Service, CST is only provided to IC's who order it, i.e., CST is optional. However, with Feature Groups C and D, CST may be provided as an alternative to the nonoptional AST. Trunks from a Telephone Company digital switch (e.g., 4E, 5E or DMS10) to an IC digital switch utilizing digital facilities are excluded from mandatory routine testing.

The Telephone Company will provide, on a quarterly basis, a CST report that lists the test results for each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

GANGELLED JUL 1 1986 AV 124 R S. 19

na sterest

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984AN - 1 (SO) By R. D. BARRON, Vice President-Missouri 83-253 Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

Access Services Tariff Section 13 2nd Revised Sheet 20 Replacing 1st Revised Sheet 20

## ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - A. Switched Access Service-(Continued)
      - 4. Manual Scheduled Testing

Manual Scheduled Testing (MST) of Switched Access Services (FGB, FGC, FGD, BSA-B, BSA-C, BSA-D and Directory Access Service not routed through an access tandem), where the Telephone Company provides a technician at its office(s) and at the IC terminal location, will consist of quarterly loss and C-message noise tests and annual balance tests. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance tests, the IC may also order, at additional charges, gain-slope and C-notched noise testing.

The Telephone Company will provide, on a quarterly basis, a MST report that lists the test results for each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

FILED MO PSC P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpos of canceling this tar

13.

RT)

supplement to this	Access Services Tariff
riff will be issued	Section 13
cept for the purpose	1st Revised Sheet 20
canceling this tariff.	Replacing Original Sheet 20
ACCESS SERVIC	
(Continued)	JUN 27 1980
13.3 Miscellaneous Services-(Continued)	MISSUURI
13.3.5 Testing Services-(Continued)	Public Semise Original

- A. Switched Access Service-(Continued)
  - 4. Manual Scheduled Testing

Manual Scheduled Testing (MST) of Switched Access Services (Feature Groups B, D and Directory Access Service not routed through an access tandem), where the Telephone Company provides a technician at its office(s) and at the IC terminal location, will consist of quarterly loss and C-message noise tests and annual balance tests. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance tests, the IC may also order, at additional charges, gain-slope and C-notched noise testing.

The Telephone Company will provide, on a quarterly basis, a MST report that lists the test results for each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

CANCELLED

Public Service Commission

APR 11 1393 #200 BY 2 m R.S. 200 Public Service Commission MISSOURI



Issued: JUN 27 1986

**Effective:** 1 1986 JUL

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 20

DEC 2 1 1

i de forma a compaña,

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.5 Testing Services-(Continued)

- A. Switched Access Service-(Continued)
  - 4. Manual Scheduled Testing

Manual Scheduled Testing (MST) of Switched Access Services (Feature Groups B, D and Directory Access Service not routed through an access tandem), where the Telephone Company provides a technician at its office(s) and at the IC terminal location, will consist of quarterly loss and C-Message Noise tests and annual balance tests. However, the IC may specify a more frequent schedule of tests. In addition to the loss/noise/balance tests, the IC may also order, at additional charges, gain-slope and C-Notched Noise testing.

With Feature Group B Switched Access Service, MST is only provided to IC's who order it, i.e., MST is optional. However, with Feature Group D, MST may be provided as an alternative to the nonoptional AST. Trunks from a Telephone Company digital switch (e.g., 4E, 5E or DMS10) to an IC digital switch utilizing digital facilities are excluded from mandatory routine testing.

The Telephone Company will provide, on a quarterly basis, a MST report that lists the test results for each trunk tested. Trunk test failures requiring IC participation for trouble resolution will be provided to the IC on an as-occurs basis.

GANGELLED JUL 1 1986 PUBLIC SE RVICE COMMISLIUM FILED CHE ANISSCOULD 88 JAN 0 1 1984 DEC 2 9 1983 Effective: Issued: By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 21

## ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - A. Switched Access Service-(Continued)
      - 5. Nonscheduled Testing

Nonscheduled Testing (NST) of Switched Access Services is where:

- the IC provides remote office test lines and 105 test lines with associated responders or their functional equivalent ("automatic testing"), or
- the Telephone Company provides a technician at its office(s), and the IC provides a technician at its terminal location(s) with suitable test equipment to perform the required tests ("cooperative testing"), or
- the Telephone Company provides a technician at its office(s), at IC terminal location(s) or End User's premises with suitable test equipment to perform the required tests ("manual testing").

Nonscheduled tests may consist of any tests, e.g., loss, noise, slope and envelope delay which the IC may require.

- 6. Obligations of the IC
  - a. The IC shall provide the Remote Office Test Line priming data to the Telephone Company, as appropriate, to support AST as set forth in Paragraph 13.3.5, A.2., preceding, or NST as set forth in Paragraph 13.3.5, A.5., preceding.
  - b. The IC shall make the facilities to be tested available to the Telephone Company at times mutually agreed upon.

Issued:

December 29, 1993

Effective: January 1, 1984

**FILED** MO PSC

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 22 Replacing Original Sheet 22

### ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - B. Special Access Service

The Telephone Company will, at the request of an IC, provide assistance in performing specific tests requested by the IC.

1. Additional Cooperative Acceptance Testing (ACAT)

When an IC provides a technician at its terminal location or at the End User's premises with suitable test equipment to perform the required tests, the Telephone Company will provide a technician at its office for the purpose of conducting Additional Cooperative Acceptance Testing. At the IC's request, the Telephone Company will provide a technician at the IC terminal location or at the End User's premises. These tests may, e.g., consist of the following:

- VG1 through VG5: Attenuation Distortion, C-Message Noise and Echo Control
- VG6 through VG12: Attenuation Distortion, C-Message Noise, Echo Control, Impulse Noise, Phase Jitter, Intermodulation Distortion, Envelope Delay Distortion and Frequency Shift.
- 2. Nonscheduled Testing (NST)

When an IC provides a technician at its terminal location or at the End User's premises with suitable test equipment to perform the required tests, the Telephone Company will provide a technician at its office for the purpose of conducting Nonscheduled Testing. At the IC's request, the Telephone Company will provide a technician at the IC terminal location or at the End User's premises. Nonscheduled tests may consist of any tests, e.g., loss, noise, slope and envelope delay which the IC may require.

Effective: July 1, 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff Section 13 Original Sheet 22

1 DEC 2003

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.5 Testing Services-(Continued)

B. Special Access Service

The Telephone Company will, at the request of an IC, provide assistance in performing specific tests requested by the IC.

1. Additional Cooperative Acceptance Testing (ACAT)

When an IC provides a technician at its terminal location or at the End User's premises with suitable test equipment to perform the required tests, the Telephone Company will provide a technician at its office for the purpose of conducting Additional Cooperative Acceptance Testing. At the IC's request, the Telephone Company will provide a technician at the IC terminal location or at the End User's premises. These tests may, e.g., consist of the following:

- VG1 through VG3 and VG5: Attenuation Distortion, C-Message Noise and Echo Control.
- VG6 through VG10: Attenuation Distortion, C-Message Noise, Echo Control, Impulse Noise, Phase Jitter, Intermodulation Distortion, Envelope Delay Distortion and Frequency Shift.
- 2. Nonscheduled Testing (NST)

When an IC provides a technician at its terminal location or at the End User's premises with suitable test equipment to perform the required tests, the Telephone Company will provide a technician at its office for the purpose of conducting Nonscheduled Testing. At the IC's request, the Telephone Company will provide a technician at the . IC terminal location or at the End User's premises. Nonscheduled tests may consist of any tests, e.g., loss, noise, slope and envelope delay which the IC may femilities

and the second second

JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSOURI JAN 0 1 1984 Effective: Issued: DEC 2 9 1983 By R. D. BARRON, Vice President-Missouri Public Service Contribution Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 23 **Replacing Original Sheet 23** 

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - B. Special Access Service-(Continued)
      - 3. Obligation of the IC

When the IC subscribes to Testing Service as set forth in this Section, the IC shall make the facilities to be tested available to the Telephone Company at times mutually agreed upon.

C. Rates and Charges

Commission - 02/16/2003 - IN-2003-0247

- 1. Switched Access
  - a. Additional Cooperative Acceptance Testing

Public Service O	Testing Periods	<u>USOC</u>	First Half Hour or Fraction <u>Thereof</u>	Each Additional Half Hour or Fraction <u>Thereof</u>
(RT) innoss (CR) (CR)	Basic Time, per technician	UBCX+	\$73.37	\$18.78
(RT) (CR)	Overtime, per technician	UBCX+	76.71(1)	22.12(1)
(RT) III (CR) V	Premium Time, per technician	UBCX+	80.04(1)	25.45(1)

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.

July 1, 1986

Issued: June 27, 1986

Access Services Tariff Section 13 Original Sheet 23

DEC 20 1000

Public Sernus Commission

#### ACCESS SERVICES

13. ADDITIONAL ENGINFERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - B. Special Access Service-(Continued)
      - 3. Obligation of the IC

When the IC subscribes to Testing Service as set forth in this Section, the IC shall make the facilities to be tested available to the Telephone Company at times mutually agreed upon.

C. Rates and Charges

- 1. Switched Access
  - a. Additional Cooperative Acceptance Testing

Testing Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or ' Fraction Thereof
Basic Time, regularly scheduled working hours, per technician	. UBC	\$49.40	\$19.80
Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician	. UBC	53.09(1)	23.50(1)
Premium Time, outside scheduled work day BAT per technician	NOBLL	<b>臣D</b> 56.78(1)	27.19(1)
	C SERVICE COM	<del>ک</del> کر	
(1) A call-out of a Telephone Company employee's scheduled work period is supplying the second state of			
By R. D. BARRON, Vic Southwestern Bell		-Missouri 8	DAN = 1. (53.) <b>3 - 2 5 8</b> Service Commission

. منابع محمد وسعا م

Access Services Tariff Section 13 1st Revised Sheet 24 **Replacing Original Sheet 24** 

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

#### 13.3 Miscellaneous Services-(Continued)

# 13.3.5 Testing Services-(Continued)

- C. Rates and Charges-(Continued)
  - 1. Switched Access-(Continued)
    - b. Automatic Scheduled Testing (AST)

	16/2003	To First Point _of Switching_	<u>USOC</u>	Monthly <u>Rates</u>
(CT)	- 02/	Basic Tests(1)(3)		
(CR)	ervice Commission	1004 Hz Loss Tests(2) performed within a one-year period, per test ordered, per transmission path	UBGX+	\$0.001
(CR)	CANCELLED - Missouri Public Service Commission - 02/16/2003	C-Message Noise Tests(2) performed within a one-year period, per test ordered, per transmission path	UBGX+	0.001
(CP)	CANCELLED -	Return Loss (Balance) Tests(2) performed within a one- year period, per test ordered, ner transmission path	LIBCY+	0.001
(CR)		per transmission path	UBGX+	0.001

- (1) These three tests represent the minimum offering, i.e., an order for testing must at a minimum consist of twelve 1004 Hz Loss Tests per transmission path, 12 C-Message Noise Tests per transmission path and one Return Loss (Balance) Test per transmission path, per year. The additional tests listed may be ordered by the IC at additional charges 60 days prior to the start of the IC-prescribed schedule.
- (2) The IC may specify a more frequent schedule of tests, 60 days prior to the start of the IC-prescribed schedule.
- (3) Subject to a one-year minimum contract period and annually thereafter.



- IN-2003-0247

----

1

ļ

1.1.1

ļ

. 1

•	No supplement to this tariff will be issued except for the purpose of canceling this tariff. ACCESS SERVICE	Access Services Tariff Section 13 Original Sheet 24
•	13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR (Continued)	AND MISCELLANEOUS SERVICES-
	13.3 Miscellaneous Services-(Continued)	
	13.3.5 Testing Services-(Continued)	
	C. Rates and Charges-(Continued)	
	1. Switched Access-(Continued)	Public Sonder Commission
-	b. Automatic Scheduled Testing	(AST)
	To First Point of Switching	Monthly USOC Rates
	Basic Offering(1)(3)	
•	<pre>1004 Hz Loss Tests(2) performed within a one-year period,</pre>	. UBCX+ \$ 0.07
	per transmission path. Return Loss (Balance) Tests(2) performed within a one- year period, per test ordered, JU	_
	per transmission path+	2.5 UBGX+ 0.07
•	<ul> <li>(1) These three tests represent the minimum of must at a minimum consist of twelve 1004 must at a minimum contract preserved and the intervent of the IC-prescribed schedule.</li> <li>(2) The IC may specify a more frequent schedule start of the IC-prescribed schedule.</li> <li>(3) Subject to a one-year minimum contract preserved and the intervent schedule.</li> </ul>	Hz Loss Tests per transmission Hz Loss Tests per transmission ission path and one Return Loss year. The additional tests listed arges 60 days prior to the start le of tests, 60 days prior to the
	Issued: DEC 2 9 1983 Effe	ctive: JAN 0 1 1984 (AN - 1 62)
	By R. D. BARRON, Vice Pr Southwestern Bell Tel St. Louis, Mi	ephone Company Public Service Commission

Access Services Tariff Section 13 1st Revised Sheet 25 Replacing Original Sheet 25

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)
        - b. Automatic Scheduled Testing (AST)-(Continued)

To First Point of Switching	<u>USOC</u>	Monthly <u>Rates</u>
Additional Tests		
Gain-Slope Tests performed within a one-year period, per test ordered, per transmission path	UBGX+	\$ 0.001
C-Notched Noise Tests performed within a one- year period, per test ordered, per transmission path	UBGX+	0.001

Issued: June 27, 1986



Access Services Tariff Section 13 Original Sheet 25

#### ACCESS SERVICES

13. ADDITIONAL ENGINEFRING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) 13.3 Miscellaneous Services-(Continued) DEC 2 C (DIG 13.3.5 Testing Services-(Continued) C. Rates and Charges-(Continued) Public Special Commission 1. Switched Access-(Continued) Automatic Scheduled Testing (AST)-(Continued) Ъ. To First Point Monthly USOC of Switching Rates Additional Tests Gain-Slope Tests performed within a one-year period, per test ordered, \$ 0.07 per transmission path. . . UEGX+ C-Notched Noise Tests performed within a oneyear period, per test ordered, per transmission path. . . UBCX+ 0.07

GANGELLED

JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 26 Replacing Original Sheet 26

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

# 13.3.5 Testing Services-(Continued)

## C. Rates and Charges-(Continued)

- 1. Switched Access-(Continued)
  - c. Cooperative Scheduled Testing (CST)

To First Point		Monthly
<u>of Switching</u>	<u>USOC</u>	Rates
Basic Tests(l)(3)		
1004 Hz Loss Tests(2)		
performed within a		
one-year period,		
per test ordered,		
per transmission path	UBSX+	\$0.76
C-Message Noise Tests(2) performed within a one-year period, per test ordered,		
per transmission path	UBSX+	0.72
Return Loss (Balance) Tests(2) performed within a one- year period, per test ordered,		
per transmission path	UBSX+	1.06

- (1) These three tests represent the minimum offering, i.e., an order for testing must at a minimum consist of four 1004 Hz Loss Tests per transmission path, four C-Message Noise Tests per transmission path and one Return Loss (Balance) Test per transmission path, per year. The additional tests listed may be ordered by the IC at additional charges 60 days prior to the start of the IC-prescribed schedule.
- (2) The IC may specify a more frequent schedule of tests, 60 days prior to the start of the IC-prescribed schedule.
- (3) Subject to a one-year minimum contract period and annually thereafter.



C. Rate 1. 1 (CT) (CT) (CR) (CR) (CR) (CR) (CR) (CR) (CR)

P	S	¢	Mo	•	-No	•	- 36

No supplement to tariff will be is except for the pu	sued rpose		Access Services Tariff Section 13 Original Sheet 26
of canceling this	ACCFSS S	FRVICES	
13. ADDITIONAL E (Continued)	NGINFERING, ADDITIONAL	LABOR AND MISCELL	ANEOUS SERVICES-
13.3 Miscellan	eous Services-(Continue	d)	
13.3.5 Testi	ng Services-(Continued)		DEC 2 D C 3
C. Rates a	nd Charges-(Continued)		the first of the second se
1. Swi	tched Access-(Continued		Public Concerning and Islam
с.	Cooperative Scheduled	Testing (CST)	
	To First Point		Monthly
	of Switching	USOC	Rates
	<pre>Basic Offering(1)(3)</pre>		
	1004 Hz Loss Tests(2 performed within a one-year period, per test ordered, per transmission p C-Message Noise Test performed within a	ath UBSX+	\$ 0.38
	one-year period, per test ordered, per transmission p	ath UBSX+	0.28
	Return Loss (Balance) Tests(2) performed within a o year period,	ANCELLED JUL 1 1986	· · · · · · · · · · · · · · · · · · ·
	per test ordered, per transmission p BY PUS	A TH. R. S. JUBX+	0.62
must at a min four C-Messag Test per tran	ismission path, per year ne IC at additional char	04 Hz Loss Tests mission path and . The additional	per transmission path, one Return Loss (Balance) tests listed may be
(2) The IC may sp		chedule of tests.	60 days prior to the star
	one-year minimum contra	ct period and ann	ually thereafter.
Issued: DEC 29	1983	Fffactive: JAN	0 1 1984 JAN - 1 1984
			ouri 83-253

i

Ì

1

Access Services Tariff Section 13 1st Revised Sheet 27 **Replacing Original Sheet 27** 

# ACCESS SERVICES

# 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

- 13.3.5 Testing Services-(Continued)
- C. Rates and Charges-(Continued)

- 1. Switched Access-(Continued)
  - c. Cooperative Scheduled Testing (CST)-(Continued)

To First Point <u>of Switching</u>	<u>USOC</u>	Monthly <u>Rates</u>
Additional Tests		
Gain-Slope Tests performed within a one-year period, per test ordered, per transmission path	UBSX+	\$ 0.83
C-Notched Noise Tests performed within a one- year period, per test ordered, per transmission path	UBSX+	0.69
per transmission path	UBSX+	0.69

Issued:	June	2



Access Services Tariff Section 13 Original Sheet 27

#### ACCESS SERVICES

13. ADDITIONAL FNGINFERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) 13.3 Miscellaneous Services-(Continued) DEC 29 1203 13.3.5 Testing Services-(Continued) C. Rates and Charges-(Continued) Public Section Commiss 1. Switched Access-(Continued) c. Cooperative Scheduled Testing (CST)-(Continued) To First Point Monthly of Switching USOC Rates Additional Tests Gain-Slope Tests performed within a one-year period, per test ordered, \$ 0.38 per transmission path. . . UBSX+ C-Notched Noise Tests performed within a oneyear period, per test ordered, 0.28 per transmission path. . . UBSX+



2003-0247

### (CP)ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

#### 13.3 Miscellaneous Services-(Continued)

- 13.3.5 Testing Services-(Continued)
  - C. Rates and Charges-(Continued)
    - 1. Switched Access-(Continued)
      - d. Manual Scheduled Testing (MST)

03 - IN-S	To First Point of Switching	<u>USOC</u>	Monthly <u>Rates</u>
(CT) 000	Basic Tests(1)(3)		
(CR) (CCR)	1004 Hz Loss Tests(2) performed within a one-year period, per test ordered, per transmission path	UBMX+	\$1.51
ANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-	C-Message Noise Tests(2) performed within a one-year period, per test ordered, per transmission path	UBMX+	1.44
CELLED - Misso	Return Loss (Balance) Tests(2) performed within a one- year period, per test ordered,		
(CR) YO	per transmission path	UBMX+	2.11

- (1) These three tests represent the minimum offering, i.e., an order for testing must at minimum consist of four 1004 Hz Loss Tests per transmission path, four C-Message Noise Tests per transmission path and one Return Loss (Balance) Test per transmission path, per year. The additional tests listed may be ordered by the IC at additional charges, 60 days prior to the start of the IC-prescribed schedule.
- (2) The IC may specify a more frequent schedule of tests, 60 days prior to the start of the IC-prescribed schedule.
- (3) Subject to a one-year minimum contract period and annually thereafter.

Issued:	June 27, 1986	Effective:	July 1, 1986
	By R. D. BA	RRON, President-Missouri Div	ision
	Southwestern Bell Telephone Company		
		St. Louis, Missou	ri



• --

- 1

:

ì

No supplement to this tariff will be issued except for the purpose of canceling this tariff. ACCESS SERVICES	Access Services Tariff Section 13 Original Sheet 28
13. ADDITIONAL FNGINFERING, ADDITIONAL LABOR AN (Continued)	D MISCELLANEOUS SFRVICES-
13.3 Miscellaneous Services-(Continued)	DEC 20 213
13.3.5 Testing Services-(Continued)	
C. Rates and Charges-(Continued)	Public Streft a Commission
1. Switched Access-(Continued)	ter − terrer − , k.r%s. (k.  β/s + r/t (t), ( <del>t</del> <del>er minister, s i s.a.g.), (s general</del> )
d. Manual Scheduled Testing (MST)	
To First Point of Switching	USOC Rates
<pre>Basic Offering(1)(3)</pre>	
1004 Hz Loss Tests(2) performed within a one-year period, per test ordered, per transmission path	UBMX+ \$ 1.02
C-Message Noise Tests(2) performed within a one-year period, per test ordered, per transmission path	. URMX+ 0.67
per transmission path.	ELLED 1 1986 SUBMET 1.35 ICE COMMISSION
<ul> <li>(1) These three tests represent the minimum offer must at minimum consist of four 1004 Hz Loss four C-Message Noise Tests per transmission Test per transmission path, per year. The a ordered by the IC at additional charges, 60 IC-prescribed schedule.</li> </ul>	Missound ering, i.e., an order for testing a Tests per transmission path, path and one Return Loss (Balance) additional tests listed may be days prior to the start of the
<ul><li>(2) The IC may specify a more frequent schedule of the IC-prescribed schedule.</li><li>(3) Subject to a one-year minimum contract period</li></ul>	
Issued: DEC 2 9 1983 Effecti	ive: JAN 0 1 1884 83 - 258
By R. D. BARRON, Vice Presi Southwestern Bell Teleph St. Louis, Misso	hone Company
Access Services Tariff Section 13 1st Revised Sheet 29 Replacing Original Sheet 29

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)
        - d. Manual Scheduled Testing (MST)-(Continued)

To First Point _of Switching	<u>USOC</u>	Monthly <u>Rates</u>
Additional Tests		
Gain-Slope Tests performed within a one-year period, per test ordered, per transmission path	UBMX+	\$ 1.66
C-Notched Noise Tests performed within a one- year period, per test ordered, per transmission path	UBMX+	1.37

(2) CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Issued: June 27, 1986



#### P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 Original Sheet 29

#### ACCESS SERVICES

13. ADDITIONAL FNGINFFRINC, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)

Manual Scheduled Testing (MST)-(Continued) d.

To First Point of Switching

Additional Tests

Gain-Slope Tests performed within a one-year period, per test ordered, \$ 1.02 per transmission path. . . UBMX+

USOC

C-Notched Noise Tests performed within a oneyear period, per test ordered, per transmission path. . . UBMX+

# GANGELLED

JUL 1 1986 JETS ME & FINIMAS. OF MISSUR

JAN 0 1 1984 JAN -Issued: DEC 2 9 1983 Effective: 1 (934) By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Public mission St. Louis, Missouri

BEC 20 1003

Monthly

Rates

0.67

ette Commission

Access Services Tariff Section 13 1st Revised Sheet 30 Replacing Original Sheet 30

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

- 13.3.5 Testing Services-(Continued)
  - C. Rates and Charges-(Continued)
    - 1. Switched Access-(Continued)
      - e. Nonscheduled Testing (NST)

Automatic Testing:

To First Point <u>of Switching</u>	<u>USOC</u>	Nonrecurring <u>Charges</u>
1004 Hz Loss, per test performed	USCX+	\$21.39
C-Message Noise, per test performed	USCX+	21.39
Return Loss (Balance), per test performed	USCX+	21.39
Gain-Slope, per test performed	USCX+	21.39
C-Notched Noise, per test performed	USCX+	21.39

Issued:

June 27, 1986



.

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 Original Sheet 30

DEC 20 (203

JAN - 1 1981

Public Service Commission

83-258

oren 82100

PLHO D. F

#### ACCESS SERVICES

13. ADDITIONAL ENGINFERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)
        - e. Nonscheduled Testing (NST)

Automatic Testing:

To First Point of Switching	USOC	Nonrecurring Charges
1004 Hz Loss, per test performed	USCX+	\$14.05
C-Message Noise, per test performed	USCX+	14.05
Return Loss (Balance), per test performed	USCX+	14.05
Cain-Slope, per test performed	USCX+	14.05
C-Notched Noise, per test performed	USCX+	14.05

GANGELLED

JUL 1 1986 8Y PUBLIC SERVICE COMMISSION OF HISSUUR

Issued: DEC 29 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 13 1st Revised Sheet 31 Replacing Original Sheet 31

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
  - C. Rates and Charges-(Continued)
    - 1. Switched Access-(Continued)
      - e. Nonscheduled Testing (NST)-(Continued)

Cooperative Testing:

Testing Periods USOC	First Half Hour or Fraction <u>Thereof</u>	Each Additional Half Hour or <u>Fraction Thereof</u>
Basic Time, per technicianUSSX+	\$73.37	\$18.78
Overtime, per technician USSX+	76.71(1)	22.12(1)
Premium Time, per technicianUSSX+	80.04(1)	25.45(1)

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Issued:

June 27, 1986

(RT) (CR)

(RT) (CR)

(RT) (CR)

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

July 1, 1986

(1) A ca

Issued:

Access Services Tariff Section 13 Original Sheet 31

Pable Service Clamesea

### ACCESS SERVICES

13. ADDITIONAL ENGINEFRING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 29 123

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)

#### Nonscheduled Testing (NST)-(Continued) e.

Cooperative Testing:

Testing Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or Fraction Thereof
Basic Time, regularly scheduled working hours, per technician	USS	\$49.40	\$19.80
Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician	USS	53.09(1)	23.50(1)
Premium Time, outside scheduled work day, per technician	USS	56.78(1)	27.19(1)
	GANCE	LLED	
	JUL 1	1986	
A call-out of a Telephone Company e	RY 124 (K HINK JEEN CR	S. 31	tive with the
employee's scheduled work period is			
ued: DEC 2 9 1983	Fffective:	JAN 0 1 984	JAN - 1 1934
By R. D. BARRON, V Southwestern Be		-Missouri	lic Service Commission

St. Louis, Misson

Access Services Tariff Section 13 1st Revised Sheet 32 Replacing Original Sheet 32

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)
        - e. Nonscheduled Testing (NST)-(Continued)

Manual Testing:

Testing Periods USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or <u>Fraction Thereof</u>
Basic Time, per technician USMX+	\$73.23	\$18.64
Overtime, per technicianUSMX+	76.60(1)	22.01(1)
Premium Time, per technician USMX+	79.98(1)	25.39(1)

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.





By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Access Services Tariff Section 13 Original Sheet 32

Public Service Commission

#### ACCESS SFRVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 20 (CO)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 1. Switched Access-(Continued)
        - e. Nonscheduled Testing (NST)-(Continued)

Manual Testing:

	Testing Periods	USOC	First Half Hour or Fraction Thereof	Fach Additional Half Hour or Fraction Thereof
	Basic Time, regularly scheduled working hours, per technician	USM	\$49.60	\$20.01
	Overtime, outside of regularly scheduled working hours on a scheduled work day,			
	per technician	USM	53.20(1)	23.61(1)
	Premium Time, outside scheduled work day,	USM	56.81(1)	27.21(1)
	per technician GA	NGEL		27.21(1)
,		JUL1 1	986	
	BY	ATR S.	32 CALADOLIUN	
	f a Telephone Company employ cheduled work period is subj	vee at ant	The not consecu	
Issued: DEC 29	9 1983 Eff	ective:	JAN 0 1 1984 J	AN - 1 (984)
	Fy F. D. BARRON, Vice E Southwestern Bell Te		Missouri	<b>3 - 253</b> Service Commission

St. Louis: Missouri

Access Services Tariff Section 13 1st Revised Sheet 33 Replacing Original Sheet 33

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.5 Testing Services-(Continued)
    - C. Rates and Charges-(Continued)
      - 2. Special Access
        - a. Additional Cooperative Acceptance Testing (ACAT)

Testing Periods USOC	First Half Hour or Fraction Thereof	Each Additional Half Hour or <u>Fraction Thereof</u>
Basic Time, per technician SNTX+	\$73.08	\$18.49
Overtime, per technician SNTX+	76.50	21.91(1)
Premium Time, per technician SNTX+	79.91(1)	25.32(1)

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

Issued:

June 27, 1986

(RT) (CR)

(RT) (CR)

(RT) (CR)

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

July 1, 1986

Access Services Tariff Section 13 Original Sheet 33

Public Sector Commission

#### ACCESS SERVICES

	OITIONAL E	NGINEERING,	ADDITIONAL	LABOR	AND	MISCELLANEOUS		2 9 1933	;
13.3	Miscellan	eous Service	es-(Contínue	ed)			•		

13.3.5 Testing Services-(Continued)

C. Rates and Charges-(Continued)

2. Special Access

a. Additional Cooperative Acceptance Testing (ACAT)

Testing Periods	USOC	First Half Hour or Fraction Thereof	Each Additional Falf Hour or Fraction Thereof
Basic Time, regularly scheduled working hours, per technician	 SNT	\$49.82	\$20.22
Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician	SNT	53.32(1)	23.72(1)
Premium Time, outside scheduled work day, per technician	SNT	56.83(1)	27.23(1)

GANGELLED

JUL 1 1986 PUBLIC SERVICE COMMISSION OF MISSOURI

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.
 Issued: DEC 2 9 1983 Effective: JAN 0 1984 JAN = 1 1934
 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 34 Replacing Original Sheet 34

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

- 13.3.5 Testing Services-(Continued)
  - C. Rates and Charges-(Continued)
    - 2. Special Access-(Continued)

b. Nonscheduled Testing (NST)

## First Half

Testing Periods USOC	Hour or Fraction <u>Thereof</u>	Each Additional Half Hour or <u>Fraction Thereof</u>
Basic Time, per technicianSNOX+	\$73.08	\$18.49
Overtime, per technicianSNOX+	76.50(1)	21.91(1)
Premium Time, per technician SNOX+	79.91(1)	25.32(1)

(1) A call-out of a Telephone Company employee at a time not consecutive with the employee's scheduled work period is subject to a minimum charge of four hours.



CANCELLED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

July 1, 1986

Issued:

Access Services Tariff Section 13 Original Sheet 34

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued) DEC 20 003 13.3 Miscellaneous Services-(Continued) 13.3.5 Testing Services-(Continued) C. Rates and Charges-(Continued) 2. Special Access-(Continued) b. Nonscheduled Testing (NST) First Half Hour or Each Additional Fraction Half Hour or Fraction Thereof Testing Periods USOC Thereof Basic Time, regularly scheduled working hours, \$20.22 SNO \$49.82 per technician. . . . Overtime, outside of regularly scheduled working hours on a scheduled work day, per technician. . . . SNO 53.32(1) 23.72(1)Premium Time, outside scheduled work day, SNO 56.83(1) 27.23(1)per technician. . . . GANBELLED JUL 1 1986

PUBLIC SERVICE COMMISSION OF MISSOURI

(1) A call-out of a Telephone Company employee at a time not consecutive with the 4employee's scheduled work period is subject to a minimum charge of four hours.

Effective:

DEC 2 9 1983 Issued:

JAN 0 1984 J/.N ~ 1 1934 88 - 253 By R. D. BARRON, Vice President-Missourt Public Service Commission Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff No supplement to this tariff will be issued Section 13 3rd Revised Sheet 35 except for the purpose Replacing 2nd Revised Sheet 35 of canceling this tariff. ACCESS SERVICES 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES 3 1994 ED - Missouri Public Service Commission - 02/16/2003 - IN-2003-0247 (Continued) MO. PUBLIC SERVICE COMM. 13.3 Miscellaneous Services-(Continued) 13.3.6 Provision of Access Service Billing Information: A. The customer will receive its monthly bills in a standard paper format or if requested by the customer, on magnetic tape in lieu of paper. Once billing has begun, the Telephone Company will accept subsequent customer requests for form changes in the primary billing data (e.g., paper to tape or tape to paper). The Access Billing Change charge set forth in Paragraph 13.3.6., D.1., following will apply for each form change requested by the customer per billing period per Revenue Accounting Office (RAO). The period of time to implement the changes will be determined by the Telephone Company. B. At the option of the customer, and for an additional charge: 1. Billing detail may be sent to the customer's premises by data transmission. 2. Additional copies of the customer's monthly bill may be provided (AT) in standard paper, magnetic tape, floppy disk or microfiche (RT) format. CANCELLI (LL) 3. When additional copies of the customer's bill are ordered by the customer, the Access Billing Change charge will apply per billing period per RAO for each form of the bill ordered. For subsequent changes in bill form or number of additional copies, the Access Billing Change charge set forth in Paragraph 13.3.6., D.1., following, will apply per customer request, per billing period per RAO. The period of time to implement these options or make changes will be determined by the Telephone Company. C. On an individual case basis, other forms of the billing data or the preparation of the billing data for data transmission will be provided when the Telephone Company has determined that these provisions are economically beneficial and the customer is mutually agreeable to the provisions. JAN 7 🕛 1995 Issued: NOV 2 3 1994 Effective: MISSOURI By HORACE WILKINS, JR., President-Missouri Public Service Commission Southwestern Bell Telephone St. Louis, Missouri

والمباحية الأ

P.S.C. Mo.-No. 36

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 13 2nd Revised Sheet 35 Replacing 1st Revised Sheet 35

FEB 1 7 1989

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANBOUS SERVICESCEIVED (Continued)
  - 13.3 Miscellaneous Services-(Continued)
- (CP) 13.3.6 Provision of Access Service Billing Information: MISSOUR!
  - A. The customer will receive its monthly bills in a standard paper format or if requested by the customer, on magnetic tape in lieu of paper.

Once billing has begun, the Telephone Company will accept subsequent customer requests for form changes in the primary billing data (e.g., paper to tape or tape to paper). The Access Billing Change charge set forth in Paragraph 13.3.6.,D.1., following will apply for each form change requested by the customer per billing period per Revenue Accounting Office (RAO).

The period of time to implement the changes will be determined by the Telephone Company.

- B. At the option of the customer, and for an additional charge:
  - 1. Billing detail may be sent to the customer's premises by data transmission.
  - 2. Additional copies of the customer's monthly bill may be provided in standard paper, magnetic tape or microfiche format at the charges specified in Paragraph 13.3.6., B.3., following.
  - 3. When additional copies of the customer's bill are ordered by the customer, the Access Billing Change charge will apply per billing period per RAO for each form of the bill (paper, magnetic tape, or microfiche) ordered. For subsequent changes in bill form or number of additional copies, the Access Billing Change charge set forth in Paragraph 13.3.6.,D.1., following, will apply per customer request, per billing period per RAO. The period of time to implement these options or make changes will be determined by the Telephone Company.

C. On an individual case basis, other forms of the billing data or the preparation of the billing data for data transmission will be provided when the Telephone Company has determined that these provisions are economically beneficial and the customer is mutually agreeable to the provisions.

FILED JAN MAR 3 0 1900 NA Public Service Commission Public Service Commission MISSOURI Issued: FEB 27 1999 Effective: MAR S0 1989 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 35 Replacing Original Sheet 35

### ACCESS SERVICES

## RECEIVED

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERDECES 9 1987 (Continued)
  - 13.3 Miscellaneous Services-(Continued)

## MISSOURI Public Service Commission

Ξ

- 13.3.6 Provision of Access Service Billing Information:
- The IC will receive its monthly bills in a standard paper format or if Α. requested by the IC, on magnetic tape in lieu of paper, subject to the following limitations:
  - Magnetic tape billing will be for all of the IC customers' 1. accounts in a Revenue Accounting Office (RAO).
  - 2. One magnetic tape will be provided per bill period per Revenue Accounting Office (RAO).
  - Β. At the option of the IC, and for an additional charge:
    - 1. Billing detail and/or information may be transmitted to the IC terminal location by data transmission.
    - 2. Additional copies of the IC monthly bill or service and features record may be provided in standard paper or microfiche format.
  - Upon acceptance by the Telephone Company of an order for data trans-C. mission, the Telephone Company will determine the period of time to implement the transmission of such material on an individual order basis.



(CP)

(RT)

(FC)

(FC)





1988 FEB 8

Public Service Commission

JAN 0 8 1988 Issued:

Effective: FEB 08 1988

Access Services Tariff Section 13 Original Sheet 35

i Poblic i

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

DEC 2 9 1983

Issued:

13.3.6 Provision of Access Service Billing Information

- A. The IC will receive its monthly bills in a standard paper format.
- B. At the option of the IC, and for an additional charge:
  - 1. IC monthly bills may be provided on magnetic tape,
  - 2. Billing detail and/or information may be transmitted to the IC terminal location by data transmission,
  - 3. Additional copies of the IC monthly bill or service and features record may be provided in standard paper or microfiche format.
- C. Upon acceptance by the Telephone Company of an order for data transmission, the Telephone Company will determine the period of time to implement the transmission of such material on an individual-order basis.
- D. The rates and charges for the provision of Access Service Billing Information are as follows:

	FID	Rates
1. Provision of Standard Billing Detail and/or Information in magnetic tape format, per record, up to 25 bytes	DMT	ICB rates and charges apply
CANCELLER	)	

FEB 8 1988

Effective:

By R. D. BARRON, Vice President-Missouri

Southwestern Bell Telephone Company St. Louis, Missouri

Public Service Commission MISSOURI

JAN 0 1 1984

83

Public Service Comm

Access Services Tariff Section 13 1st Revised Sheet 35.01 Replacing Original Sheet 35.01

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.6 Provision of Access Service Billing Information:-(Continued)

D. If charges for Access Services Billing exist in the Telephone Company's Intrastate Access Service Tariffs and its Interstate Access Service Tariff, the total charge to the customer will be calculated based on one half of the intrastate charge and one half of the interstate charge. If charges for Access Services Billing exist only in the Telephone Company's Intrastate Access Services Tariff, full intrastate charges will apply.

E. Standard Bill Periods for Switched Access Services and Special Access Services are as follows:

Switched Access FGA/BSA-A FGB/BSA-B FGC/BSA-C FGD/BSA-D 5th of the Month 5th of the Month 15th of the Month 5th and 15th of the Month 25th of the Month

Special Access

To the extent the Telephone Company can accommodate the customer's request to change an existing bill period, a nonrecurring charge will apply. The Billing Account Number (BAN) Administrative Change Charge as set forth in Section 6 does not apply when a customer requests a change of an existing bill period.

The nonrecurring charge for FGA and BSA-A is applied per line. For FGB, FGC, FGD, BSA-B, BSA-C and BSA-D, the nonrecurring charge is applied per end office and access tandem.

Issued: December 16, 1994

January 16, 1995

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri



Access Services Tariff Section 13 Original Sheet 35.01

#### ACCESS SERVICES

 ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

KC / 2.3 1994

- 13.3 Miscellaneous Services- (Continued)
  - 13.3.6 Provision of Access Service Billing Information: (Continued)ENVICE COMM.
- (AT)

(AT)

D. If charges for Access Services Billing exist in the Telephone Company's Intrastate Access Service Tariffs and its Interstate Access Service Tariff, the total charge to the customer will be calculated based on one half of the intrastate charge and one half of the interstate charge. If charges for Access Services Billing exist only in the Telephone Company's Intrastate Access Services Tariff, full intrastate charges will apply.

# CANCELLED

JAN 151995 Public Service Commission MISSOURI



**JAN 7** 1995

MISSOURI JAN 7 ervice Commission

Issued: NOV 2 3 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri

Effective:

Access Services Tariff Section 13 5th Revised Sheet 36

Replacing 4th Revised Sheet 36

ACCESS SERVICES

## DEC 16 1994

- ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-13. (Continued) MO. PUBLIC SERVICE COMM.
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.6 Provision of Access Service Billing Information: (Continued)
      - D. The rates and charges for the provision of Access Service Billing Information are as follows:

		Nonrecurring <u>Charge</u>
	s Billing Change charge er billing period, per RAO	\$ 7.25
		Nonrecurring
Second	udary Bill	<u>Charge</u>
- Pape	er	
	page	\$ .0325
per	tape	82.76
per	frame	.0009
per	record	ICB
-	ppy Disk record	.000550
		Nonrecurring <u>Charge</u>
3. Change	ge of Access Services Bill Period (NRBCH)	
FGA/B: - per	SSA-A Line	(NR) \$29.00
FGB/B		29.00
FGC/B	SSA-C	29.00
FGD/B		29.00
	e end office and access tandem al Access	(NR) 44.00

Issued:

#### JAN 1 6 1995 Effective:

DEC 1 6 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri

JAN 1 6 1995

MISSOURI Public Service Commission

(AT)

(AT)

	No supplement to this Acc	ess Services Tariff
	tariff will be issued	Section 13
_	except for the purpose 4	th Revised Sheet 36
	of canceling this tariff. Replacing 3	rd Revised Sheet 36
	ACCESS SERVICES	RECEIVED
	13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOU (Continued)	IS SERVICES 2 1994
	13.3 Miscellaneous Services-(Continued)	MO. PUBLIC SERVICE COMM.
	13.3.6 Provision of Access Service Billing Information:-(	(Continued)
	D. The rates and charges for the provision of Access Se Information are as follows:	ervice Billing
		Nonrecurring Charge
	<ol> <li>Access Billing Change charge per billing period, per RAO.</li> </ol>	\$ 7.25
(MT	)	<b>N</b>
(FC) (CT	) 2. Secondary Bill	Nonrecurring <u>Charge</u>
-	- Paper	
	per page	\$ .0325
	per tape	82.76
(MT) (CT	per frame	.0009
(MT) (CT (AT	-	ICB
(AT	) per record	(NR) .000550
	CANCELLED	



· | )

6.4N.F.

JAN 7 1995

MISSOURI JAN 7 PUBLIC Service Commission

Issued: NOV 2 3 1994

-

ļ

i

\_

Effective: 010 1994

By HORACE WILKINS, JR., President-Missouri Southwestern Bell Telephone St. Louis, Missouri ----

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

1

Access Services Tariff Section 13 3rd Revised Sheet 36 Replacing 2nd Revised Sheet 36

. - --

#### ACCESS SERVICES

## RECEIVED

.....

13.3	liscellaneous Services-(Cont	inued)	N Public Se	AISSOURI Irvice Commis
?) 13.3.	6 Provision of Access Serv	vice Billing Information:-	(Continued)	
D.	The rates and charges for Information are as follows		ervice Bill	ing
				curring arge
	1. Access Billing Change per billing period,	charge per RAO	\$7	.25
			FID	Rates
C)	<ol> <li>Data Transmission to a premises of Billing De and/or Information,</li> </ol>	etail		
	per record transmit	tted	BOD DT	ICB rates and charges apply
				curring arge
C)	<ol> <li>Additional copies of t monthly bill.</li> </ol>	the customer's		
				.0325 .76
		Be	02	.0009
		CANCELLED		
		JAN 71995 BY W BY H R.S. # Source MISSOURI	FI	LED
	D	BY 4 M.N.S. while Service Commission	MAR	3 0 1999
<u> </u>	F	MI22001	ublic Servic	
	FEB 27 1989			

St. Louis, Missouri

÷. ;

	P		S		С		Mo.	-N	0		- 36
--	---	--	---	--	---	--	-----	----	---	--	------

Access Services Tariff Section 13 2nd Revised Sheet 36 Replacing 1st Revised Sheet 36

RECEIVED

#### ACCESS SERVICES

DEC 29 1987

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS **MASSOURI** (Continued) **Public Service Commissior** 

13.3 Miscellaneous Services-(Continued)

13.3.6 Provision of Access Service Billing Information:-(Continued)

(MT)

(DR)

D. The rates and charges for the provision of Access Service Billing Information are as follows:

FID Rates (FC) 1. Data Transmission to an IC Terminal Location of Billing Detail and/or Information, per record transmitted . . BOD DT ICB rates . . . . and charges apply 2. Additional copies of IC (FC) monthly bill or service and features record in standard paper or microfiche format, per page . . . . . . NOB NEL ICB rates and per microfiche record. . . . . BOD FH charges apply CANCELLED MAR 30 1989 Public Service Commission FILED FEB 8 1988 Public Service Commission Effective: FEB 08 1988 Issued: JAN 0.8 1988 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company

St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.	Access Services Tariff Section 13 lst Revised Sheet 36 Replacing Original Sheet 36
ACCESS SERVICES	
13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND (Continued)	MISCELLANEOUSSERVICES
13.3 Miscellaneous Services-(Continued)	JUN 27 1980
<ul><li>13.3.6 Provision of Access Service Billing I</li><li>D. Rates and charges for the provision of Information are as follows:-(Continued)</li></ul>	Access Service Service Commission
	FID Rates_
<ol> <li>Data Transmission to an IC Terminal Location of Billing Detail and/or Information, per record transmitted</li> </ol>	• BOD DT ICB rates and charges apply
3. Additional copies of IC monthly bill or service and features record in standard paper or microfiche format, per page	. NOB NEL ICB rates and . BOD FH charges apply

CANCELLED FEB 8 1988 BY 2 March P. S.# 36 Public Service Commission MISSOURI

FALED JUL 1 1986 86-84 Public Service Commission

Issued: JUN 27 1986

- -

3

1

Effective: JUL 1 1986

Access Services Tariff Section 13 Original Sheet 36

DEC 20 (23)

Public Service Commission

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)
  - 13.3 Miscellaneous Services-(Continued)
    - 13.3.6 Provision of Access Service Billing Information-(Continued)
      - D. Rates and charges for the provision of Access Service Billing Information are as follows:-(Continued)

		FID	<u>Rates</u>
2.	Data Transmission to an IC Terminal Location of Billing Detail and/or Information, per record transmitted	. BOD DT	ICB rates and charges apply
3.	Additional copies of IC monthly bill or service and features record in standard paper or microfiche format, per page per microfiche record		ICB rates and charges apply

GANGELLED

JUL 1 1986

ΒY PUBLIC SERVICE COMMISSION

OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

日日に民団 リスペーコ 1983 1984 **83 - 253** Public Service Commussion

(RT) (RT)

Access Services Tariff Section 13 1st Revised Sheet 37 Replacing Original Sheet 37

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

13.3 Miscellaneous Services-(Continued)

13.3.7 Protective Connecting Arrangements

The following Protective Connecting Arrangements (PCA's) are grandfathered and are offered subject to on-the-shelf availability:

- IN-200	Description	<u>USOC</u>	Monthly <u>Rates</u>	Nonrecurring <u>Charges</u>
(XD) Service Commission - 02/16/2003 - IN-200	Automatic PCA with a contact type signaling interface for two- or four- wire voice-band connections of CPE communications systems to Telephone Company Special Access Services	CDQ	\$34.42	None
	Automatic PCA for connection of a customer, authorized user or joint user-provided communications system arranged for CPE dial or automatic channel signaling to a Telephone Company Special Access Service	C234W	10.48	\$162.87
(CR) (RT) (RT) (AT) (AT) (CT)	PCA which provides for connection of CPE automatic telephone answering devices to Telephone Company Access Service by means of a two-wire interface	PA6++	ICB rates and c	harges apply.
CANCE	PCA for connection of answering or recording equipment to Telephone Company lines for one-way voice transmission in each direction but not simultaneously. Recording of two-way conversations is prevented			
(CR)	by the PCA	RDL	5.28	82.38

Effective: July 1, 1986



Access Services Tariff Section 13 Original Sheet 37

stantsalon

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.7 Protective Connecting Arrangements

The following Protective Connecting Arrangements (PCA's) mentioned in Section 2, Paragraph 2.5.5, B., preceding, are grandfathered and are offered subject to on-the-shelf availability:

Descr	iption	USOC	Monthly Rates	Nonrecurring Charges
Automatic PCA with signaling interfac wire voice-band co communications sys Company Special Ac	e for two- or four- nnections of CPE tems to Telephone	. CDO	\$ 6.61	None
Automatic PCA for customer, authoriz user-provided comm arranged for CPE d channel signaling Company Special Ac terminates at the PBX arranged for d signaling (four-wi	ed user or joint unications system ial or automatic to a Telephone cess Service which distant end in a ial or automatic	. C234W	11.40	98.40
-	wire interface	. cts NGELL	ICB rates and charges	ICB rates and charge apply
not simultaneously two-way conversati	one-way voice one-way voice the direction but BY Recording of PUBLIC ions is prevented	OF MISSOUR	37 MISSION	07.70
by the PCA		. RDL	6.10	34.72
sued: DEC 2 9 1983	Ef	fective:	JAN 0 1 1984 J	
F	y R. D. BARRON, Vice Southwestern Bell T St. Louis,	elephone Co		<b>8 3 - 2 5 8</b> Service Commission

Access Services Tariff Section 13 1st Revised Sheet 38 Replacing Original Sheet 38

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

13.3.7 Protective Connecting Arrangements-(Continued)

(LJ) IN-2003-0247	Description	<u>USOC</u>	MonthlyNonrecurringRatesCharges
(CT) - (CT) - (CT)	PCA for use with CPE answer only equipment where two-way trans- mission is required	PFZ++	ICB rates and charges apply.
(CT) [7]	Same application as PFZ++ with voice control disconnect and automatic receive volume limiting	PF9++	ICB rates and charges apply.
vice Commission	PCA for use with CPE to provide data on PBX trunks. Also requires standard PBX trunk PCA	PGA++	ICB rates and charges apply.
(CT) (CT) (ITA)	PCA to permit connection of CPE mes- sage registers to Telephone Company Switched Access Service for indi- cations of message registration for outgoing calls over the asso-		
(RT) -	ciated central office trunks	PGB++	ICB rates and charges apply.
CANCELLED	Alarm coupler for use with rotary dial, one-way transmission CPE alarm signaling device	PGH++	ICB rates and charges apply



Access Services Tariff Section 13 Original Sheet 38

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

DEC 2 ) (...)

i Public Certica Commission

13.3 Miscellaneous Services-(Continued)

13.3.7 Protective Connecting Arrangements-(Continued)

Description	USOC	Monthly <u>Rates</u>	Nonrecurring Charges
Similar application to RDL, but provides for simultaneous two-wa voice transmission. Recording o two-way voice conversations is prevented as with RDL	f	ICB rates and charges apply	ICB rates and charges apply
Same application as RDM with voi control disconnect and automatic receive volume limiting		ICB rates and charges apply	ICB rates and charges apply
PCA for use to provide data on PBX trunks. Also requires standard PBX trunk PCA		ICB rates and charges apply	ICB rates and charges apply
PCA to permit connection of mes- sage registers to facilities of the Telephone Company for indi- cations of message registration for outgoing calls over the asso ciated central office trunks (facilities). Each trunk would also have a PCA (typically CDH or CD8) for connection of the PBX. Association of the trunk with the station is made by the CPE	- GANGE	1986 5. 38 OM MISSION	ICB rates and charges apply
Alarm coupler for use with rotar dial, one-way transmission alarm signaling device		ICB rates and charges apply	ICB rates and charges apply
Southwestern	Effective: , Vice President Bell Telephone Louis, Missouri	t-Missouri	<del>Unit - 1 (83) -</del> 83 - 253 Service Commission

Access Services Tariff Section 13 1st Revised Sheet 39 Replacing Original Sheet 39

## ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

## 13.3.7 Protective Connecting Arrangements-(Continued)

	IN-2003-0247	Description	<u>USOC</u>	Monthly <u>Rates</u>	Nonrecurring <u>Charges</u>
(AT) (CT)	1.1	PCA to permit the connection of CPE to Telephone Company Switched Access Ser- vice arranged for two-way service,			
(CT)	- 02/16/2003	i.e., outward dialing by hotel/motel guests and re-ring by the operator of the IC long distance switchboard			
(CT)	ission	(the equivalent of a toll terminal)	PDA++	ICB rates and c	harges apply.
	Commissior	For termination of CPE tie lines, with CPE channel signaling, in			
(CR)	Service (	Centrex systems four-wire	C2H	\$22.55	\$348.30
(AT)		PCA used for automatic connection of CPE voice transmitting and/or re-			
(111)	Public	ceiving terminal equipment to Tele-			
(CR)	uriF	phone Company Switched Access Service	C2ACP	6.66	52.90
(AT)	- Missouri	PCA to provide for connection of CPE terminal equipment to Telephone			
(AT)	CANCELLED	Company Switched Access Service via three-wire interface	PDJ++	ICB rates and c	harges apply.
	O				

Issued: June 27, 1986

Effective: July 1, 1986



Access Services Tariff Section 13 Original Sheet 39

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES 200000 (Continued)

13.3 Miscellaneous Services-(Continued)

Public Service Commission

13.3.7 Protective Connecting Arrangements-(Continued)

Description		USOC	Monthly <u>Rates</u>	Nonrecurring Charges
PCA to permit the connection of a Telephone Company special record- ing trunk arranged for two-way service, i.e., outward dialing by hotel/motel guests and re-ring by operator of the Telephone Company long distance switchboard (the				
equivalent of a toll terminal)	• • • •	CED	ICB rates and charges apply	ICB rates and charges apply
For termination of CPF tie lines, with CPE channel signaling, in Centrex systems four-wire		C2H	\$8.13	\$24.39
PCA used for automatic connection of voice transmitting and/or re- ceiving terminal equipment to Telephone Company Access Service.	• • • •	C2ACP	10.61	8.81
PCA to provide for connection of terminal equipment to Telephone Company Access Service via three- wire interface		STC	ICB rates and charges apply	ICB rates an charges appl
	GAN	GELL	ED	
	JUL BY 2 PUB-IC SEF	<u>R .S.</u> 3	9	

Issued: DEC 2 9 1983

Fffective: JAN 0 1 1984

FIIL民日 37.11-1.1557 984 83-253 71513 Service to converse

Py F. D. BARRON, Vice President-Missouri Southwestern Rell Telephone Company St. Louis, Missouri

Access Services Tariff Section 13 1st Revised Sheet 40 Replacing Original Sheet 40

## (CP)ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

## 13.3.7 Protective Connecting Arrangements-(Continued)

	Description	<u>USOC</u>	Monthly <u>Rates</u>	Nonrecurring <u>Charges</u>
(AT)	PCA for connection of CPE voice commu- nications systems and/or terminal equipment via two-wire interface to			
(AT)	Telephone Company Switched Access Service (only loop-start trunks not equipped for toll diversion) or			
(CT)	equipment	PDK++	ICB rates and	charges apply
(AT)	Mannual PCA used to connect a cord switchboard position of a CPE system, which provides supervisory signals,			
(CT)	to Telephone Company Switched Access Service	PDQ++	ICB rates and	charges apply
(AT)	Automatic PCA used to connect Telephone Company Switched Access Service arranged for two-way combination service to and from the attendant position and from			
(AT) (CR)	the dial switching equipment of a CPE system.	CDH	\$ 6.46	\$53.80
(AT)	Automatic PCA used to connect Telephone Company Switched Access Service arranged for one-way incoming service to the			
(CT)	attendant position of CPE system	PDV++	ICB rates and	charges apply
(CT)	Automatic PCA used to connect Telephone Company Switched Access Service arranged for one-way outgoing service from the attendant position			
(CT)	of a CPE system	DZ++	ICB rates and	charges apply

Issued:

June 27, 1986

Effective: July 1, 1986



Access Services Tariff Section 13 Original Sheet 40

#### ACCESS SERVICES

- 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES 20.22
  - 13.3 Miscellaneous Services-(Continued)

EIS AUT i Public Screece Commission

13.3.7 Protective Connecting Arrangements-(Continued)

Description	USOC	Monthly Rates	Nonrecurring Charges
PCA for connection of voice commu- nications systems and/or terminal equipment via two-wire interface to Telephone Company Access Service (only loop-start trunks not equipped for toll diversion) or terminal			
equipment	STP	ICB rates and charges apply	ICB rates and charges apply
Manual PCA used to connect a cord switchboard position of a system, which provides supervisory signals,			
to Telephone Company Access Service	CDA	ICB rates and charges apply	ICB rates and charges apply
Automatic PCA used to connect Tele- phone Company Access Service arranged for two-way combination service to and from the attendant position and from the dial switching equipment of a system	СДН	\$11.80	\$44.09
Automatic PCA used to connect Tele- phone Company Access Service arranged for one-way incoming service to the attendant position of a system	) <b>CE</b> [	charges apply	ICB rates and charges apply
Automatic PCA used to connect Tele- phone Company Access Service arranged for one-way outgoing service from the attendant position of a system	IR:	1986 5.40 MCB <sup>S</sup> Fates and charges apply	ICB rates and charges apply
		ал тала ал ал ал ал ал ал ал ал ал ал ал ал	FILED
sued: DEC 2 9 1983 Effec	ctive:	JAN 0 1 1984	JAN - 1 1924
By R. D. BARRON, Vice Pre Southwestern Bell Tele St. Louis, Mis	ephone	company a	83-253 lic Service Commissi

Access Services Tariff Section 13 1st Revised Sheet 41 Replacing Original Sheet 41

## (CP)ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

## 13.3 Miscellaneous Services-(Continued)

## 13.3.7 Protective Connecting Arrangements-(Continued)

	Description	<u>USOC</u>	Monthly <u>Rates</u>	Nonrecurring <u>Charges</u>
(AT)	Automatic PCA used to connect Telephone Company Switched Access Service arranged			
(AT)	for one-way outgoing service from the dial switching equipment of a CPE system	PFA++	ICB rates and charg	ges apply
(AT)	Automatic PCA used to connect Telephone Company Switched Access Service arranged for two-way service to and from the			
(CR)	attendant position of a CPE system	CD9	4.85	\$52.13
(AT)	PCA used for automatic connection of CPE voice transmitting and/or receiving terminal equipment bridged to Telephone			
(CR)	Company Switched Access Service	C2AKS	6.66	52.90
(AT)	Automatic PCA used to connect Telephone Company Switched Access Service arranged for one-way service, i.e., outward dial- ing by hotel/motel guests to the operator			
(RT) (CT)	position (the equivalent of a toll terminal)	PFV++	ICB rates and char	ges apply
(AT)	PCA to provide for connection of CPE originate-only or originate and			
(CT)	answer terminal equipment	PFW++	ICB rates and char	ges apply



Access Services Tariff Section 13 Original Sheet 41

#### ACCESS SERVICES

13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVECES COS (Continued)

13.3 Miscellaneous Services-(Continued)

P. Hip C.

. 13.3.7 Protective Connecting Arrangements-(Continued)

Description	USOC	Monthly Rates	Nonrecurring Charges
Automatic PCA used to connect Tele- phone Company Access Service arranged for one-way outgoing service from the dial switching equipment of a system	. CD8	ICB rates and charges apply	ICB rates and charges apply
Automatic PCA used to connect Tele- phone Company Access Service arranged for two-way service to and from the attendant position of a system	. СД9	\$ 8.81	\$44.09
PCA used for automatic connecting of voice transmitting and/or receiving terminal equipment bridged to Tele- phone Company Access Service, which is terminated in a station	. C2AKS	10.61	8.81
Automatic PCA used to connect Tele- phone Company Access Service arranged for one-way service, i.e., outward dialing by hotel/motel guests to the operator position of a Telephone Com- pany long distance switchboard (the equivalent of a toll terminal)	. CFT	ICB rates and charges apply	ICB rates and charges apply
PCA to provide for connection of originate-only or originate and manswer terminal equipment BAND	<b>Esuda</b> d 1 1986	GLOB rates and charges apply	ICB rates and charges apply
PUBLIC SERVI	R.S.4 ICE COMMIS MISSOUL		
	ective:	JAN 0 1 1984 JA 8	<b>6 6 6 1</b>
By R. D. BARRON, Vice Pr Southwestern Bell Tel	lephone (	Company [Public Se	rvice Commission

St. Louis, Missouri

Access Services Tariff Section 13 Original Sheet 42

## (CP)ACCESS SERVICES

## 13. ADDITIONAL ENGINEERING, ADDITIONAL LABOR AND MISCELLANEOUS SERVICES-(Continued)

- 13.3 Miscellaneous Services-(Continued)
  - 13.3.8 900 Call Restriction

An optional service available, where facilities permit, in Telephone Company electronic end offices. This service provides end office blocking of originating 900 calls by routing such calls to a recorded announcement.

Charges associated with 900 Call Restriction will be waived, on a one-time basis, to subscribers of a new telephone number for a period of 60 days after the new number is effective. For subsequent blocking requests, the Telephone Company will assess a nonrecurring charge to the subscriber on a per line basis. Any request by a subscriber to remove 900 Call Restriction must be submitted in writing to the Telephone Company.

	Nonrecurring Charge
(A) FGA/BSA-A (RBVXF)	
Per line - first request	\$25.00
Per line - subsequent request	\$25.00
(B) WATS Access Line Service (RBVXF)	
Per line - first request	\$25.00
Per line - subsequent request	\$25.00
- Any customers of WATS Access Line Service, including end users purchasing WAL Service out of	f

the Telephone Company's general exchange tariff, may be considered a customer of 900 Call Restriction for WAL Service.

Issued: November 4, 1994

